UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

(Mark one)

[X]QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2003

or []TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____

Commission File Number: 0-24277

CLARUS CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

58-1972600

(State or other jurisdiction of incorporation or organization) (I.R.S. Employer Identification Number)

One Pickwick Plaza Greenwich, Connecticut 06830

(Address of principal executive offices) (Zip code)

(203) 302-2000

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter periods that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. YES [X] NO []

Indicate by check mark whether the Registrant is an accelerated filer (as defined in Rule 12b-2 of the Act). YES [X] NO []

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practical date.

COMMON STOCK, (\$.0001 PAR VALUE)

16,502,058 shares outstanding as of November 10, 2003

INDEX CLARUS CORPORATION

<TABLE> <CAPTION> PART I FINANCIAL INFORMATION

	Consolidated Statements of Operations (unaudited) - Three and nine months ended September 30, 2003 and 2002
	Consolidated Statements of Cash Flows (unaudited) - Nine months ended September 30, 2003 and 2002
	Notes to Unaudited Consolidated Financial Statements (unaudited) - September 30, 2003
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations
Item 3.	Quantitative and Qualitative Disclosures About Market Risk
Item 4.	Procedures and Controls
PART I	I OTHER INFORMATION
Item 1.	Legal Proceedings
Item 4.	Submission of Matters to a Vote of Security Holders
Item 6.	Exhibits and Reports on Form 8-K 17
SIGNA <td>ГURE 17 E></td>	ГURE 17 E>

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

CLARUS CORPORATION CONSOLIDATED BALANCE SHEETS (UNAUDITED) (IN THOUSANDS, EXCEPT SHARE AND PER SHARE AMOUNTS)

<TABLE> <CAPTION>

<caption></caption>					
	SEPTEMB	DECEN	ABER 31	l,	
	2003	200	02		
<\$>	<c></c>		<c></c>		
ASSETS					
CURRENT ASSETS:					
Cash and cash equivalents		\$10,894	\$	42,225	
Marketable securities	,	77,702	52,	385	
Accounts receivable, less allowance for doubtful acc	counts				
of \$0 and \$586 in 2003 and 2002, respectively				467	7
Prepaids and other current assets		980		1,262	
Assets held for sale			48		
Total current assets	89	9,576	96,88	37	
PROPERTY AND EQUIPMENT, NET			4		809
OTHER ASSETS:					
Deposits and other long-term assets		37		68	
TOTAL ASSETS		\$89,617	\$	97,764	
LIABILITIES AND STOCKHOLD	ERS' EQUIT	ſΥ			
CURRENT LIABILITIES:					
Accounts payable and accrued liabilities			375	\$1,93	6
Deferred revenue	1	1,132	1,24		
Current portion of long-term debt				,000	
Liabilities to be assumed			220	1	
Total current liabilities	3	3,007	8,40	4	
LONG-TERM LIABILITIES:					
Other long-term liabilities					

Total liabilities	3,007	8,404		
STOCKHOLDERS' EQUITY:				
Preferred stock, \$.0001 par value; 5,000,000 shares authoriz	ed; none issued			
Common stock, \$.0001 par value; 100,000,000 shares author	orized;			
16,564,358 and 15,762,707 shares issued and 16,489,358 a	nd 15,687,707			
outstanding in 2003 and 2002, respectively		2 2		
Additional paid-in capital	366,393	361,715		
Accumulated deficit	(276,562)	(272,436)		
Treasury stock, at cost	(2)	(2)		
Accumulated other comprehensive income		98 14	-6	
Deferred compensation	(3,319)	(65)		
Total stockholders' equity	86,610	89,360		
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	 [\$89,6	17 \$9	7,764

</TABLE>

SEE ACCOMPANYING NOTES TO UNAUDITED CONSOLIDATED FINANCIAL STATEMENTS.

1

CLARUS CORPORATION CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED) (IN THOUSANDS, EXCEPT PER SHARE AMOUNTS)

<TABLE>

<CAPTION>

<caption></caption>	THREE MONTHSNINE MONTHSENDED SEPTEMBER 30,ENDED SEPTEMBER 30,
	2003 2002 2003 2002
<s> REVENUES: License fees</s>	<pre> \$ 162 \$ \$ 2,649</pre>
Services fees	25 1,354 104 5,352
Total revenues	25 1,516 104 8,001
COST OF REVENUES:	
License fees Services fees	5 21 990 4,809
Total cost of revenues	995 4,830
OPERATING EXPENSES: Research and development Sales and marketing General and administrative Intangible impairment loss Provision for doubtful accounts Depreciation and amortization	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Total operating expenses	800 5,036 5,143 36,031
OPERATING LOSS OTHER INCOME/(EXPENSE) INTEREST INCOME INTEREST EXPENSE	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
NET LOSS	\$ (672) \$ (3,989) \$ (4,126) \$ (31,017)
= Loss per common share: Basic Diluted	\$ (0.04) \$ (0.26) \$ (0.26) \$ (1.99) \$ (0.04) \$ (0.26) \$ (0.26) \$ (1.99)
Weighted average shares outstand Basic Diluted 	

 ing: 15,975 15,630 15,867 15,597 15,975 15,630 15,867 15,597 |2

CLARUS CORPORATION CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (IN THOUSANDS, EXCEPT SHARE AMOUNTS)

<TABLE>

<CAPTION>

NINE MONTHS ENDED SEPTEMBER 30, 2003 2002 _____ _____ <S> <C> <C> CASH FLOWS FROM OPERATING ACTIVITIES: Net loss \$(4,126) \$(31,017) Adjustments to reconcile net loss to net cash used in operating activities: Depreciation and amortization on property and equipment 3,404 762 Loss/(Gain) on sale of marketable securities 125 (15)Impairment of intangible assets 10,360 Amortization of intangible assets 455 Noncash sales and marketing expense 450 Noncash general and administrative expense 182 Provision for doubtful accounts 18 (297)Loss/(Gain) on disposal of property & equipment 36 918 Changes in operating assets and liabilities: 449 1.900 Accounts receivable Prepaid and other current assets 984 282 Assets held for sale 48 Deposits and other long-term assets 31 443 Accounts payable and accrued liabilities (1,306)(61)Deferred revenue (116)(4,923)Liabilities to be assumed (220)Other long-term liabilities (9) NET CASH (USED IN)/PROVIDED BY OPERATING ACTIVITIES (2,590)(18,653)CASH FLOWS FROM INVESTING ACTIVITIES: Purchases of marketable securities (100,915)(97,690) Proceeds from sale of marketable securities 14,025 4,228 Proceeds from maturity of marketable securities 61,899 73,717 Proceeds from sale of investments 200 Proceeds from sale of property & equipment 83 11 Purchases of property and equipment (4)(182)NET CASH (USED IN)/PROVIDED BY INVESTING ACTIVITIES (24, 984)(19,644)CASH FLOWS FROM FINANCING ACTIVITIES: Proceeds from the exercises of stock options 268 1,243 Repayment of debt (5,000)Proceeds from issuance of common stock related to employee stock purchase plan 119 ---NET CASH (USED IN)/PROVIDED BY FINANCING ACTIVITIES (3,757)387 27 Effect of exchange rate change on cash DECREASE IN CASH AND CASH EQUIVALENTS (31,331) (37, 883)CASH AND CASH EQUIVALENTS, Beginning of Period 42,225 55,628 _____ CASH AND CASH EQUIVALENTS, End of Period \$10,894 \$17,745

SUPPLEMENTAL CASH FLOW DISCLOSURE:

Cash paid for interest

\$ ---

\$113

</TABLE>

SEE ACCOMPANYING NOTES TO UNAUDITED CONSOLIDATED FINANCIAL STATEMENTS.

3

CLARUS CORPORATION NOTES TO UNAUDITED CONSOLIDATED FINANCIAL STATEMENTS

NOTE 1. BASIS OF PRESENTATION

The accompanying unaudited consolidated financial statements of Clarus Corporation and subsidiaries ("Clarus" or the "Company," which may be referred to as "we," "us," or "our") for the three and nine months ended September 30, 2003 and 2002, have been prepared in accordance with accounting principles generally accepted in the United States of America and instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information in notes required by accounting principles generally accepted in the United States of America for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) necessary for a fair presentation of the unaudited consolidated financial statements have been included. The results of the three and nine months ended September 30, 2003 are not necessarily indicative of the results to be obtained for the year ending December 31, 2003. These interim financial statements should be read in conjunction with the Company's audited consolidated financial statements and footnotes thereto included in the Company's Form 10-K for the fiscal year ended December 31, 2002, filed with the Securities and Exchange Commission.

NOTE 2. SIGNIFICANT EVENTS

As part of our previously announced strategy to limit operating losses and enable the Company to redeploy its assets and use its substantial cash and cash equivalent assets to enhance stockholder value, on December 6, 2002, we sold substantially all of our electronic commerce business, which represented substantially all of our revenue-generating operations and related assets. During January 2003, we sold the assets relating to our Cashbook product representing the remainder of our operating assets.

NOTE 3. EARNINGS PER SHARE

Basic net loss per share attributable to common stockholders is computed by dividing the net loss attributable to common stockholders by the weighted average number of shares of common stock outstanding for each period. Diluted net loss per share attributable to common stockholders is computed by giving effect to all potentially dilutive securities, including options, warrants and redeemable convertible preferred stock. Potentially dilutive securities are excluded from the computation of diluted net loss per share attributable to common stockholders if their effect is anti-dilutive. For the periods ended September 30, 2003 and 2002, basic net loss per share attributable to common stockholders is the same as diluted net loss per share attributable to common stockholders because all potentially dilutive securities were anti-dilutive in computing diluted net loss per share for these periods.

Options and warrants to acquire 983,334 and 1,288,974 shares of common stock during the periods ended September 30, 2003 and 2002, respectively, were outstanding, but not included in the calculation of weighted average shares outstanding because the option and warrant exercise prices were higher than the average market price of the Company's common stock during those periods. In addition, diluted net loss per share attributable to common stockholders excludes the potentially dilutive effect of options and warrants to purchase approximately 149,097 and 764,454 shares of the Company's common stock during the periods ended September 30, 2003 and 2002, respectively, as the Company incurred losses and their inclusion would have been anti-dilutive.

NOTE 4. STOCK-BASED COMPENSATION

The Company accounts for its stock option plans in accordance with the provisions of Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees", and related interpretations. As such, compensation expense is measured on the date of grant only if the current market price of the underlying stock exceeds the exercise price. Such compensation expense is recorded on a straight-line basis over the related vesting period.

SFAS No. 123, "Accounting for Stock-Based Compensation", permits entities to recognize as expense over the vesting period the fair value of all stock-based awards on the date of grant. Alternatively, SFAS No. 123 allows entities to continue to apply the provisions of APB Opinion No. 25 and provide pro forma net income (loss) and pro forma net income (loss) per share disclosures for employee stock option grants as if the fair-value-based method defined in SFAS No. 123 had been applied. The Company has elected to continue to apply the provisions of APB Opinion No. 25 and provide provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of APB Opinion No. 25 and provide the provisions of apply the provisions of APB Opinion No. 25 and provide the provisions of apply the provisions of APB Opinion No. 25 and provide the provisions of apply the provisions of the provisions o

4

In April 2003, the Company granted 500,000 shares of restricted stock to Warren B. Kanders, the Executive Chairman of the Board. The shares vest in ten years or earlier upon satisfaction of various conditions including performance based conditions relating to the price of the Company's common stock. Deferred compensation of \$2,680,000 was recorded at the date of grant representing the fair value of the shares and adjusted as of September 30, 2003 to \$3,435,000 to account for the increase in fair market value from grant date through September 30, 2003. During the three months ended September 30, 2003, \$93,250 was amortized to compensation expense and \$171,500 was amortized to compensation expense during the nine months ended September 30, 2003 to reflect vesting through those periods.

Had compensation cost been determined consistent with the provisions of SFAS No. 123, the Company's pro forma net loss and pro forma net loss per share in accordance with SFAS No. 123 for the three- and nine-month periods ended September 30, 2003 and 2002 would have been as follows (in thousands, except per share amounts):

<TABLE> <CAPTION>

	Septemb	ber 30, 2002	Nine months September 30, 2003 2002		,	
<s> Net loss, as reported</s>	<c></c>	<c></c>	<c></c>	<c></c>		
Add stock-based employee compensation expense net loss, net of tax	e included in 96	n reported 18	18			
Deduct total stock-based employee compensation under fair-value based method for all awards, n	-		(946)	(87)	(4,072)	(478)
Pro forma net loss	\$(1,5 ======	22) \$(4 ======	4,058)	\$(8,016)	\$(31,299)) =====
Basic and diluted net loss per share: As reported Add stock-based employee compensation expens	e included in	n reported			(1.99)	
net loss, net of tax Deduct total stock-based employee compensation under fair-value based method for all awards, n	n expense de	termined			\$ (.26)	\$ (.03)
Pro forma basic and diluted net loss per share		 (.09=====)) == =	(.27) \$	(.51) \$(2	2.01)

</TABLE>

For computing the fair value of stock-based employee awards, the fair value of each option grant has been estimated as of the date of grant using the Black-Scholes option-pricing model with the following assumptions:

2003	3 2	002
		-
Dividend yield	N/A	0%
Expected volatility	N/A	76%
Risk-free interest rate	N/A	2.63%-4.43%
Expected life	N/A	Four years

As there were no grants issued during the three-month or nine-month periods

ended September 30, 2003 the assumptions are not applicable.

NOTE 5. RESTRUCTURING AND RELATED COSTS

During 2002 and 2001, the Company's management approved restructuring plans to reorganize and reduce operating costs. During the first quarter of 2002, the Company determined that amounts previously charged during 2001 of approximately \$202,000 that related to employee separation and related charges were no longer required and this amount was credited to sales and marketing expense in the accompanying consolidated statement of operations during 2002. Restructuring and related charges of \$8.6 million were expensed during 2002. The charges for 2002 were comprised of \$4.6 million for employee separation and related costs for 183 employees and \$4.0 million for facility closures and consolidation costs.

The facility closures and consolidation costs for 2002 relate to the abandonment of the Company's leased facilities in Suwanee, Georgia; Limerick, Ireland; Maidenhead, England; and Toronto, Canada. Total facilities closure and consolidation costs include remaining lease liabilities, construction costs and brokerage fees to sublet the abandoned spaces, net of estimated

5

sublease income. The estimated costs of abandoning these leased facilities, including estimated costs to sublease, were based on market information trend analysis provided by a commercial real estate brokerage firm retained by the Company. The Company incurred an additional charge in the fourth quarter 2002 of \$2.1 million for facility closure and consolidation costs as a result of the termination of its lease for the facility in Suwanee, Georgia.

The following is a reconciliation of the components of the accrual for restructuring and related costs, the amounts charged against the accrual during 2002 and 2003 and the balance of the accrual as of September 30, 2003:

<TABLE> <CAPTION>

NINE MONTHS ENDED SEPTEMBER 30, 2003

	BALANCE ACCRUALS DECEMBER DURING EXPENDITUI		DITURES				BALANC BER 31,	-	SEPTEMBER 30,		
	31, 2001	2002 E	DURING 2002	DURING		2002	DURING	,	2003	,	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>				
(in thousands)											
Employee sepa	aration										
costs	\$680	\$4,645	\$(4,196)	\$(202)	\$927	\$(708	3) \$21	19			
Facility closure	e costs 1,20	09 3,90	05 (4,977)		137	(12	2) 12	5			
						·					
Total restructu	ring and										
related costs	\$1,889	\$8,550	\$(9,173)	\$(202)	\$1,064	4 \$	(720)	\$344			

</TABLE>

The accrual for restructuring and related costs is included in accounts payable and accrued liabilities in the accompanying consolidated balance sheets.

NOTE 6. COMPREHENSIVE INCOME (LOSS)

The Company utilizes SFAS No. 130 "Reporting Comprehensive Income." SFAS No. 130 establishes standards for reporting and presentation of comprehensive income (loss) and its components of net income (loss) and "Other Comprehensive Income (Loss)". "Other Comprehensive Income (Loss)" refers to revenues, expenses and gains and losses that are not included in net income (loss) but rather are recorded directly in stockholders' equity. The components of comprehensive loss for the three and nine months ended September 30, 2003 and 2002 were as follows:

<TABLE> <CAPTION>

	THREE MO	NTHS ENDED S	SEPTEMBER 3	NINE MONTHS ENDED SEPTEMBER 30		
	2003	2002	2003	2002		
- <\$>	<c></c>	<c></c>	<c></c>	<c></c>		
(in thousands) Net loss Increase/(decrease) in unrealized g	\$ (672) ain	\$ (3,989)	\$ (4,126	5) \$	(31,017)	

on marketable securities Foreign currency translation adjustments	3	(28) 24	30 (78)	(192) 27	
Comprehensive loss	\$ (669)	\$ (3,993)	\$(4,174)	\$(31,182)	

</TABLE>

NOTE 7. CONTINGENCIES

The Company is a party to the following pending judicial and administrative proceeding. Following its public announcement on October 25, 2000, of its financial results for the third quarter of 2000, the Company and certain of its directors and officers were named as defendants in fourteen putative class action lawsuits filed in the United States District Court for the Northern District of Georgia. The fourteen class action lawsuits were consolidated into one case, Case No. 1:00-CV-2841, pursuant to an order of the court dated November 17, 2000. A consolidated amended complaint was then filed on May 14, 2001 on behalf of all purchasers of common stock of the Company during the period beginning December 8, 1999 and ending on October 25, 2000.

Generally the amended complaint alleges claims against the Company and the other defendants for violations of Sections 10(b) and 20(a) of the Securities Exchange Act of 1934, as amended, and Rule 10b-5 promulgated thereunder. Generally, it is alleged that the defendants made material misrepresentations and omissions in public filings made with the Securities and Exchange Commission and in certain press releases and other public statements. The amended complaint alleges that the market price of the Company's common stock was artificially inflated during the class periods. The plaintiffs seek unspecified compensatory damages and costs (including attorneys' and expert fees), expenses and other unspecified relief on behalf of the classes. The Court denied a motion to dismiss brought by the defendants and the case is currently in discovery.

6

After reviewing the proceeding that is currently pending (including the probable outcome, reasonably anticipated costs and expenses, availability and limits of insurance coverage) the Company believes the outcome of this proceeding will not have a material adverse effect on our liquidity, financial condition or results of operations. However, the results of complex legal proceedings are difficult to predict. An unfavorable resolution of the proceeding that is currently pending could adversely affect the Company's business, results of operations, liquidity or financial condition. The Company has not reflected any amounts in the financial statements related to this matter.

NOTE 8. NEW ACCOUNTING PRONOUNCEMENTS

In January 2003, the Financial Accounting Standards Board (FASB) issued Interpretation No. 46, "Consolidation of Variable Interest Entities, an interpretation of ARB No. 51." This interpretation addresses the consolidation by business enterprises of variable interest entities as defined in the interpretation. The application of this interpretation did not have a material effect on the Company's consolidated financial statements.

In December 2002, the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation, Transition and Disclosure", an amendment of SFAS No. 123. SFAS No. 148 amends SFAS No. 123, "Accounting for Stock-Based Compensation", to provide alternative methods of transition for a voluntary change to the fair value method of accounting for stock-based employee compensation. In addition, this statement amends the disclosure requirements of SFAS No. 123 to require prominent disclosures in both annual and interim financial statements. Certain of the disclosures are required for quarterly financial reporting and are included in the notes to these consolidated financial statements.

In November 2002, the FASB issued Interpretation No. 45, "Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness to Others an interpretation of FASB Statements No. 5, 57 and 107 and a rescission of FASB Interpretation No. 34". This interpretation elaborates on the disclosures to be made by a guarantor in its interim and annual financial statements about its obligations under guarantees issued. Interpretation No. 45 also clarifies that a guarantor is required to recognize, at inception of a guarantee, a liability for the fair value of the obligation undertaken. The initial recognition and measurement provisions of the interpretation are applicable to guarantees issued or modified after December 31, 2002. The disclosure requirements are effective for financial statements of interim or annual periods ending after December 15, 2002. The Company does not currently have any guarantees requiring disclosure. Accordingly the application of this interpretation did not have a material effect on the Company's consolidated financial statements.

In September 2002, the FASB issued SFAS No. 146, "Accounting for Costs Associated with Exit or Disposal Activities". SFAS No. 146 addresses financial accounting and reporting for costs associated with exit or disposal activities and nullifies Emerging Issues Task Force ("EITF") Issue No. 94-3, "Liability Recognition for Certain Employee termination Benefits and Other Costs to Exit an Activity (including Certain Costs Incurred in a Restructuring)". The provisions of SFAS No. 146 are effective for exit or disposal activities that are initiated after December 31, 2002, with early application encouraged. The adoption of SFAS No. 146 did not have a material impact on the Company's consolidated financial statements.

In August 2001, the FASB issued SFAS No. 143 "Accounting for Asset Retirement Obligations". SFAS No. 143 addresses financial accounting and reporting obligations associated with the retirement of tangible long-lived assets and the associated asset retirement costs. The Company adopted SFAS No. 143 on January 1, 2003. The adoption of SFAS No. 143 did not have a material impact on the Company's consolidated financial statements.

In November 2002, the EITF reached a consensus on Issue No. 00-21, "Revenue Arrangements with Multiple Deliverables". EITF Issue No. 00-21 provides guidance on how to account for arrangements that involve the delivery or performance of multiple products, services and/or rights to use assets. The provisions of EITF Issue No. 00-21 will apply only to revenue arrangements entered into in fiscal periods beginning after June 15, 2003. Clarus Corporation does not expect the adoption of EITF Issue No. 00-21 to have any impact on its financial position or results of operations.

In May 2003, the FASB issued SFAS No. 150, "Accounting for Certain Financial Instruments with Characteristics of both Liabilities and Equity". SFAS No. 150 requires that certain financial instruments that are settled in cash, including certain types of mandatorily redeemable securities, be classified as liabilities rather than as equity or temporary equity. SFAS No. 150 becomes effective for financial instruments entered into or modified after May 31, 2003, and otherwise is effective at the beginning of the first interim period after June 15, 2003 except for private companies in which the effective date is January 1, 2004. SFAS No. 150 will not affect Clarus Corporation's consolidated financial statements.

NOTE 9. RELATED PARTY TRANSACTIONS

In 2003, the Company entered into an oral agreement with Kanders & Company, an entity owned and controlled by the Company's Executive Chairman, Warren B. Kanders, pursuant to which the Company subleased approximately 1,989 square feet in Greenwich, Connecticut from Kanders & Company for \$9,572 a month (subject to increases every three years). In June 2003, this agreement with Kanders & Company was terminated as the underlying lease held by Kanders & Company for the

7

Greenwich property was voluntarily terminated. Included in prepaids and other current assets is \$102,000 which represents all out-of-pocket expenses incurred by the Company in connection with the Greenwich property.

In September 2003, the Company and Kanders & Company, entered into a 15-year lease with a five-year renewal option, as co-tenants to lease approximately 11,500 square feet in Stamford, Connecticut. The Company and Kanders & Company have initially agreed to allocate lease payments on the basis of Kanders & Company renting 2,900 square feet initially for \$6,163 per month, and the Company renting 8,600 square feet initially for \$18,275 per month, which are subject to increase every three years. The lease provides the co-tenants with an option to terminate the lease in years eight and ten in consideration for a termination payment. The Company and Kanders & Company have also agreed to pay for their proportionate share of the build-out construction costs, fixtures, equipment and furnishings related to preparation of the space. In connection with the lease, the Company obtained a stand-by letter of credit in the amount of \$850,000 to secure lease obligations for the Stamford facility. Kanders & Company reimburses the Company for a pro rata portion of the approximately \$5,000 annual cost of the letter of credit. During the three- and nine-month periods ended September 30, 2003, the Company expensed \$30,000 and \$88,000, respectively, for accruals and payments to Kanders Aviation LLC, an affiliate of the Company's Executive Chairman, Warren B. Kanders, for reimbursement of expenses relating to aircraft travel by directors and officers of the Company. This travel related to Board meetings, the closing of the Atlanta facility and meetings for potential redeployment transactions. As of September 30, 2003, \$60,000 of this expense remained outstanding and is included in accounts payable and accrued liabilities in the accompanying balance sheet.

As of September 30, 2003, the Company had outstanding receivables of \$199,000 from Kanders & Company and Kanders Aviation LLC, less accounts payable of \$60,000 for a net receivable of \$139,000. Both of these companies are owned and controlled by Warren B. Kanders, the Company's Executive Chairman. The expenses relate to out-of-pocket expenses incurred by the Company related to the Greenwich property, travel as discussed above and general and administrative costs paid by the Company on behalf of Kanders & Company. The amount outstanding due from these related parties of \$199,000 at September 30, 2003 is included in prepaids and other assets in the accompanying balance sheet.

NOTE 10. RECLASSIFICATIONS

Certain prior period amounts have been reclassified to conform to the current period presentation.

8

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

FORWARD-LOOKING STATEMENTS

This report contains certain forward-looking statements, including information about or related to our future results, certain projections and business trends. Assumptions relating to forward-looking statements involve judgments with respect to, among other things, future economic, competitive and market conditions and future business decisions, all of which are difficult or impossible to predict accurately and many of which are beyond our control. When used in this report, the words "estimate," "project," "intend," "believe," "expect" and similar expressions are intended to identify forward-looking statements. Although we believe that our assumptions underlying the forward-looking statements are reasonable, any or all of the assumptions could prove inaccurate, and we may not realize the results contemplated by the forward-looking statements. Management decisions are subjective in many respects and susceptible to interpretations and periodic revisions based upon actual experience and business developments, the impact of which may cause us to alter our business strategy or capital expenditure plans that may, in turn, affect our results of operations. In light of the significant uncertainties inherent in the forward-looking information included in this report, you should not regard the inclusion of such information as our representation that we will achieve any strategy, objectives or other plans. The forward-looking statements contained in this report speak only as of the date of this report, and we have no obligation to update publicly or revise any of these forward-looking statements.

These and other statements, which are not historical facts, are based largely upon our current expectations and assumptions and are subject to a number of risks and uncertainties that could cause actual results to differ materially from those contemplated by such forward-looking statements. These risks and uncertainties include, among others, our planned effort to redeploy our assets to enhance stockholder value following the sale of substantially all of our revenue generating operations and assets, and the risks and uncertainties set forth in the section headed "Factors That May Affect Our Future Results" of Part I of our Annual Report on Form 10-K, as amended, for the fiscal year ended December 31, 2002. The Company cannot guarantee its future performance.

CRITICAL ACCOUNTING POLICIES AND USE OF ESTIMATES

The Company's discussion of financial condition and results of operations are based on its consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these consolidated financial statements require management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent liabilities at the date of the consolidated financial statements. Estimates also affect the reported amounts of revenues and expenses during the reporting periods. The Company continually evaluates its estimates and assumptions including those related to revenue recognition, allowance for doubtful accounts, impairment of long-lived assets, and contingencies and litigation. The Company bases its estimates on historical experience and other assumptions that are believed to be reasonable under the circumstances. Actual results could differ from these estimates.

The Company believes the following critical accounting policies include the more significant estimates and assumptions used by management in the preparation of its consolidated financial statements.

- - The Company has recognized revenue from two primary sources, software licenses and services. Revenue from software licensing and services fees is recognized in accordance with Statement of Position ("SOP") No. 97-2, "Software Revenue Recognition", and SOP No. 98-9, "Software Revenue Recognition with Respect to Certain Transactions" and related interpretations. The Company recognizes software license revenue when: (1) persuasive evidence of an arrangement exists; (2) delivery has occurred; (3) the fee is fixed or determinable; and (4) collectibility is probable.

- - The Company maintained allowances for doubtful accounts based on expected losses resulting from the inability of the Company's customers to make required payments.

- - The Company had significant long-lived assets, primarily intangibles, as a result of acquisitions completed during 2000. During 2002, the Company evaluated the carrying value of its long-lived assets, including intangibles, according to Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets" and SFAS No. 144 "Accounting for the Impairment or Disposal of Long-Lived Assets". Prior to 2002, the Company periodically evaluated the carrying value of its long-lived assets, including intangibles, according to SFAS No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of". During the fourth quarter of 2001, the Company's evaluation of the performance of the SAI/Redeo companies compared to initial projections, negative economic trends and a decline in industry growth rate projections indicated that the carrying value of these intangible assets exceeded management's revised estimates of future undiscounted cash flows. This assessment resulted in a \$36.8 million impairment charge to goodwill based on the amount by which the carrying amount of these assets exceeded fair value. As a result of a change in the Company's strategic direction during the second quarter of 2002, the Company determined that remaining goodwill and intangible assets should be

9

tested for further impairment. The Company recorded an additional impairment charge to goodwill of \$6.7 million and an impairment charge to intangible assets of \$3.6 million during the three months ended September 30, 2002. The Company recorded no amortization expense related to intangible assets with definite lives during the three and nine months ended September 30, 2003. As a result of adopting SFAS No. 142, the Company did not record amortization expense related to goodwill during 2002.

- - The Company is a party to a pending judicial and administrative proceeding described more fully in Part I, Item 3 of the Company's Annual Report on Form 10-K, as amended, for the fiscal year ended December 31, 2002, Part II, Item 1 of the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2003, and this Quarterly Report on Form 10-Q. After reviewing the proceeding that is currently pending (including the probable outcome, reasonably anticipated costs and expenses, availability and limits of insurance coverage), the Company believes the outcome of this proceeding will not have a material adverse effect on our liquidity, financial condition or results of operations. However, the results of complex legal proceedings are difficult to predict. An unfavorable resolution of the proceeding that is currently pending could adversely affect the Company's business, results of operations, liquidity or financial condition.

SOURCES OF REVENUE

Prior to December 6, 2002, the Company's revenue consisted of license fees and services fees. License fees were generated from the licensing of the Company's suite of software products. Services fees were generated from consulting, implementation, training, content aggregation and maintenance support services. Following the sale of substantially all of the Company's operating assets, the

Company's revenue has consisted solely of the recognition of deferred service fees that are recognized ratably over the maintenance term. This revenue will continue to decrease during 2003 and will be fully recognized by September 30, 2004. Prior to a redeployment of the Company's assets, the Company's principal income, net of general and administrative expenses, will consist of interest, dividend and other investment income from short-term investments that is reported as interest income in the Company's statement of operations.

REVENUE RECOGNITION

Prior to the December 6, 2002 sale of substantially all of the Company's revenue generating operations and assets, the Company recognized revenue from two primary sources, software licenses and services. Revenue from software licensing and services fees was recognized in accordance with SOP No. 97-2, "Software Revenue Recognition", and SOP No. 98-9, "Software Revenue Recognition with Respect to Certain Transactions" and related interpretations. The Company recognizes software license revenue when: (1) persuasive evidence of an arrangement exists; (2) delivery has occurred; (3) the fee is fixed or determinable; and (4) collectibility is probable.

COST OF REVENUES AND OPERATING EXPENSES

Cost of license fees historically included royalties and software duplication and distribution costs. The Company recognized these costs as the products were shipped.

Cost of services fees historically included personnel related expenses and third-party consulting fees incurred to provide implementation, training, maintenance, content aggregation, and upgrade services to customers and partners. These costs were recognized as they were incurred for time and material arrangements and were recognized using the percentage of completion method for fixed price arrangements.

Research and development expenses historically consisted primarily of personnel related expenses and third-party consulting fees. The Company accounted for software development costs under SFAS No. 86, "Accounting for the Costs of Computer Software to be Sold, Leased or Otherwise Marketed". The Company charged research and development costs related to new products or enhancements to expense as incurred until technological feasibility was established, after which the remaining costs were capitalized until the product or enhancement was available for general release to customers. The Company defined technological feasibility as the point in time at which a working model of the related product or enhancement exists. Historically, the costs incurred during the period between the achievement of technological feasibility and the point at which the product is available for general release to customers were not material.

Sales and marketing expenses historically consisted primarily of personnel related expenses, including sales commissions and bonuses, expenses related to travel, customer meetings, trade show participation, public relations, promotional activities, regional sales offices, and advertising.

General and administrative expenses consist primarily of personnel related expenses for financial, administrative and management personnel, fees for professional services, board of director fees and the provision for doubtful accounts. During 2002, the Company allocated the total cost of its information technology function and costs related to the occupancy of its corporate headquarters, to each of the functional areas. Information technology expenses included personnel related expenses, communication charges, and software support. Occupancy charges include rent, utilities, and maintenance services.

10

RESTRUCTURING AND RELATED COSTS

During 2002 and 2001, the Company's management approved restructuring plans to reorganize and reduce operating costs. During the first quarter of 2002, the Company determined that amounts previously charged during 2001 of approximately \$202,000 that related to employee separation and related charges were no longer required and this amount was credited to sales and marketing expense in the accompanying consolidated statement of operations during 2002. Restructuring and related charges of \$8.6 million were expensed during 2002. The charges for 2002 were comprised of \$4.6 million for employee separation and related costs for 183 employees and \$4.0 million for facility closures and consolidation costs.

The facility closures and consolidation costs for 2002 relate to the abandonment of the Company's leased facilities in Suwanee, Georgia; Limerick, Ireland; Maidenhead, England; and Toronto, Canada. Total facilities closure and consolidation costs include remaining lease liabilities, construction costs and brokerage fees to sublet the abandoned space, net of estimated sublease income. The estimated costs of abandoning these leased facilities, including estimated costs to sublease, were based on market information trend analysis provided by a commercial real estate brokerage firm retained by the Company. The Company incurred an additional charge in the fourth quarter 2002 of \$2.1 million for facility closure and consolidation costs as a result of the termination of its lease for the facility in Suwanee, Georgia.

The following is a reconciliation of the components of the accrual for restructuring and related costs, the amounts charged against the accrual during 2002 and 2003 and the balance of the accrual as of September 30, 2003:

<TABLE> <CAPTION>

NINE MONTHS ENDED SEPTEMBER 30, 2003									
	BALANCE DECEMBE 31, 2001	R DURIN		DITURES DURING	BALANCE ADJUSTMI	E ENTS DE	BALANC CEMBER 31, JRING 2003	E EXPENDITURES 2003	SEPTEMBER 30,
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>		
(in thousands)									
Employee sep	\$680	\$4,645	\$(4,196)	\$(202)	\$927	\$(708)	\$219		
Facility closur	re costs 1,2				137	(12)	125		
Total restructor related costs	uring and \$1,889 ========	\$8,550 	\$(9,173)	\$(202)	\$1,064	\$(720)) \$344 		

 | | | | | | | | |The accrual for restructuring and related costs is included in accounts payable and accrued liabilities in the accompanying consolidated balance sheets.

OVERVIEW

AS PART OF OUR PREVIOUSLY ANNOUNCED STRATEGY TO LIMIT OPERATING LOSSES AND ENABLE THE COMPANY TO REDEPLOY ITS ASSETS AND USE ITS SUBSTANTIAL CASH AND CASH EQUIVALENT ASSETS TO ENHANCE STOCKHOLDER VALUE, ON DECEMBER 6, 2002 WE SOLD SUBSTANTIALLY ALL OF OUR ELECTRONIC COMMERCE BUSINESS, WHICH REPRESENTED SUBSTANTIALLY ALL OF OUR REVENUE-GENERATING OPERATIONS AND RELATED ASSETS. THE INFORMATION APPEARING BELOW, WHICH RELATES TO PRIOR PERIODS, IS THEREFORE NOT INDICATIVE OF THE RESULTS THAT MAY BE EXPECTED FOR ANY SUBSEQUENT PERIODS. THE FISCAL 2003 THREE-MONTH AND NINE-MONTH PERIODS PRIMARILY REFLECT, AND FUTURE PERIODS PRIOR TO A REDEPLOYMENT OF ASSETS ARE EXPECTED TO PRIMARILY REFLECT, GENERAL AND ADMINISTRATIVE EXPENSES ASSOCIATED WITH THE CONTINUING ADMINISTRATION OF THE COMPANY AND ITS EFFORTS TO REDEPLOY ITS ASSETS.

RESULTS OF OPERATIONS

REVENUES

Total Revenues. Total revenues for the three months ended September 30, 2003 decreased to \$25,000 from \$1.5 million during the same period in 2002. Total revenues for the nine months ended September 30, 2003 decreased to \$104,000 from \$8.0 million during the same period in 2002. The decrease in total revenues in both periods resulted primarily from the sale of substantially all of the Company's operating assets, discussed above.

License Fees. As a result of the sale of substantially all of the Company's operating assets, discussed above, the Company did not recognize any license fee revenue during 2003. License fees were \$162,000 or 10.7% of total revenues for the

11

three months ended September 30, 2002. License fees were \$2.6 million or 33.1% of total revenues for the nine months ended September 30, 2002.

Services Fees. Services fees decreased to \$25,000 or 100% of total revenues for

the three months ended September 30, 2003, from \$1.4 million or 89.3% of total revenues for the same period in 2002. Services fees decreased to \$104,000 or 100.0% of total revenues for the nine months ended September 30, 2003, from \$5.4 million or 66.9% of total revenues for the same period in 2002. This decrease is primarily attributable to the sale of substantially all of the Company's operating assets, discussed above.

COST OF REVENUES

Total Cost of Revenues. As a result of the sale of substantially all of the Company's operating assets, the Company did not record any cost of revenues during 2003. Cost of revenues were \$995,000 and \$4.8 million during the three and nine months ended September 30, 2002.

Cost of License Fees. As a result of the sale of substantially all of the Company's operating assets, the Company did not record any cost of license fees during 2003. Cost of license fees was \$5,000 during the three months ended September 30, 2002 and \$21,000 for the nine months ended September 30, 2002.

Cost of Services Fees. As a result of the sale of substantially all of the Company's operating assets, the Company did not record any cost of services fees during 2003. Cost of services fees were \$990,000 during the three months ended September 30, 2002 and \$4.8 million during the nine months ended September 30, 2002.

RESEARCH AND DEVELOPMENT EXPENSE

As a result of the sale of substantially all of the Company's operating assets, the Company did not record any research and development expenses during 2003. Research and development expenses were approximately \$1.3 million or 82.8% of total revenues during the three months ended September 30, 2002 and \$6.4 million or 80.5% of total revenues during the nine months ended September 30, 2002.

SALES AND MARKETING

As a result of the sale of substantially all of the Company's operating assets, the Company did not record any sales and marketing expenses during 2003. Sales and marketing expenses were \$1.0 million during the third quarter in 2002 and \$7.7 million during the nine months ended September 30, 2002.

GENERAL AND ADMINISTRATIVE

General and administrative expenses decreased to \$848,000 during the quarter ended September 30, 2003 from \$2.1 million during the same period in 2002. General and administrative expenses decreased to \$4.4 million during the nine months ended September 30, 2003 from \$8.0 million during the same period in 2002. The decrease in general and administrative expenses for the three and nine months ended September 30, 2003 was primarily attributable to the reduction in expenses due to the Company's change in strategic direction as discussed earlier.

PROVISION FOR DOUBTFUL ACCOUNTS

The provision for doubtful accounts for the three months ended September 30, 2003 reflected a recovery of \$48,000 as compared to a recovery of \$300,000 for the comparable period in 2002. The provision for doubtful accounts in the nine months ended September 30, 2003 were \$18,000 as compared to a recovery of \$297,000 for the nine months ended September 30, 2002. The changes in the provision for doubtful accounts were due to the collection of accounts previously thought to be uncollectible.

DEPRECIATION AND AMORTIZATION

Depreciation and amortization decreased to an amount less than \$1,000 in the quarter ended September 30, 2003 from \$971,000 in the same period of 2002. Depreciation and amortization decreased to \$762,000 in the nine months ended September 30, 2003 from \$3.9 million in the same period of 2002. The decrease is primarily attributable to the sale of substantially all of the Company's operating assets, the write-down of certain assets related to the Company's Maidenhead, England, Limerick, Ireland and Suwanee, Georgia facilities during 2002 and the write-off of intangible assets with definite lives during 2002.

OTHER INCOME

For the three months ended September 30, 2003, the Company recorded a loss of \$125,000 from a loss on the sale of securities offset by a slight gain on foreign currency. For the three months ended September 30, 2002, the Company recorded other income of \$14,000. For the nine months ended September 30, 2003, the Company recorded realized gains and losses on marketable securities, foreign currency, sale of assets and early extinguishment of debt totaling to a gain of \$3,000. For the nine months ended September 30, 2002, the Company recorded other income of \$26,000, which was comprised of losses on foreign currency and realized gains and losses on marketable securities.

INTEREST INCOME

Interest income decreased to \$228,000 in the three months ended September 30, 2003 from \$568,000 in the same period of 2002. Interest income decreased to \$976,000 for the nine months ended September 30, 2003 from \$2.0 million for the same period of 2002. The decrease in interest income was due to lower levels of cash and cash equivalents available for investment and a lower rate of return on investments.

INTEREST EXPENSE

Interest expense was zero and \$56,000 for the three months ended September 30, 2003 and 2002, respectively. Interest expense for the nine months ended September 30, 2003 was \$66,000 compared to \$169,000 during the same period of 2002. In March 2000, the Company entered into a \$5.0 million borrowing arrangement with an interest rate of 4.5% with Peachtree Equity Partners L.P., assignee of Wachovia Capital Investments, Inc. The interest expense in 2003 and 2002 is related to this agreement. The debt was repaid on April 17, 2003 and interest owing at the time of extinguishment was waived and a gain of \$66,000 was recognized in other income during the nine months ended September 30, 2003.

INCOME TAXES

As a result of the operating losses incurred since the Company's inception, no provision or benefit for income taxes was recorded during the three months and nine months ended September 30, 2003 and 2002, respectively.

LIQUIDITY AND CAPITAL RESOURCES

The Company's cash and cash equivalents decreased to \$10.9 million at September 30, 2003 from \$42.2 million at December 31, 2002. Marketable securities increased to \$77.7 million at September 30, 2003 from \$52.9 million at December 31, 2002. The overall decrease in cash and cash equivalents and marketable securities is due primarily to cash used in operating activities and repayment of indebtedness.

Cash used in operating activities was approximately \$2.6 million during the nine months ended September 30, 2003. This was primarily attributable to the Company's net loss partially offset by a decrease in accounts receivable and prepaid and other current assets. Cash used in operating activities was approximately \$18.7 million during the nine months ended September 30, 2002. The cash used was primarily attributable to the Company's net loss and to a decrease in deferred revenue partially offset by noncash items, an increase in accounts payable and a decrease in accounts receivable and prepaid and other assets.

Cash used in investing activities was approximately \$25.0 million during the nine months ended September 30, 2003. The cash was used primarily for the purchase of marketable securities. Cash used for investing activities was approximately \$19.6 million during the nine months ended September 30, 2002. The cash was used primarily for purchases of marketable securities partially offset by the sale and maturity of marketable securities.

Cash used in financing activities was approximately \$3.7 million during the nine months ended September 30, 2003, compared to cash provided by financing activities of \$387,000 during the nine months ended September 30, 2002. The cash used by financing activities during the nine months ended September 30, 2003 was primarily attributable to repayment of the Company's outstanding indebtedness to Peachtree Equity Partners, L.P. as mentioned earlier, offset by proceeds from stock option exercises. For the nine-month period ending September 30, 2002, cash provided by financing was attributable to proceeds from shares issued under the employee stock purchase plan and stock option exercises.

At September 30, 2003, the Company had net operating loss carryforwards, research and experimentation credit, and alternative minimum tax credit carryforwards for U.S. federal income tax purposes that expire in varying amounts beginning in the year 2009. The Company's ability to benefit from certain net operating loss carryforwards is limited under Section 382 of the Internal Revenue Code as the Company is deemed to have had an ownership change of greater than 50%. Accordingly, certain net operating losses may not be realizable in future years due to this limitation.

13

The following summarizes the Company's contractual obligations and commercial commitments at September 30, 2003 with initial or remaining terms of one or more years, and the effect such obligations are expected to have on our liquidity and cash flow in future periods:

<TABLE>

<caption></caption>

	TOTAL	2003	2004	2005	2006	2007	THEREAFTER	
<s> (in thousands</s>	<c></c>							
Operating lea	/	30 42	2 277	315	231	219	1,096	
Total	\$2,180	\$ 42			\$231	\$219	\$1,096	

</TABLE>

The Company does not have commercial commitments under capital leases, lines of credit, stand-by lines of credit, guaranties, stand-by repurchase obligations or other such arrangements, other than the stand-by letter of credit described in "Related Party Transactions" below.

The Company does not engage in any transactions or have relationships or other arrangements with an unconsolidated entity. These include special purpose and similar entities or other off-balance sheet arrangements. The Company also does not trade in energy, weather or other commodity-based contracts.

RELATED PARTY TRANSACTIONS

In 2003, the Company entered into an oral agreement with Kanders & Company, an entity owned and controlled by the Company's Executive Chairman, Warren B. Kanders, pursuant to which the Company subleased approximately 1,989 square feet in Greenwich, Connecticut from Kanders & Company for \$9,572 a month (subject to increases every three years). In June 2003, this agreement with Kanders & Company was terminated as the underlying lease held by Kanders & Company for the Greenwich property was voluntarily terminated. Included in prepaids and other current assets is \$102,000 which represents all out-of-pocket expenses incurred by the Company in connection with the Greenwich property.

In September 2003, the Company and Kanders & Company, entered into a 15-year lease with a five-year renewal option, as co-tenants to lease approximately 11,500 square feet in Stamford, Connecticut. The Company and Kanders & Company have initially agreed to allocate lease payments on the basis of Kanders & Company renting 2,900 square feet initially for \$6,163 per month, and the Company renting 8,600 square feet initially for \$18,275 per month, which are subject to increase every three years. The lease provides the co-tenants with an option to terminate the lease in years eight and ten in consideration for a termination payment. The Company and Kanders & Company have also agreed to pay for their proportionate share of the build-out construction costs, fixtures, equipment and furnishings related to preparation of the space. In connection with the lease, the Company obtained a stand-by letter of credit in the amount of \$850,000 to secure lease obligations for the Stamford facility. Kanders & Company reimburses the Company for a pro rata portion of the approximately \$5,000 annual cost of the letter of credit.

During the three- and nine-month periods ended September 30, 2003, the Company expensed \$30,000 and \$88,000, respectively, for accruals and payments to Kanders Aviation LLC, an affiliate of the Company's Executive Chairman, Warren B. Kanders, for reimbursement of expenses relating to aircraft travel by directors and officers of the Company. This travel related to Board meetings, the closing of the Atlanta facility and meetings for potential redeployment transactions. As of September 30, 2003, \$60,000 of this expense remained outstanding and is included in accounts payable and accrued liabilities in the accompanying balance

sheet.

As of September 30, 2003, the Company had outstanding receivables of \$199,000 from Kanders & Company and Kanders Aviation LLC, less accounts payable of \$60,000 for a net receivable of \$139,000. Both of these companies are owned and controlled by Warren B. Kanders, the Company's Executive Chairman. The expenses relate to out-of-pocket expenses incurred by the Company related to the Greenwich property, travel as discussed above and general and administrative costs paid by the Company on behalf of Kanders & Company. The amount outstanding due from these related parties of \$199,000 at September 30, 2003 is included in prepaids and other assets in the accompanying balance sheet.

After the closing of the sale of the e-commerce software business, Steven Jeffery resigned as the Company's Chief Executive Officer and Chairman of the Board of Directors. Under Mr. Jeffery's employment agreement, he is entitled to receive a severance payment equal to one year's salary of \$250,000, payable over one year. In addition, Mr. Jeffery entered into a three-year consulting agreement with the Company and will receive total consideration of \$250,000 payable over a term of two years. Both of these obligations were accrued as of December 31, 2002. During the nine-month period ended September 30, 2003, Mr. Jeffery received \$281,000 related to these arrangements leaving a balance of \$219,000, which is included in accrued liabilities and accounts payable as of September 30, 2003.

14

In the opinion of management, the rates, terms and considerations of the transactions with the related parties described above approximate those that the Company would have received in transactions with unaffiliated parties.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

There have been no material changes to our exposures to market risk since December 31, 2002.

ITEM 4. PROCEDURES AND CONTROLS

The Company's management carried out an evaluation, with the participation of the Company's principal executive officer and principal financial officer, of the effectiveness of the Company's disclosure controls and procedures. Based upon that evaluation, management concluded that as of the end of the period covered by this Form 10-Q the Company's disclosure controls and procedures are effective.

Except as set forth below, there have been no changes in the Company's internal control over financial reporting that have materially affected the Company's internal control over financial reporting during the quarter ended September 30, 2003.

In the Company's Form 10-Q for the quarter ended June 30, 2003, the Company noted that staff reductions arising out of the discontinuance of its operations and its effort to reduce administrative expenses pending redeployment of its assets resulted in management's recognition of a deficiency in the Company's financial reporting process and the need to retain additional financial reporting assistance to the extent necessary. The Company believes that it has taken adequate steps to address the deficiencies previously noted by allowing more time for the preparation of its financial statements and increasing the internal review process over the preparation of the financial statements and the related disclosures. Therefore, the Company believes there is no longer a need to report the deficiency previously reported in the Company's Form 10-Q for the quarter ended June 30, 2003.

15

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

The Company is a party to various pending judicial and administrative proceedings, which, except as set forth below, are described more fully in Part I, Item 3 of the Company's Annual Report on Form 10-K, as amended, for the fiscal year ended December 31, 2002 and Part II, Item 1 of the Company's

Quarterly Report on Form 10-Q for the quarter ended June 30, 2003. After reviewing the proceedings that are currently pending (including the probable outcomes, reasonably anticipated costs and expenses, availability and limits of insurance coverage), we do not believe that any liabilities that may result from these proceedings are likely to have a material adverse effect on our liquidity, financial condition or results of operations. However, the results of complex legal proceedings are difficult to predict. An unfavorable resolution of the proceedings that are currently pending could adversely affect the Company's business, results of operations, liquidity or financial condition.

During 2002, ten former employees of the Company commenced an action in the United States District Court for the Northern District of Georgia-Atlanta Division seeking back pay, employee benefits, interest and attorneys' fees. On July 31, 2003, the case was dismissed without prejudice to the right of any party to reopen the matter on or before August 15, 2004.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

As previously disclosed in the Company's quarterly report on Form 10-Q for the quarter ended June 30, 2003, we held our annual meeting of stockholders on July 24, 2003. Of the 16,463,024 shares of common stock entitled to vote at the meeting, 14,657,320 shares of common stock were present in person or by proxy and entitled to vote. Such number of shares represented approximately 89% of our outstanding shares of common stock. Listed below are the matters voted upon at our annual meeting of stockholders and the respective voting results:

<TABLE> <CAPTION>

	Voted FOR	Withh		ained/ Broker Non-Votes	
	<c></c>	<c></c>		<c></c>	
Election of Directors: Tench Coxe Burtt R. Ehrlich Donald L. House Stephen P. Jeffrey Warren B. Kanders Nicholas Sokolow	13,5 13,6 12 13,1	-	1,081 972 1,80 1,62 2,1	,061 ,396 - 61,533 4,245 121,164	-
	Voted FOR	Votec AGAI		Abstaine	ed
Approval of the proposal to amend our Amended and Restated Certificate of Incorporation to restrict certain transfers of its securities in order to help assure the preservation of our tax net operating loss carryforwards:			3	1,235,829	11,795
Ratification of the appointment of KPMG LLP as our independent auditors for the year ending December 31, 2003: 					

16 12,504,744 | | 677,341 | 8,781 || ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K | | | | | | |
(a) Exhibits					
Number Exhibit					
10.1 Lease dated as of Senter	nher 23-2003 hets	veen Reckson (Onerati	no	

- 10.1 Lease dated as of September 23, 2003 between Reckson Operating Partnership, L.P., Clarus Corporation, and Kanders & Company, Inc.
- 31.1 Certification of Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Principal Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.

32.1 Certification of Principal Executive Officer pursuant to Section 906 of

the Sarbanes-Oxley Act of 2002.

32.2 Certification Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

(b) Reports on Form 8-K

During the quarter ended September 30, 2003, the Company filed Current Reports on Form 8-K on August 18, 2003 and August 19, 2003, with respect to Items 7 and 9, relating to press releases dated August 18, 2003 and August 19, 2003, announcing the Company's earnings for the three-month period ended June 30, 2003.

SIGNATURE

Pursuant to the requirements of the Securities and Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

CLARUS CORPORATION

Date November 12, 2003

17

EXHIBIT INDEX

Number Exhibit

- -----

- 10.1 Lease dated as of September 23, 2003 between Reckson Operating Partnership, L.P., Clarus Corporation, and Kanders & Company, Inc.
- 31.1 Certification of Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Principal Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- 32.2 Certification Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

RECKSON OPERATING PARTNERSHIP, L.P. LANDLORD

AND

CLARUS CORPORATION AND KANDERS & COMPANY, INC. TENANT

ONE LANDMARK SQUARE STAMFORD, CONNECTICUT

LEASE

DATED: SEPTEMBER 23, 2003

TABLE OF CONTENTS

<table> <caption></caption></table>
<s> <c> Demise, Premises, Term, Rents</c></s>
Demise, Plemises, Term, Rents2
Use
Preparation of the Demised Premises5
When Demised Premises Ready for Occupancy
Security Deposit7
Adjustments of Rent for Changes in Real Estate Taxes9
Adjustment of Rent for Changes in Operating Costs10
Subordination, Attornment, Notice to Lessor and Mortgagees
Quiet Enjoyment14
Assignment, Mortgaging, Subletting14
Compliance with Laws and Requirements of Public Authorities
19
Insurance18
Rules and Regulations
Rules and Regulations19
Rules and Regulations
Rules and Regulations. 19 Alterations and Tenant's Property. 20 Repairs and Maintenance. 24
Rules and Regulations. 19 Alterations and Tenant's Property. 20 Repairs and Maintenance. 24 Electrical Energy. 24
Rules and Regulations.19Alterations and Tenant's Property.20Repairs and Maintenance.24Electrical Energy.24Heat, Ventilation and Air Conditioning.25
Rules and Regulations.19Alterations and Tenant's Property20Repairs and Maintenance24Electrical Energy24Heat, Ventilation and Air Conditioning25Landlord's Other Services25
Rules and Regulations.19Alterations and Tenant's Property20Repairs and Maintenance24Electrical Energy24Heat, Ventilation and Air Conditioning25Landlord's Other Services25Access, Changes in Building Facilities, Name27
Rules and Regulations.19Alterations and Tenant's Property.20Repairs and Maintenance.24Electrical Energy.24Heat, Ventilation and Air Conditioning.25Landlord's Other Services.25Access, Changes in Building Facilities, Name.27Shoring, Notice of Accidents, etc.28

Surrender	33
Conditions of Limitation	
Re-entry by Landlord - Default Provisions	
Damages	36

Waivers
No Other Waivers or Modifications
Sums Due Landlord
Consents - Broker
Notices
Estoppel Certificate40
No Other Representations, Construction, Governing Law41
Parties Bound41
Certain Definitions and Constructions41
Subordination and Miscellaneous43
Partnership Tenant44
Tenant's Authority to Enter Lease45
Parking45
Satellite Dish46
Renewal Option47
Cancellation Option48
Right of First Offer49
Penthouse Space
Building Renovations
EXHIBIT A52
EXHIBIT B53
EXHIBIT C54
EXHIBIT D57
EXHIBIT E64
EXHIBIT F65
EXHIBIT G
SCHEDULE A TO LETTER OF CREDIT

L.P., a Delaware limited partnership, having an address at 360 Hamilton Avenue, White Plains, New York 10601 (hereafter referred to as "Landlord"), and the lessee, CLARUS CORPORATION, a Delaware corporation, having its principal place of business at One Pickwick Plaza, Greenwich, Connecticut 06830 and KANDERS & COMPANY, INC., a Delaware corporation, having its principal place of business at One Pickwick Plaza, Greenwich, Connecticut 06830 (hereinafter referred to collectively as the "Tenant").

ARTICLE 1 DEMISE, PREMISES, TERM, RENTS

1.01 Landlord hereby leases to Tenant, and Tenant hereby hires from Landlord, that certain space in the premises hereinafter described, in the building (referred to herein as the "Building") known as One Landmark Square in the City of Stamford, Connecticut (the "City"), for the term hereinafter stated, for the rents hereinafter reserved and upon and subject to the conditions (including limitations, restrictions and reservations) and covenants hereinafter provided. Each party hereto expressly covenants and agrees to observe and perform all of the conditions and covenants herein contained on its part to be observed and performed.

1.02 Landlord and Tenant hereby agree and stipulate that for purposes of this Lease, the space consists of a total of approximately 11,500 rentable square feet on the twenty-second (22nd) floor in the Building which is outlined on the floor plan(s) annexed as Exhibit "A" and hereby made a part hereof. The premises constitute and are hereinafter called the "Premises" or the "Demised Premises."

1.03 The term of this Lease, for which the Demised Premises are hereby leased, shall commence upon execution of this Lease. Subject to the provisions of Article 4 and Section 3.02 hereof, Tenant's right to occupy the Demised Premises and Tenant's obligation to pay fixed rent and all items of additional rent shall commence on the date which is the earliest of(i) the day on which the Demised Premises are ready for occupancy (as defined in Article 4), (ii) the day Tenant or anyone claiming under or through Tenant shall take possession of any part of the Demised Premises (excluding any possession pursuant to Section 3.02), or (iii) the day which is six (6) months after the date of this Lease (the "Rent Commencement Date"). The term of this Lease shall end at noon of the last day of the calendar month in which occurs the end of a fifteen (15) year period from the Rent Commencement Date (the "Expiration Date") or shall end on such earlier date upon which the term may expire or be cancelled or terminated pursuant to any of the conditions or covenants of this Lease or pursuant to law. Promptly following the Rent Commencement Date, Landlord shall send to Tenant a notice fixing the Rent Commencement Date.

1.04 Tenant shall pay to Landlord without notice or demand and without abatement, deduction or setoff, in lawful money of the United States of America, at the office of the Landlord or at such other place as Landlord may designate, the fixed rent and additional rent reserved under this Lease for each year of the term thereof, which payments shall consist of:

(A) Fixed rent (the "fixed rent") as follows:

<TABLE> <CAPTION>

			4 15 1
	Fixed Base Rent	Monthly Fixed	Annual Fixed
Lease Yea	ars Per RSF	Rent Amour	t Rent Amount
<s></s>	<c></c>	<c> <c< td=""><td>></td></c<></c>	>
1	\$25.50	\$24,437.50 \$2	293,250.00
2	\$34.00	\$32,583.33 \$3	391,000.00
3-4	\$35.00	\$33,541.67 \$	402,500.00
5-6	\$37.00	\$35,458.33 \$	425,500.00
7-8	\$40.00	\$38,333.33 \$	460,000.00
9-10	\$42.00	\$40,250.00	\$483,000.00
11-13	\$44.00	\$42,166.67	\$506,000.00
14-15	\$45.00	\$43,125.00	\$517,500.00

 | | |payable in equal monthly installments in advance on the first day of each and every calendar month during the term of this Lease, and (B) Additional rent (the "additional rent") consisting of all other sums of money as shall become due from and payable by Tenant to Landlord hereunder (for default in payment of which Landlord shall have the same remedies as for a default in payment of fixed rent).

1.05 Tenant shall pay to Landlord as additional rent the real estate taxes attributable to the value of improvements made to the Demised Premises by or on behalf of Tenant in excess of building standard improvements in connection with the fit-up of the Demised Premises for Tenant. For purposes of this Section, the amount of the per square foot assessment attributable to the excess improvements shall be the amount in excess of average square foot assessment then being used by the tax assessor's office for the fit-up of space within the Building or, if that is not applicable, to fit-up of similar space in a similar building in Stamford, Connecticut. The real estate taxes resulting from such assessment(s) shall be computed separately and billed to Tenant, and shall be payable by Tenant as additional rent within ten (10) days of the rendition of the bill, at each time that real estate taxes are payable by Landlord with respect to the Demised Premises during the term of this Lease.

If Tenant substantially renovates or alters the Demised Premises or any part thereof after the initial fit-up, Tenant shall pay Landlord the real estate taxes attributable to the increased assessment resulting from such renovation or alteration, in excess of the then assessment for building standard. The real estate taxes resulting from such excess assessment shall be computed separately and billed to Tenant and shall be payable by Tenant as additional rent within ten (10) days of the rendition of a bill at each time that real estate taxes are payable by Landlord with respect to the Demised Premises during the term of the Lease.

1.06 Tenant shall pay the fixed rent and additional rent herein reserved promptly as and when the same shall become due and payable. If the Rent Commencement Date shall occur on a day other than the first day of a calendar month, the fixed rent for such calendar month shall be prorated for the period from the Rent Commencement Date to the last day of the calendar month and shall be due and payable on the Rent Commencement Date and the first Lease Year shall include said partial month. Notwithstanding the provisions of the next preceding sentence or of Section 4.01(a), Tenant shall pay the first full calendar monthly installment of fixed rent on the execution of this Lease. If Tenant shall fail to pay within ten (10) days after the same is due any fixed rent or additional rent, such unpaid amounts shall bear interest at the annual rate equal to two percentage points (2%) above the Prime Rate of Citibank, N.A., New York, New York.

ARTICLE 2 USE

2.01 Tenant shall use and occupy the Demised Premises only for general and executive offices and for no other purposes.

2.02 The use of the Demised Premises for the purposes specified in this Article shall not in any event be deemed to include, and Tenant shall not use, or permit the use of, the Demised Premises or any part thereof for:

- (A) sale of, or traffic in, any spirituous liquors, wines, ale or beer kept in the Demised Premises;
- (B) sale at retail of any other products or materials kept in the Demised Premises, by vending machines (except vending machines serving Tenant's employees only) or otherwise, or demonstrations to the public, except as may be specifically agreed to by Landlord in writing;
- (C) manufacturing, printing or electronic data processing, except for the operation of normal business office reproducing or printing equipment, electronic data processing equipment and other business machines for Tenant's own requirements at the Demised Premises; provided only that such use shall not exceed that portion of the mechanical or electrical capabilities of the Building equipment allocable to the Demised Premises;
- (D) the rendition of medical, dental or other diagnostic or therapeutic services;

- 5
- (F) a restaurant, bar, or the sale of confectionery, tobacco, newspapers, magazines, soda, beverages, sandwiches, ice cream, baked goods or similar items, or the preparation, dispensing or consumption of food and beverages in any manner whatsoever except as otherwise provided in Section 2.07 hereof.

2.03 Tenant shall not suffer or permit the Demised Premises or any part thereof to be used in any manner, or anything to be done therein, or suffer or permit anything to be brought into or kept therein, which would in any way (i) violate any of the provisions of any grant, lease or mortgage to which this Lease is subordinate, (ii) violate any laws or requirements of public authorities, (iii) make void or voidable any fire or liability insurance policy then in force with respect to the Building, (iv) make unobtainable from reputable insurance companies authorized to do business in the State of Connecticut at standard rates any fire insurance with extended coverage, or liability, elevator or boiler or other insurance required to be furnished by Landlord under the terms of any lease or mortgage to which this Lease is subordinate, (v) cause, or in Landlord's opinion be likely to cause, physical damage to the Building or any part thereof, (vi) constitute a public or private nuisance, (vii) impair in the opinion of the Landlord the appearance, character or reputation of the Building, (viii) discharge objectionable fumes, vapors or odors into the Building air conditioning system or into Building flues or vents not designed to receive them or otherwise in such manner as may unreasonably offend other occupants, (ix) impair or interfere with any of the Building services or the proper and economic heating, cleaning, air conditioning or other servicing of the Building or the Demised Premises or impair or interfere with or tend to impair or interfere with the use of any of the other areas of the Building by, occasion or discomfort, annoyance or inconvenience to, Landlord or any of the other tenants or occupants of the Building, or (x) cause Tenant to default in any of its other obligations under this Lease. The provisions of this Section, and application thereof, shall not be deemed to be limited in any way to or by the provisions of the following Sections of this Article or any of the Rules and Regulations referred to in Article 13 or Exhibit "C" attached hereto, except as may therein be expressly otherwise provided.

2.04 If any government license or permit, other than a Certificate of Occupancy, shall be required for the proper and lawful conduct of Tenant's business in the Demised Premises, or any part thereof and if failure to secure such license or permit would in any way affect Landlord, then Tenant, at its expense, shall duly procure and thereafter maintain such license or permit and submit the same to inspection by Landlord. Tenant shall at all times comply with the terms and conditions of each such license or permit, but in no event shall failure to procure and maintain same by Tenant affect Tenant's obligations hereunder.

2.05 Tenant shall not at any time use or occupy or suffer or permit anyone to use or occupy the Demised Premises, or do or permit anything to be done in the Demised Premises, in violation of the Certificate of Occupancy for the Demised Premises or for the Building.

2.06 Tenant shall not place a load upon any floor of the Demised Premises exceeding the floor load per square foot which such floor was designed to carry and which is allowed by certificate, rule, regulation, permit or law. Landlord reserves the right to prescribe the weight and position of all safes and vaults, which must be placed by Tenant, at Tenant's expense.

2.07 Notwithstanding anything to the contrary contained herein, Tenant may, at its sole cost and expense, install and use a household-type oven and stove top, refrigerator and/or microwave oven in the pantry portion of the Demised Premises for use by Tenant's employees and for occasional use for food preparation for clients of Tenant. All such installations and all work relating thereto (e.g. installation of required venting, etc.) shall be performed in accordance with the provisions of Article 14 hereof. Without limiting the generality of any other provisions of this Lease, (a) all such installations shall be performed in compliance with (i) laws, rules, orders and regulations of governmental authorities having jurisdiction thereof, and (ii) rules and regulations of the Landlord attached as Exhibit C hereto; (b) prior to the expiration or earlier termination of this Lease, all such installations shall be removed from the

Demised Premises and the area where such installations are made shall be restored to substantially the condition which exists on the date of this Lease, including removal of all venting and repair of all wall and/or roof penetrations; and (c) Tenant shall indemnify, defend and hold Landlord harmless from any and all claims, damages, losses, liabilities, costs and expenses (including reasonable attorneys' fees and disbursements) arising out of or in connection with the use of any such machines and appliances. In the event that due to Tenant's use of the Demised Premises, insects or vermin are present in or about the Demised Premises, Tenant shall pay the cost for periodic extermination services as Landlord may require. In the event that Landlord determines that Tenant's use of water in the Demised Premises exceeds that of ordinary office use, Landlord may install a water meter to measure Tenant's water consumption and Tenant shall pay the cost of installation of such meter and for Tenant's usage of water as measured thereby.

ARTICLE 3

6

PREPARATION OF THE DEMISED PREMISES

3.01 Landlord's Initial Work. (a) Landlord, at its expense, will cause the Demised Premises to be in "broom clean" condition ("Landlord's Initial Work")

(b) Landlord's agreement to do Landlord's Initial Work shall not require it to incur overtime costs and expenses and shall be subject to unavoidable delays due to acts of God, governmental restrictions, strikes, labor disturbances, shortage of materials and supplies and for any other causes or events beyond Landlord's reasonable control.

3.02 Tenant's Initial Work. Tenant and its employees, agents and contractors shall be permitted access to the Demised Premises upon full execution and delivery of this Lease for the sole purpose of performing work and making installations in the Demised Premises in order to prepare same for Tenant's initial occupancy thereof, which shall include the installation of a building standard sprinkler system in the Demised Premises and the installation of sixty (60) tons of base building HVAC to service the Demised Premises ("Tenant's Initial Work"). All architectural, mechanical and engineering plans and all of Tenant's contractors shall be subject to Landlord's prior written approval which approval shall not be unreasonably withheld. Notwithstanding anything to the contrary contained in this Lease, Tenant shall be required to use the following subcontractors in connection with Tenant's Initial Work: (a) Simplex for fire/life safety systems, (b) Automated Building Controls, Inc. for the energy management system; and (c) Fairfield County Sprinkler for installation of the building standard sprinkler system. In addition, with respect to the installation of sixty (60) tons of base building HVAC to service the Demised Premises (specifically limited to providing (i) the HVAC units and the main trunk line into the Demised Premises with respect to the HVAC units to be installed within the core of the Demised Premises, and (ii) the HVAC units only (i.e. without the distribution system therefor) with respect to the HVAC units to be installed above the drop ceiling) (the "HVAC Work"), Landlord will provide Tenant with a list of the HVAC subcontractors from whom Tenant must solicit bids for the HVAC Work and, upon receipt of such bids, Tenant will promptly provide Landlord with copies thereof. After review of the bids for the HVAC Work, Landlord will have sole discretion to select the HVAC subcontractor which Tenant must select to perform the HVAC Work (which subcontractor need not necessarily be the low bidder). Any work relating to the HVAC system starting from the end of the main trunk line into the Demised Premises (with respect to the HVAC units to be installed within the core of the Demised Premises) and from the HVAC units themselves (with respect to the HVAC units to be installed above the drop ceiling) shall be bid out separately by Tenant. For avoidance of confusion, the parties acknowledge that the HVAC Work is being specifically defined herein in order to facilitate the calculation of the portion of the Tenant Improvement Allowance (as defined in Exhibit D, Paragraph 6) payable by Landlord with respect to the HVAC Work. It is expressly acknowledged and agreed that with respect to the HVAC units to be installed as part of the HVAC Work above the drop ceiling (as opposed to in the core of the Demised Premises), the distribution system therefor will not be installed as part of the HVAC Work but, rather, will be installed as a separate component of Tenant's Initial Work, and therefore the cost of such distribution system will not be included in calculating the HVAC Work component of the Tenant Improvment Allowance.

Landlord shall have the right to designate the specific types of HVAC units to be installed as part of the HVAC Work and the exact locations of all such HVAC units which will be installed within the core of the Demised Premises. All other HVAC units will be installed within the Demised Premises as part of the HVAC Work will be located above the drop ceiling to be installed in the Demised Premises as part of Tenant's Initial Work in locations as shown on the plans for Tenant's Initial Work, which plans are subject to approval by Landlord in accordance with the terms of this Lease. All of Tenant's duties and obligations set forth in Article 14 with respect to Tenant's Work in general and in Exhibit D shall be applicable to and binding upon Tenant with respect to Tenant's Initial Work.

ARTICLE 4 WHEN DEMISED PREMISES READY FOR OCCUPANCY

4.01 The Demised Premises shall be deemed ready for occupancy on the earliest date on which all of the following conditions have been met:

7

- (1) A Certificate of Occupancy (temporary or final or a permission to occupy) for the Demised Premises (not the Building) has been issued not inconsistent with Tenant's use of the Demised Premises as permitted under Section 2.01 hereof.
- (2) Landlord's Initial Work and Tenant's Initial Work have been substantially completed. "Substantially completed" as used herein is defined to mean when the only items to be completed are punchlist items, which items not materially interfere with Tenant's occupancy of the Demised Premises.
- (3) The facilities and systems serving the Building and passing through the Demised Premises have been completed to the extent required to provide adequate services to the Demised Premises.

4.02 If the occurrence of any of the conditions listed in Section 4.01, and thereby the making of the Demised Premises ready for occupancy, shall be delayed due to any act or omission of Tenant within Tenant's control or any of its employees, agents or contractors, including but not limited to, failure by Tenant to plan or execute work to be performed by Tenant or by a party employed by Tenant diligently and expeditiously, then the Demised Premises shall be deemed ready for occupancy on the date when it would have been ready but for any such delay. Each such delay shall hereinafter be referred to as a "tenant delay".

4.03 If the whole of the Demised Premises shall not be ready for occupancy at approximately the same time, Tenant may, with the written consent of Landlord, take possession of any part or parts of the Demised Premises for its use and occupancy before the Rent Commencement Date, provided that a temporary or permanent Certificate of Occupancy shall have been obtained for the part or parts of the Demised Premises in respect of which Tenant desires to take possession. Tenant shall be deemed to have taken possession of a part of the Demised Premises for use and occupancy (herein called "actual possession") when any personnel of Tenant or anyone claiming under or through Tenant shall first occupy such part for the conduct of business. Tenant's actual possession of any part of the Demised Premises prior to the Rent Commencement Date shall be subject to all of the obligations of this Lease, including the payment of fixed and additional rent, which payment shall be reasonably apportioned.

4.04 On the Rent Commencement Date or at such time as Tenant shall take actual possession of the whole or part of the Demised Premises, whichever shall be earlier, it shall be conclusively presumed that the same were in satisfactory condition as of the Rent Commencement Date or the date or dates of such taking of possession, unless within thirty (30) days after such date Tenant shall have given Landlord notice specifying in which respects Landlord's Initial Work was not in satisfactory condition. However, nothing contained in this Section shall be deemed to relieve Landlord from latent defects.

ARTICLE 5 SECURITY DEPOSIT

sum of EIGHT HUNDRED FIFTY THOUSAND AND 00/100 DOLLARS (\$850,000.00) (the "Security Deposit"), receipt of which is hereby acknowledged by Landlord. The Security Deposit shall be held by Landlord in a passbook-type interest bearing account, as security for the faithful performance by Tenant of all of the terms, covenants, and conditions of this Lease by Tenant to be kept and performed during the term hereof. If at any time during the term of this Lease any of the rent herein reserved or any other sum payable by Tenant to Landlord hereunder shall be overdue and unpaid beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), then Landlord may, at the option of Landlord (but Landlord shall not be required to), appropriate and apply any portion of the Security Deposit toward any such overdue rent or other sum. Provided Tenant is not then in default in the payment of fixed rent or additional rent or the performance of its other obligations under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), then, at the end of each of the first ten (10) Lease Years, upon reasonable advance request by Tenant, Landlord shall refund to Tenant a portion of the Security Deposit (or, if applicable, Landlord shall authorize Tenant to reduce the Letter of Credit) so that the amount of the Security Deposit then remaining shall be as follows:

8

<TABLE>

<capt< th=""><th></th><th></th></capt<>		
	End of	Amount of Security
	Lease Year	Deposit Remaining
<s></s>		<c></c>
	1	\$783,300.00
	2	\$716,600.00
	3	\$649,900.00
	4	\$583,200.00
	5	\$516,500.00
	6	\$449,800.00
	7	\$291,600.00
	8	\$224,900.00
	9	\$158,200.00
	10	\$ 91,500.00
<td>LE></td> <td>- ,</td>	LE>	- ,

Notwithstanding the foregoing, the \$158,200.00 refund or reduction of the Letter of Credit following the end of the seventh (7th) Lease Year is further conditioned upon Tenant providing Landlord with either (a) its then most recent audited financial statement reflecting a net worth of at least \$20,000,000.00; or (b) proof acceptable to Landlord in Landlord's reasonable discretion that at such time Tenant has free and clear cash on hand equal to at least three (3) times Tenant's cash requirements for the trailing twelve (12) month period.

Notwithstanding the foregoing, if, at any time during the term, Tenant shall be in default in the payment of fixed rent or additional rent or the performance of its other obligations under this Lease, then the amount of the Security Deposit shall not be reduced as aforesaid and the Security Deposit shall be promptly increased to reflect the amount set forth in the first sentence of this Section 5.01 above.

5.02 In the event of the failure of Tenant to keep and perform any of the terms, covenants and conditions of this Lease to be kept and performed by Tenant, then Landlord, at its option, may appropriate and apply only so much of the Security Deposit as may be necessary to compensate the Landlord for loss or damage sustained or suffered by Landlord due to such breach on the part of Tenant. Should the entire Security Deposit, or any portion thereof, be appropriated and applied by Landlord for the payment of overdue rent or other sums due and payable to Landlord by Tenant hereunder, then Tenant shall, upon the written demand of Landlord, forthwith remit to Landlord a sufficient amount in cash to restore the Security Deposit to the original sum deposited, and Tenant's failure to do so within five (5) days after receipt of such demand shall constitute a breach of this Lease. Provided Tenant is not then in default hereunder, the Security Deposit shall be returned in full to Tenant at the end of the term of this Lease, or upon the earlier termination of this Lease.

5.03 Landlord may deliver the Security Deposit to the purchaser of Landlord's

interest in the Building, in the event that such interest be sold, and thereupon Landlord shall be discharged from any further liability with respect to the Security Deposit, provided the transferee actually receives same.

5.04 In lieu of a cash deposit set forth in Section 5.01, Tenant may deliver, upon its execution of this lease, the Security Deposit to Landlord in the form of a clean, irrevocable, non-documentary and unconditional, stand-by letter of credit in the amount of the Security Deposit (the "Letter of Credit") issued by and drawable upon any commercial bank, trust company, national banking association or savings and loan association satisfactory to Landlord with offices for banking and drawing purposes in New York, New York or Stamford, Connecticut (the "Issuing Bank"), which has outstanding unsecured, uninsured and unguaranteed indebtedness, or shall have issued a letter of credit or other credit facility that constitutes the primary security for any outstanding indebtedness (which is otherwise uninsured and unguaranteed), that is then rated, without regard to qualification of such rating by symbols such as "+" or "-" or numerical notation, "Aa" or better by Moody's Investors Service and "AA" or better by Standard & Poor's Ratings Service (and is not on credit-watch with negative implications), and has combined capital, surplus and undivided profits of not less than \$750,000,000. The Letter of Credit shall (i) name Landlord as beneficiary at the following address: c/o Reckson Associates Realty Corp., 225 Broadhollow Road, CS 5341 Melville, New York 11747, Attention: Corporate Controller, (ii) be in the amount of the Security Deposit, (iii) have a term of not less than one year. (iv) permit multiple drawings, (v) be fully transferable by Landlord multiple times without the consent of Tenant or the payment of any fees or charges, and (vi) otherwise be in form and content satisfactory to Landlord. If upon any transfer of the Letter of Credit, any fees or charges shall be so imposed, then such fees or charges shall be payable solely by Tenant and the Letter of Credit shall so specify. The Letter of Credit shall provide that it shall be deemed automatically renewed, without amendment, for consecutive periods of one year each thereafter during the Term through the date that is at least sixty (60) days after the Expiration Date, or any renewal or extension thereof, unless the Issuing Bank sends a notice (the "Non-Renewal Notice") to Landlord by certified mail, return receipt requested, attention: Corporate Controller (at the address set forth in this Section 5.04 above) not less than 60 days prior to the then-current expiration date of the Letter of Credit, stating that the Issuing Bank has elected not to renew the Letter of Credit. Landlord shall have the right, upon receipt of a Non-Renewal Notice, to draw the full amount of the Letter of Credit, by sight draft on the Issuing Bank, and shall thereafter hold or apply the cash proceeds of the Letter of Credit

9

pursuant to the terms of this Article 5. The Letter of Credit shall state that drafts drawn under and in compliance with the terms of the Letter of Credit will be duly honored upon presentation to the Issuing Bank at an office location in the New York, New York or Stamford, Connecticut. The Letter of Credit must be payable to Landlord or an authorized representative of Landlord upon presentation of only the Letter of Credit and a sight draft, and shall not contain as a condition to draw the requirement of Landlord's certification or other statement as to the existence of Tenant's default. The Letter of Credit shall be subject in all respects to the International Standby Practices 1998, International Chamber of Commerce Publication No. 590. The Letter of Credit shall provide that all fees and expenses are for the account of the applicant. The Letter of Credit shall provide that all costs relating to the Letter of Credit will be billed to and paid by Tenant. Landlord shall not be responsible for bank error. The Letter of Credit shall not be a limitation on Landlord's damages or other rights under this lease, or a payment of liquidated damages or an advance payment of rent. Tenant is required to cooperate with Landlord, at Tenant's expense, if it becomes necessary to have the Letter of Credit re-issued to a new owner or modified or amended as reasonably required by Landlord to carry out the intent, terms and conditions of this Article 5. In the event that Tenant elects to deliver a Letter of Credit in lieu of a cash deposit, then such Letter of Credit shall be in the form annexed hereto as Exhibit G.

- 6.01 (A) The term "Tax Base Year" shall mean the tax fiscal year of July 1, 2003 to June 30, 2004.
 - (B) The term "Common Areas" shall mean the land and pedestrian deck, as shown on Exhibit "F", together with the parking garage and loading dock facility.
 - The term "Real Estate Taxes" shall mean 100% of the taxes and (C) assessments levied, assessed or imposed at any time by any governmental authority upon or against the Building and 48.24% of such taxes and assessments levied against the Common Areas, and also any tax or assessment levied, assessed or imposed at any time by any governmental authority in connection with the receipt of income or rents from the Building and Common Areas, to the extent that same shall be in lieu of or in addition to all or a portion of any of the aforesaid taxes or assessments upon or against the Building and Common Areas. The term "Real Estate Taxes" shall not mean any interest or penalties which may become due by reason of the failure to pay any such taxes when due and payable; or any municipal, state or federal income, estate, inheritance, transfer, corporate or franchise taxes assessed against Landlord unless and to the extent that same is assessed in lieu of part or all of real estate taxes as presently constituted and are computed as if Landlord owned no other property.
 - (D) The term "Tenant's Proportionate Share" shall be 3.85%.
 - (E) The phrase "Real Estate Taxes payable by Landlord" shall not include Real Estate Taxes for which Landlord is reimbursed by Tenant under Section 1.05 or by other tenants of the Building under similar provisions of their leases.
 - In the event that, after a statement has been sent to Tenant, (F) an assessment or valuation which had been utilized in computing the Real Estate Taxes for a tax fiscal year is reduced (as a result of settlement, final determination or legal proceedings or otherwise), and as a result thereof a refund of Taxes is actually received by or on behalf of Landlord, promptly after receipt of such refund, Landlord shall send Tenant a statement adjusting the Real Estate Taxes for such tax fiscal year (taking into account the expenses mentioned in the last sentence of this subparagraph (f) and setting forth Tenant's Proportionate Share of such refund) and Tenant shall be entitled to receive such share by way of a credit against the additional rent next becoming due after the sending of such statement; provided, however, that Tenant's Proportionate Share of such refund shall be limited to the amount, if any, which Tenant had theretofore paid to Landlord as increased additional rent for such tax fiscal year on the basis of the assessed valuation before it had been reduced. All expenses, including attorneys' and appraisers' fees and disbursements, expenses and other witnesses' fees, incurred in contesting the validity or amount of any Real Estate Taxes or in obtaining a refund shall be considered as part of the Real Estate Taxes for such year.
 - (G) In the event that any time during a tax fiscal year after the Tax Base Year the assessment or valuation which had been utilized in computing the Real Estate Taxes for the additional rent for any tax fiscal year is reduced (as a

10

result of settlement, final determination or legal proceedings or otherwise), so that the Taxes payable for such tax fiscal year is less than the Taxes payable by Tenant for the Tax Base Year then and in such event: (i) the Taxes for the Tax Base Year shall become the lower assessment or valuation effective as of the tax fiscal year in which said settlement or final determination was reached (hereinafter the "Updated Tax Base Year"); and (ii) Tenant shall pay as additional rent Tenant's Proportionate Share of the amount by which the Taxes for each succeeding tax fiscal year after the Updated Tax Base Year, exceeds the taxes for the Updated Tax Base Year. Landlord promptly shall send to Tenant a statement setting forth the basis for the Updated Tax Base Year and all adjustments and additional rent payments.

- 6.02 (A) In addition to the fixed rent, Tenant agrees to pay as additional rent an amount equal to Tenant's Proportionate Share of the excess of Real Estate Taxes payable by Landlord for each tax fiscal year of the City which is subsequent to the Tax Base Year, over the Real Estate Taxes payable by Landlord for the Tax Base Year.
 - (B) Tenant's obligation to pay such additional rent under (a) above, shall commence on the July 1st or January 1st (whichever comes first) after the Tax Base Year; and Tenant shall pay such additional rent, with respect of each fiscal year subsequent to the Tax Base Year, in two equal installments, on such July 1st or January 1st (as the case may be) and each subsequent July 1st and January 1st during the balance of the term of this Lease (but in no event shall such payments be due earlier than ten (10) days after delivery to Tenant of a statement therefor).
 - (C) Within thirty (30) days after the Expiration Date, Landlord shall pay to Tenant an amount equal to the portion, if any, of such additional rent which is attributable to the period subsequent to the Expiration Date.
 - (D) The dates for the payment of taxes, July 1st and January 1st, are based on the present dates for the payment of real estate taxes in the City. If the City or other governmental authority changes the dates for the payment of Real Estate Taxes then the changed dates shall be sequentially substituted for the July 1 and January 1 dates contained herein.

6.03 Upon Tenant's request, Landlord shall furnish to Tenant a copy of the Assessor's report or reports showing the assessment for the Building and the Common Areas and the report or reports showing the increased assessment therefor and all applicable tax bills, or such other evidence coming from the Assessor's and/or Tax Collector's office which will show the assessments and tax involved or some other reasonable documentation of the matter.

ARTICLE 7 ADJUSTMENT OF RENT FOR CHANGES IN OPERATING COSTS

- 7.01 (A) In the event that the Operating Costs for an Operating Year shall exceed the Operating Costs for the Base Year, Landlord shall adopt a date (the "Adjustment Date"), in the succeeding Operating Year, and within ten (10) days of the Adjustment Date, Tenant shall pay to Landlord an amount equal to the Tenant's Proportionate Share of such excess (without setoff or deduction of any kind and as additional rent), less the amount of additional rent, if any, which Tenant paid to Landlord, as additional rent under subparagraph (b) below, during the Operating Year in question.
 - (B) Tenant also shall pay to Landlord, as additional rent (in equal monthly installments) in each Operating Year, retroactive to the first month of such year, a sum equal to one-twelfth (1/12th) of Tenant's Proportionate Share of the amount by which the projected Operating Costs for the current Operating Year exceed the Operating Costs for the Base Year. In no event shall Tenant's Proportionate Share of Operating Costs for any Operating Year during the Term hereof be less than Tenant's Proportionate Share of Operating Costs for the Base Year.
 - (C) If the Operating Costs for any Operating Year shall be less than the Operating Costs for the previous year, Landlord shall credit such amount of Tenant's Proportionate Share to Tenant, but in no event will the fixed rent be reduced below that

amount stated in Article 1.

7.02 For the purposes of this Article, the following terms shall have the following meanings:

11

(A) The term "Base Year" shall mean January 1, 2004 through December 31, 2004.

(B) The term "Operating Year" shall mean each twelve month period adopted by the Landlord subsequent to the Rent Commencement Date.

(C) The term "Common Areas" shall mean the land and pedestrian deck, as shown on Exhibit "F", together with the parking garage and loading dock facility.

(D) The term "Adjustment Date" shall be a date adopted by the Landlord subsequent to each Operating Year.

(E) The term "Tenant's Proportionate Share" shall be 3.85%.

(F) The term "Operating Costs" shall mean the aggregate of all expenses paid or incurred by Landlord for the operation of the Building, and 48.24% of such expenses paid or incurred by Landlord for the operation of the Common Areas, and shall include without limitation the following:

(I) Wages and salaries paid by Landlord, including all fringe benefits and taxes related thereto paid by Landlord, of employees directly and solely engaged in cleaning, maintenance and repair of the Building, Building equipment and Common Areas, and performing the functions of garbage and snow removal, landscaping and security, including a customary managing agent's fee, or cost to Landlord of an independent contractor performing any such services;

(II) Any and all supplies and materials utilized by Landlord or independent contractors of Landlord in the performance of the items set forth in subparagraph (i) immediately preceding;

(III) The cost of supplying utilities to the Building and Common Areas;

(IV) Insurance premiums paid by Landlord with respect to the Building and Common Areas; and

(V) Expenditures incurred by Landlord after substantial completion of the Building for any equipment, device or capital improvement which is required by any law, statute or regulation or a requirement of the insurance carrier or which is designed as a labor-saving measure or designed to effect other economies or efficiencies in the operation or maintenance of the Demised Premises, Common Areas or the Building Equipment except to the extent such expenses exceed \$50,000.00, in which case the annual amortization (on a straight-line basis over a depreciable life in accordance with generally accepted accounting principles consistently applied, with interest calculated at an annual rate of one (1) percentage point above the prime rate at the time of Landlord's having made such expenditure) of such expenses shall be included in Operating Costs; and

(VI) Legal and accounting fees and disbursements, and any other expense or charge of any nature whatsoever which, in accordance with generally accepted accounting principles with respect to the operation of a first-class office building, would be construed as an operating expense, excluding, however, real property taxes, depreciation, interest on and amortization of debt, and any items otherwise properly constituting such an operating expense to the extent payment therefor is received from or payable by tenants for services rendered or performed directly for the account of such tenants or for which a tenant pays directly under an electricity schedule.

7.03 Landlord shall advise Tenant by written statement certified to be correct by Landlord or its agent, of increased Operating Costs for any Operating Year. The statement shall delineate the amount of Tenant's Proportionate Share caused by such increase, shall establish the Adjustment Date, and set forth the manner in which the adjustment is computed. Landlord's failure to render a statement with respect to increases in Operating costs for any Operating Year shall not prejudice Landlord's right to thereafter render a statement with respect thereto or with respect to any subsequent Operating Year. Furthermore, nothing herein contained shall restrict Landlord from issuing a statement or from revising an estimate at any time that there is an increase in Operating Costs during any Operating Year or at any time thereafter. The rights and obligations of Landlord and Tenant under the provisions of this Article with respect to any additional rent shall survive the termination of this Lease.

7.04 Tenant and Landlord agree that for all purposes in any way connected with or arising out of this Article, the statement delivered by Landlord pursuant to Section 7.03 shall be binding and conclusive on both parties hereto unless objected to by Tenant in

12

writing within sixty (60) days after receipt thereof. Tenant's objection shall be in writing and specify the respects in which the statement is claimed to be incorrect. Provided that Tenant has made timely payment of Tenant's Proportionate Share of Operating Costs, the Tenant shall have the right to require the production of Landlord's books which relate to these items of cost and the right, within thirty (30) days subsequent to the production of Landlord's books, to deliver notice of disagreement with respect to any item of Operating Costs. Any review of Landlord's books shall be (a) performed by a reputable firm of certified public accountants engaged by Tenant on a fee-paid basis (as opposed to a contingency fee basis), who shall execute and deliver to Landlord an undertaking, whereby such accounting firm: (i) covenants not to disclose to any person or entity (other than Tenant) any information received by or made available to such accounting firm in connection with the review, (ii) agrees not to solicit or accept engagement by other tenants in the Building for the purpose of performing a review of Landlord's books on their behalf, and (iii) agrees that while it may review the applicable books at Landlord's record-keeping office, no copies may be made, nor may any such books be removed from such record-keeping office; (b) performed during Business Hours (as defined in Section 17.01 hereof), upon prior appointment with Landlord at Landlord's record-keeping office; and (c) shall be completed within one hundred eighty (180) days following Tenant having received the statement delivered by Landlord pursuant to Section 7.03.

7.05 Anything to the contrary herein notwithstanding, Landlord shall have the right at any time to calculate Operating Costs for the Building and the other buildings comprising the Landmark Square Complex (the "Complex") separately rather than together and to collect from Tenant as additional rent Tenant's Proportionate Share of the excess of the Building's Operating Costs for the Boyer over the Building's Operating Costs for the Base Year; in such event Landlord shall give Tenant at least thirty (30) days advance notice of the said manner of calculation ("Landlord's Change Notice"). At the time said adjustments are made in calculating Operating Costs for the Operating Years (or portions thereof) falling after the effective date of Landlord's Change Notice, appropriate adjustment shall be made in Operating Costs for the Base Year.

7.06 Notwithstanding that in certain provisions of this Lease, it is specified that Landlord shall perform certain obligations and services as an Operating Cost, whereas the Lease is silent in regards to other obligations and services, all obligations and services to be performed by Landlord shall be included as an Operating Cost to the extent that they fall within the definition of Operating Costs in Section 7.02(f) hereof. The costs of any services included in Operating Costs, which services are performed by subsidiaries or affiliates of Landlord or Landlord's agents shall be competitive in price for comparable contracts and transactions with unaffiliated entities for the performance of such services in comparable buildings.

ARTICLE 8 SUBORDINATION, ATTORNMENT, NOTICE TO LESSOR AND MORTGAGEES

8.01 This Lease, and all rights of Tenant hereunder, are and shall be subject and subordinate in all respects to all present and future ground leases,

over-riding leases and underlying leases and/or grants of term of the land and/or the Building or the portion thereof in which the Demised Premises are located in whole or in part now or hereafter existing and to all mortgages and building loan agreements, including leasehold mortgages and building loan agreements, which may now or hereafter affect the land and/or the Building and/or any of such leases, whether or not such mortgages shall also cover other lands and/or buildings, to each and every advance made or hereafter to be made under such mortgages, and to all renewals, modifications, replacements and extensions of such leases and such mortgages and spreaders, consolidations and correlations of such mortgages. This Section shall be self-operative and no further instrument of subordination shall be required. In confirmation of such subordination. Tenant shall promptly execute and deliver an instrument, in recordable form, if required, that Landlord, the lessor of any such lease or the holder of any such mortgage or any of their respective successors in interest may request to evidence such subordination, and Tenant hereby constitutes and appoints Landlord attorney-in-fact for Tenant to execute any such instrument for and on behalf of Tenant. The leases to which this Lease is, at the time referred to, subject and subordinate pursuant to this Article are hereinafter sometimes called "superior leases" and the mortgages to which this Lease is at the time referred to, subject and subordinate are hereinafter sometimes called "superior mortgages" and the lessor of a superior lease or its successor in interest at the time referred to is hereinafter sometimes called a "lessor".

8.02 Landlord hereby notifies Tenant that in accordance with the terms of a superior mortgage, this Lease may not be modified or amended so to reduce the rent, shorten the term, or adversely affect in any other respect to any material extent the rights of the Landlord hereunder, or be cancelled or surrendered without the prior written consent of the holder of the superior mortgage in each instance, except that said holder's consent shall not be required to the institution or prosecution of any action or proceedings against Tenant by reason of a default on the part of Tenant under the terms of this Lease.

13

8.03 This Lease shall not terminate or be terminable by Tenant by reason of any termination of the ground lease, by summary proceedings, foreclosure of a superior mortgage, or otherwise. Tenant agrees without further instruments of attornment in each case, to attorn to the lessor under the ground lease, or the mortgagee under the superior mortgage, as the case may be, to waive the provisions of any statute or rule of law now or hereafter in effect which may give or purport to give Tenant any right of election to terminate this Lease or to surrender possession of the Demised Premises in the event the ground lease is terminated or a superior mortgage is foreclosed, and that unless and until the lessor, or mortgagee, as the case may be, shall elect to terminate this Lease, this Lease shall not be affected in any way whatsoever by any such proceeding or termination, and Tenant shall take no steps to terminate this Lease without giving written notice to the lessor under the ground lease, or mortgagee under a superior mortgage, and a reasonable opportunity to cure (without such lessor or mortgagee being obligated to cure), any default on the part of the Landlord under this Lease.

8.04 If any act or omission by Landlord would give Tenant the right, immediately or after lapse of time, to cancel or terminate this Lease or to claim a partial or total eviction, Tenant will not exercise any such right until (a) it has given written notice of such act or omission to each holder of a superior mortgage and to each holder of a superior lease, whose name and address shall have previously been furnished to Tenant, by delivering notice of such act or omission addressed to each such party at its last address so furnished and (b) a period for remedving such act or omission shall have elapsed following such giving of notice and following the time when such senior interest holder shall have become entitled under such senior interest, as the case may be, to remedy the same (which period shall be equal to, and immediately follow, the full period to which Landlord would be entitled under this Lease to effect such remedy) provided such senior interest holder shall, with reasonable diligence, give Tenant notice of its intention to remedy such act or omission and shall commence and continue to act upon such intention. Notwithstanding the foregoing, nothing shall impose any obligation on such holder actually to remedy such act or omission.

8.05 Landlord shall use commercially reasonable efforts to deliver a subordination, nondisturbance and attornment agreement ("Nondisturbance Agreement") from Landlord's current and future mortgagees on such mortgagee's standard form. Tenant shall be responsible for, and shall pay to Landlord upon

Landlord's request therefor, all reasonable costs incurred by Landlord in Landlord's effort to provide such agreement. In the event that a mortgagee shall be unwilling to enter into a Nondisturbance Agreement as aforesaid, this Lease shall remain in full force and effect and the obligations of Tenant shall not in any manner be affected.

ARTICLE 9 QUIET ENJOYMENT

9.01 Landlord covenants that if, and so long as, Tenant is not in default of the terms of this Lease beyond the applicable notice and cure period provided herein for the cure thereof (if any), Tenant shall quietly enjoy the premises without hindrance or molestation by Landlord or by any other person lawfully claiming the same, subject to the covenants, agreements, terms, provisions and conditions of this Lease.

ARTICLE 10 ASSIGNMENT, MORTGAGING, SUBLETTING

10.01 Neither this Lease, nor the term and estate hereby granted, nor any part hereof or thereof, nor the interest of Tenant in any sublease or the rentals thereunder, shall be assigned, mortgaged, pledged, encumbered or otherwise transferred by Tenant by operation of law or otherwise, and neither the Demised Premises, nor any part thereof, shall be encumbered in any manner by reason of any act or omission on the part of Tenant or anyone claiming under or through Tenant, nor shall be sublet to be used or occupied or permitted to be used or occupied, or utilized for desk space or for mailing privileges, by anyone other than Tenant or for any purpose other than as permitted by this Lease, without the prior written consent of Landlord in every case, except as expressly otherwise provided in this Article.

Furthermore, no assignment shall be binding on Landlord unless the assignee shall execute, acknowledge and deliver to Landlord (a) a duplicate original instrument of assignment in form and substance reasonably satisfactory to Landlord, duly executed by Tenant, and (b) an agreement, in form and substance reasonably satisfactory to Landlord, duly executed by the assignee, whereby the assignee shall unconditionally assume in accordance with the terms and conditions of this Lease observance and performance of, and agree to be bound by all of the terms, covenants and conditions of this Lease on Tenant's part to be observed or performed, including, without limitation, the provisions of this Article with respect to all future assignments; but the failure or refusal of the assignee to execute or deliver such an agreement shall not release the assignee from its liability for the obligations of Tenant hereunder assumed by acceptance of the assignment of this Lease.

14

10.02 If this Lease is assigned, whether or not in violation of the provisions of this Lease, Landlord may collect rent from the assignee. If the Demised Premises or any part thereof is sublet or is used or occupied by anybody other than Tenant, whether or not in violation of this Lease, Landlord may, after default by Tenant, and expiration of Tenant's time to cure such default, collect rent from the subtenant or occupant. In either event, Landlord may apply the net amount collected to the rents herein reserved, but no such assignment, subletting, occupancy or collection shall be deemed a waiver of any of the provisions of Section 10.01, or the acceptance of the assignee, subtenant or occupant as tenant, or a release of Tenant from the further performance by Tenant of Tenant's obligations under this Lease. The consent by Landlord to assignment, mortgaging, subletting or use or occupancy by others shall not in any way be considered to relieve Tenant from obtaining the express written consent of Landlord to any other or further assignment, mortgaging or subletting or use or occupancy by others not expressly permitted by this Article. References in this Lease to use or occupancy by others, that is anyone other than Tenant, shall not be construed as limited to subtenants and those claiming under or through sub-tenants but as including also licensees and others claiming under or through Tenant, immediately or remotely.

10.03 Tenant, upon prior written notice to Landlord, but without Landlord's consent, may permit any corporations or other business entities which control, are controlled by, or are under common control with Tenant (herein called "related corporations") to use the whole or part of the Demised Premises for any

of the purposes permitted to Tenant, subject however to compliance with Tenant's obligations under this Lease. Such use shall not be deemed to vest in any such related corporation any right or interest in this Lease or in the Demised Premises, nor shall such use release, relieve, discharge or modify any of Tenant's obligations hereunder.

10.04 Tenant, upon prior written notice to Landlord, but without Landlord's consent, may assign or transfer its entire interest in the Lease and the leasehold estate hereby created or sublet the whole of the Demised Premises on one or more occasions to a "wholly owned subsidiary" or "affiliate" of Tenant or to a "successor corporation" of Tenant, as such terms are hereinafter defined, provided that Tenant shall not then be in default in any of the terms. covenants, conditions and agreements of this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), including but not limited to the payment of the fixed rent or additional rent payable by Tenant hereunder. A "wholly owned subsidiary" of Tenant shall mean any corporation all of whose outstanding voting stock shall at the time be owned, directly or indirectly, by Tenant or by one or more of its wholly owned subsidiaries. An "affiliate" of Tenant shall mean any corporation or other entity or business form which directly or indirectly controls or is controlled by or is under common control with Tenant. For purposes of this definition, "control" (including "controlling," "controlled by" and "under common control with") as used with respect to any corporation or other entity or business form, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such corporation or other entity or business form, whether through the ownership of voting securities, or by contract or otherwise. A "successor corporation" as used in this Article, shall mean (i) a corporation into which or with which Tenant, its corporate successors or assigns, is merged or consolidated, in accordance with applicable statutory provisions for the merger or consolidation of corporations, provided that by operation of law or by effective provisions contained in the instruments of merger or consolidation the liabilities of the corporations participating in such merger or consolidation are assumed by the corporation surviving such merger or consolidation, or (ii) a corporation or other entity or business form acquiring this Lease and the term hereby demised, the goodwill and all or substantially all of the other property and assets of Tenant, its corporate successors or assigns, and assuming all or substantially all of the liabilities of Tenant, its successors and assigns, or (iii) any corporate successor to a successor corporation or other entity or business form becoming such by either of the methods described in Clauses (i) and (ii); provided that, prior to such merger or consolidation, or such acquisition and assumption as the case may be, Tenant delivers to Landlord (a) a fairness opinion from a reputable investment bank stating that the price being paid for the assets or the entity being acquired is fair, and (b) proof acceptable to Landlord in Landlord's reasonable discretion that at such time Tenant has free and clear cash on hand equal to at least three (3) times Tenant's cash requirements for the trailing twelve (12) month period. The acquisition by Tenant, its corporate successors or assigns, of all or substantially all of the assets, together with the assumption of all or substantially all of the obligations and liabilities of any corporation, shall be deemed to be a merger of such corporation into Tenant for the purpose of this Article.

10.05 If Tenant is a corporation, partnership or other business form, and if at any time during the term of this Lease the person or persons who, as of the date this Lease is executed by Tenant, own or owns a majority of (a) such corporation's voting stock (as hereinafter defined), (b) the interests in such partnership, or (c) the equitable ownership interests in such other business form, as the case may be, cease or ceases to own a majority of such voting stock, partnership interest or other equitable ownership interest, as the case may be, then the occurrence of any such event shall be deemed to be an assignment of this Lease with respect to which the Landlord's prior written consent shall be required. This Section 10.05 shall not apply to the public trading of shares of stock of Tenant on a National Exchange (as defined under federal securities laws). This Section 10.05 shall not apply to Kanders & Company, Inc. so long as Clarus Corporation (as it exists on the date hereof as opposed to a "successor corporation" as defined in Section 10.04) is one of the entities comprising Tenant hereunder.

10.06 (A) With respect to each and every sublease or subletting authorized by the provisions of this Article:

(I) No subletting shall be for a term ending later than one day prior to the Expiration Date of this Lease, and that part, if any, of the proposed term of any sublease or any renewal or extension thereof which shall extend beyond a date one day prior to the Expiration Date or the earlier termination of the term of this Lease, is hereby deemed to be a nullity.

(II) Upon the execution of any such sublease as may be authorized by this Article, Tenant shall promptly deliver to Landlord a copy of each such sublease.

(B) Notwithstanding anything to the contrary set forth herein, within thirty (30) days of Landlord's receipt of Tenant's notice that Tenant desires to sublease (i) all of the Demised Premises for any length of time, or (ii) any portion of the Demised Premises for all or substantially all of the balance of the then current term of this Lease (i.e. resulting in less than one (1) year remaining in the term of this Lease upon the expiration of such sublease), except with respect to a sublease within the scope of Article 10.04 herein, Landlord may elect to terminate this Lease, as to the entire Demised Premises or as to the relevant portion (at Landlord's option) by delivering notice of such election to Tenant within such thirty (30) day period; provided, however, if Landlord elects to terminate this Lease, Tenant may withdraw its request for consent to such sublease within five (5) days after delivery of Landlord's termination notice, in which case Landlord's election to terminate shall be void and of no force and effect. If Landlord shall have elected to terminate this Lease pursuant to this Section 10.06(b) and Tenant has not withdrawn its request as aforesaid, the Term of this Lease shall cease and come to an end on that day with the same force and effect as though that were the original date set forth as the Expiration Date, and Tenant shall deliver broom clean possession of the Demised Premises or the applicable portion thereof to Landlord, in accordance with the terms of this Lease. Thereafter, neither party shall have any obligations to the other hereunder, except for any fixed rent or additional rent due and owing to the Landlord up to and including the termination of this Lease, any obligations that expressly survive the termination or expiration of this Lease, and as the parties hereto may have agreed otherwise in this Lease or by separate writing. If Landlord fails to exercise such option within said thirty (30) days and Tenant fails to complete such sublease with a third party (as hereinafter provided within ninety (90) days thereafter), Tenant shall again comply with all the conditions of this Section, as if the notice and option hereinabove referred to had not been given and received.

(C) In the event Landlord does not exercise its option to terminate this Lease in whole or in part, as the case may be, Landlord covenants not to unreasonably withhold its consent (which must be in writing) to such subletting or assignment, provided, however, that Landlord shall not, in any event, be obligated to consent to any such proposed subletting or assignment unless:

(I) At least thirty (30) days prior to the proposed subletting or assignment, Tenant furnishes Landlord with the name and business address of the proposed subtenant or assignee, a counterpart of the proposed subleasing agreement or assignment, and satisfactory information with respect to the nature and character of the business of the proposed subtenant or assignee together with current financial information and references reasonably satisfactory to Landlord;

(II) in the reasonable judgment of Landlord the proposed subtenant or assignee is of a character and engaged in a business such as are in keeping with the standards of Landlord in those respects for the Building;

(III) the purposes for which the proposed subtenant or assignee intends to use the portion of the Demised Premises sublet to it are uses expressly permitted by and not expressly prohibited by this Lease;

(IV) in the case of a subletting, Tenant shall not have listed or publicly advertised the rental rate less than the fixed rent and additional rent (pursuant to Articles 6 and 7) then payable hereunder for such space; but the provisions of this subsection, however, shall not be deemed to prohibit Tenant from negotiating a sublease at a lesser rate of rent and consummating the same insofar as it may be permitted under the provisions of this Article;

(V) the proposed subtenant or assignee is not an existing occupant of the Building or any other building owned by Landlord or any affiliate of Landlord in Connecticut;
(VI) the proposed subtenant or assignee is not then negotiating with Landlord for space in the Building or in any other building owned by Landlord or any affiliate of Landlord in Connecticut; and

(VII) in the case of a subletting of a portion of the Demised Premises, the portion so sublet shall be regular in

16

shape and suitable for normal renting purposes.

(D) Each subletting pursuant to this Article shall be subject to all the covenants, agreements, terms, provisions and conditions contained in this Lease. Tenant covenants and agrees that notwithstanding any such subletting to Landlord or any such subletting to any other subtenant and/or acceptance of rent or additional rent by Landlord from any subtenant, Tenant shall and will remain fully liable for the payment of the fixed rent and additional rent due and to become due hereunder and for the performance of all the covenants, agreements, terms, provisions and conditions contained in this Lease on the part of Tenant to be performed and all acts and omissions of any licensee or subtenant or anyone claiming under or through any subtenant which shall be in violation of any of the obligations of this Lease and any such violation shall be deemed to be a violation by Tenant. Tenant further covenants and agrees that notwithstanding any such subletting, no other and further subletting of the Demised Premises or any part thereof shall or will be made except upon compliance with and subject to the provisions of this Article.

(E) Tenant shall reimburse Landlord for any reasonable costs incurred by Landlord to review any proposed subletting or assignment including Landlord's attorneys' fees and disbursements.

(F) Tenant shall reimburse Landlord for Landlord's costs incurred in connection with plan review or plan approval should Tenant propose alterations to the Demised Premises to make same suitable for the occupancy by Tenant's subtenant or assignee.

(G) Notwithstanding anything to the contrary contained in this Lease, no sublease shall be permitted (i) during the first twelve (12) full calendar months following the Rent Commencement Date, or (ii) at any time at a sublease rental rate lower than the then current fixed rent payable by Tenant pursuant to this Lease. This Section 10.06(g) shall not apply to subleases under Section 10.04.

10.07 Landlord's approval is required for the use of any name other than Tenant's name for identification on any signs or directory listings in the Building.

The listing of any name other than that of Tenant, even though approved by Landlord, shall not:

(A) Constitute a waiver of Landlord's right to withhold consent to any sublet or assignment pursuant to this Article;

(B) Be deemed an implied consent by Landlord to any sublet of the Premises or any portion thereof, to any assignment or transfer of the lease, or to any unauthorized occupancy of the Premises, except in accordance with the express terms of the Lease; or

(C) Operate to vest any right or interest in the Lease or in the Premises.

The use of any such other name or listing as provided above shall constitute a privilege extended by Landlord to Tenant, and shall be revocable at Landlord's will by notice to Tenant; provided, however, Tenant may use the following names during the term of this Lease: Clarus Corporation, Kanders & Company, Inc., Warren B. Kanders Foundation, Inc. and Armor Holdings, Inc. 11.01 Tenant shall promptly notify Landlord of any written notice it receives of the violation of any law or requirements of any Federal, State, Municipal or other public authority, and at its expense Tenant shall comply with all laws and requirements of such public authorities which shall, with respect to the Building or the Demised Premises or the use and occupation thereof or the abatement of any nuisance, impose any violation, order or duty on Landlord or Tenant, arising from (i) Tenant's use of the Demised Premises, (ii) the manner of conduct of Tenant's business or operation of its installations, equipment or other property therein, (iii) any cause or condition created by or at the instance of Tenant, or (iv) breach of any of Tenant's obligations hereunder.

ARTICLE 12 INSURANCE

12.01 Tenant shall not do, or permit anything to be done, or keep or permit anything to be kept in the Demised Premises which

	_
1	7
н	1

would increase the fire or other casualty insurance rate on the Building or the property therein over the rate which would otherwise then be in effect (unless Tenant pays the resulting increased amount of premium as provided in Section 12.02) or which would result in insurance companies of good standing refusing to insure the Building or any of such property in amounts and at normal rates reasonably satisfactory to Landlord.

12.02 If, by reason of any act or omission on the part of Tenant, the rate of fire insurance with extended coverage on the Building or equipment or other property of Landlord shall be higher than it otherwise would be, Tenant shall reimburse Landlord, on demand, for that part of the premiums for fire insurance and extended coverage paid by Landlord because of such act or omission on the part of Tenant, which sum shall be deemed to be additional rent and collectible as such.

12.03 In the event that any dispute should arise between Tenant and Landlord concerning rates, a schedule or make up of rates for the Building or the Demised Premises, as the case may be, issued by a Fire Insurance Rating Organization or other similar body making rates for fire insurance and extended coverage for the premises concerned, shall be presumptive evidence of the facts therein stated and of the several items and charges in the fire insurance rates with extended coverage then applicable to such premises.

12.04 Tenant shall obtain and keep in full force and effect during the term of this Lease at its own cost and expense (a) General Comprehensive Commercial Liability insurance on an occurrence basis with minimum limits of liability in an amount of not less than \$3,000,000.00 combined single limit coverage for injury, death, damage to property, water damage and sprinkler leakage including legal liability arising out of any one occurrence, which insurance shall contain contractual liability insurance covering the matters set forth in Article 21, which coverage shall be subject to periodic increase as reasonably required by Landlord, protecting Tenant as the insured and Landlord and any other parties whose names have been provided by Landlord to Tenant from time to time as additional insureds (in an endorsement form satisfactory to Landlord); (b) Fire and Extended Coverage Insurance on Tenant's property, insuring against damage by fire, and such other risks and hazards as are insurable under present and future standard forms of fire and extended coverage insurance policies, to Tenant's property for the full insurable value thereof; and (c) any other insurance required by law.

12.05 All such policies of insurance shall be issued by a financially responsible company or companies, authorized to issue such policy or policies, and licensed to do business in the State of Connecticut, which shall be reasonably satisfactory to Landlord, and shall contain endorsements providing as follows: (a) that any such insurance shall not be subject to cancellation, termination, reduction or change, except with thirty (30) days' prior written notice sent by registered or certified mail to Landlord by the insurance

company; and (b) that Landlord shall not be liable for any damage by fire or other casualty covered by such insurance, regardless of the cause, it being understood that Tenant shall look solely to its insurer or insurers for reimbursement. Landlord and Tenant waive their right to recover damages against each other for any reason whatsoever to the extent the damaged party recovers from its insurance carrier. Any insurance policy procured by Tenant which does not name the Landlord as an additional insured shall contain an express waiver of any right of subrogation by the insurance company against Landlord. All liability and property damage policies shall contain an endorsement that Landlord, although named as an additional insured, shall nevertheless be entitled to recover under said policies for any loss or damage occasioned to it, its servants, agents and employees. The original policy or policies together with satisfactory evidence of payment of the premium thereof, shall be delivered to Landlord on or before the commencement of Tenant's Initial Work, and upon renewals of such policies, not less than thirty (30) days prior to the expiration of the term of any such coverage. The minimum limits of any insurance coverage required herein to be carried by Tenant shall not limit Tenant's liability under Article 21 hereof.

12.06 In the event that Tenant at any time or times shall fail to obtain or maintain in full force and effect any or all of the insurance polices and coverage required of it hereunder, or should Tenant violate any of the provisions of Section 12.05 herein, Landlord, at its election after ten (10) days written notice to Tenant, and as agent for Tenant, may obtain such insurance or coverage, or additional insurance or coverage as the case may be, pay the premiums thereon or take such other steps as may be necessary to meet the requirements of this Article and thereafter, upon demand, obtain reimbursement of the costs so expended from Tenant. The failure of Landlord to obtain evidence of the required insurance coverage shall not relieve Tenant of its obligations under this Article.

12.07 If Tenant shall decide not to insure for (or to self-insure or co-insure part of any loss for) business interruption and/or if Tenant shall at any time fail to maintain property insurance as, and to the extent, required of it hereunder, Tenant hereby releases Landlord from all loss or damage which could have been covered by such insurance if Tenant had maintained such insurance. In no event, however, shall the foregoing clause increase the liability Landlord may otherwise have under this Lease for such loss or damage.

18

ARTICLE 13 RULES AND REGULATIONS

13.01 Tenant and its employees and agents shall faithfully observe and comply with the rules and regulations set forth in the attached Exhibit "C" (the "Rules and Regulations"), and such reasonable changes therein (whether by modification, elimination or addition) as Landlord at any time or times hereafter may make and communicate in writing to Tenant, which do not unreasonably affect the conduct of Tenant's business in the Demised Premises; provided, however, that in case of any conflict or inconsistency between the provisions of this Lease and any Rules and Regulations enacted subsequent to the date of this Lease the provisions of this Lease shall control. Landlord agrees not to discriminate against Tenant in the implementation and enforcement of such Rules and Regulations.

13.02 Notwithstanding anything to the contrary in any of the Rules and Regulations set forth in Exhibit "C":

(A) Tenant may bring into and keep in the Demised Premises such small quantities of inflammable or combustible objects or materials as are permitted by local law and as are incidental to the use of the Demised Premises for the purposes permitted by Article 2, but this shall not be deemed to relieve Tenant of responsibility to comply with all other obligations of this Lease that may be applicable to or result from the introduction or maintenance of such objects or materials in the Demised Premises, including but not limited to compliance with the provisions of Sections 12.01 and 12.02.

(B) Subject to the provisions of Paragraph 2.02(c), Landlord shall not unreasonably withhold its consent to the installation, maintenance and operations by Tenant in the Demised Premises of data processing machines, office duplicating machines, teletypewriter machines and other business machines and machinery customarily used in offices in the ordinary course of business, provided, however, that Tenant shall comply with all other obligations of this Lease that may be applicable to or result from such installation, maintenance or operation.

(C) Landlord shall not unreasonably withhold from Tenant any approval provided for in the Rules and Regulations.

(D) Whenever Landlord shall claim, by written notice to Tenant, that Tenant is violating any of the provisions of the Rules and Regulations, and Tenant shall in good faith dispute such claim by written notice given to Landlord within ten (10) days after service of Landlord's notice of the violation, the dispute shall be determined by arbitration pursuant to the rules of the American Arbitration Association.

(E) Tenant shall utilize only security and cleaning services approved in writing by Landlord.

ARTICLE 14 ALTERATIONS AND TENANT'S PROPERTY

14.01 Tenant shall make no alterations, decorations, installations, additions or improvements (hereinafter collectively referred to as "Tenant's Work") in or to the Demised Premises. Tenant may make written request to Landlord that certain Tenant's Work be performed in the Demised Premises, but all such Tenant's Work shall be performed, if at all, (i) in the sole and absolute discretion of Landlord, (ii) by Landlord or its designee (except for Tenant's Initial Work, which may be performed by Tenant's contractors and subcontractors subject to all of the provisions of Section 14.02 hereof), and (iii) at the sole cost and expense of Tenant. Any Tenant's Work to be performed in, on or to the Demised Premises shall be performed by Landlord (which term as used in this Article 14.01 shall be deemed to include Landlord and/or Landlord's contractor) (except for Tenant's Initial Work, which may be performed by Tenant's contractors and subcontractors subject to all of the provisions of Section 14.02 hereof), and Tenant shall pay Landlord for all costs and charges for such Tenant's Work (including, without limitation, the cost of any drawings, plans, layouts and/or specifications prepared by Landlord or its consultants with respect to such Tenant's Work). Notwithstanding the foregoing, Landlord shall not unreasonably withhold its consent to the performance of any Minor Alteration (as hereinafter defined). The term "Minor Alteration", as used herein, means any Alteration which will not, in the reasonable opinion of Landlord (i) be structural in nature; (ii) affect the exterior or any structural portions or components of the Building; (iii) be visible from outside of the Demised Premises; (iv) affect the usage or proper functioning of any of the Building systems (including, without limitation, the heating, ventilation, air conditioning, plumbing, electrical, sprinkler or security systems serving the Building); (v) jeopardize health safety or life safety; (vi) require a change to the certificate of occupancy for the Building; (vii) require the issuance of a building permit or other authorization by any governmental or quasi-governmental entity exercising jurisdiction over the Building; (viii) require the consent of any mortgagee or ground lessor of the Building and/or the Real Property; (ix) cause any previously non-mandatory legal requirement to become a mandatory legal

19

requirement with regard to the Building (including, without limitation, any such legal requirement set forth in the Americans with Disabilities Act); or (x) have a cost of completion in excess of \$50,000.00. However, Landlord's consent shall not be required for any Decorative Alterations to be made in the Demised Premises. The parties acknowledge and agree that the term "Decorative Alterations" shall only include those Minor Alterations for which a reasonable and prudent tenant would not customarily engage a third party contractor or professional (e.g., the hanging of standard office artwork or the relocation or placement of standard, moveable office furniture). The parties further acknowledge and agree that the term "Decorative Alterations" shall specifically exclude all wall covering and floor covering work.

14.02 With respect to any Alteration to which Landlord consents under Section 14.01 which will be performed by a contractor other than Landlord or Landlord's

contractor and with respect to Decorative Alterations (any of such Alterations, the "Permitted Tenant's Work"), the following provisions shall apply:

(A) All Permitted Tenant's Work done by Tenant shall at all times comply with (i) laws, rules, orders and regulations of governmental authorities having jurisdiction thereof, and (ii) rules and regulations of the Landlord attached as Exhibit C.

(B) With respect to all Permitted Tenant's Work, architectural and engineering plans and specifications prepared by and at the expense of Tenant shall be submitted to Landlord for its prior written approval in accordance with the following requirements:

(I) With respect to any Permitted Tenant's Work to be performed by Tenant pursuant to this Lease, Tenant shall, at its expense, furnish Landlord with complete architectural, mechanical and electrical construction documents for work to be performed by Tenant (the "Tenant's Plans"). All of the Tenant's Plans shall: (A) be compatible with the Landlord's building plans, (B) comply with all applicable laws and the rules, regulations, requirements and orders of any and all governmental agencies, departments or bureaus having jurisdiction, and (C) be fully detailed, including locations and complete dimensions;

(II) Tenant's Plans shall be subject to the approval of Landlord which shall not be unreasonably withheld or delayed;

(III) Tenant shall, at Tenant's expense, (A) cause Tenant's Plans to be filed with the governmental agencies having jurisdiction thereover, (B) obtain when necessary all governmental permits, licenses and authorizations required for the work to be done in connection therewith, and (C) obtain all necessary certificates of occupancy, both temporary and permanent. Landlord shall execute such documents as may be reasonably required in connection with the foregoing and Landlord shall otherwise cooperate with Tenant in connection with obtaining the foregoing, but without any expense to Landlord. Tenant shall make no amendments or additions to Tenant's Plans without the prior written consent of Landlord in each instance;

(IV) No work shall commence in the Premises until (A) Tenant has procured all necessary permits therefor and has delivered copies of same to Landlord, (B) Tenant has procured a paid builder's risk insurance policy naming Landlord and its managing agent as additional insureds and has delivered to Landlord a certificate of insurance evidencing such policy, and (C) Tenant or its contractor has procured a workmen's compensation insurance policy covering the activities of all persons working at the Premises naming Landlord and its managing agent as additional insureds and has delivered to Landlord a certificate of insurance evidencing such policy;

(V) Tenant may use any licensed architect or engineer to prepare its plans and to file for permits. However, all such plans and permit applications shall be subject to review, revision and approval by Landlord or its architect;

(VI) Tenant, at its expense, shall perform all work in connection with all Permitted Tenant's Work, in accordance with Tenant's Plans, and such work shall be subject to Landlord's supervisory fee charge of 5% of the cost thereof (with a cap of \$25,000.00 with respect to Tenant's Initial Work). In receiving such fee, Landlord assumes no responsibility for the quality or manner (including, without limitation, the means, methods and/or techniques) in which such work has been performed; and

(VII) Tenant agrees that it will not, either directly or indirectly, use any contractors and/or labor and/or materials if the use of such contractors and/or labor and/or materials would or will, in Landlord's reasonable opinion, create any difficulty with other contractors and/or labor engaged by Tenant or Landlord or others in the construction, maintenance or operation of the Building or any part thereof.

(C) Tenant's Permitted Tenant's Work shall be subject to the following additional conditions: (i) the Permitted Tenant's

Work will not result in a violation of, or require a change in, any Certificate of Occupancy applicable to the Premises or the Building; (ii) the outside appearance, character or use of the Building shall not be affected; (iii) no part of the Building outside of the Premises shall be physically affected; and (iv) the proper functioning of any mechanical and electrical system of the Building shall not be affected.

(D) Tenant shall defend, indemnify and save harmless Landlord against any and all mechanics' and other liens filed in connection with its Permitted Tenant's Work, repairs or installations, including the liens of any conditional sales of, or chattel mortgages upon, any materials, fixtures or articles so installed in and constituting part of the Premises and against any loss, cost, liability, claim, damage and expense, including reasonable counsel fees, penalties and fines incurred in connection with any such lien, conditional sale or chattel mortgage or any action or proceeding brought thereon. As a condition precedent to Landlord's consent to the making by Tenant of Permitted Tenant's Work, Tenant agrees to obtain and deliver to Landlord at the completion of the work, written and unconditional waivers of mechanics' liens for all work, labor and services to be performed and materials to be furnished, signed by all contractors, subcontractors, materialmen and laborers to become involved in such work.

(E) Tenant, at its expense, shall procure the satisfaction or discharge of all such liens within twenty (20) days of the filing of such lien against the Premises or the Building. If Tenant shall fail to cause such lien to be discharged within the aforesaid period, then, in addition to any other right or remedy, Landlord may, but shall not be obligated to, discharge the same either by paying the amount claimed to be due or by procuring the discharge of such lien by deposit or by bonding proceedings, and in any such event Landlord shall be entitled, if Landlord so elects, to compel the prosecution of an action for the foreclosure of such lien by the lienor and to pay the amount of the judgment in favor of the lienor with interest, costs and allowances. Any amount so paid by Landlord, and all costs and expenses incurred by Landlord in connection therewith, together with interest thereon at the maximum rate permitted by law from the respective dates of Landlord's making of the payments or incurring of the cost and expense, shall constitute additional rent and shall be paid on demand.

(F) Nothing in this Lease contained shall be construed in any way as constituting the consent or request of Landlord, expressed or implied, to any contractor, subcontractor, laborer or materialman for the performance of any labor or the furnishing of any material for any improvement, alteration or repair of the Premises, nor as giving any right or authority to contract for the rendering of any services or the furnishing of any materials that would give rise to the filing of any mechanics' liens against the Premises.

(G) Permitted Tenant's Work shall be performed in compliance with all applicable requirements of insurance bodies, and in good and first class workmanlike manner, using materials and equipment at least equal in quality and class to the original installations of the Building. Permitted Tenant's Work shall be performed in such a manner as not to interfere with the occupancy of any other tenant in the Building, nor delay or impose any additional expense upon Landlord in the construction, maintenance or operation of the Building. In addition to workmen's compensation insurance as described above, during the performance of Permitted Tenant's Work, Tenant, at its expense, shall carry, or cause to be carried, general liability insurance for any occurrence in or about the Building, of which Landlord and its managing agent shall be named as additional insureds, in such limits as Landlord may reasonably prescribe, with insurers reasonably satisfactory to Landlord. Tenant shall furnish Landlord with reasonably satisfactory evidence that all such insurance is in effect at or before the commencement of Permitted Tenant's Work and, on request, at reasonable intervals thereafter during the continuance of Permitted Tenant's Work. No Permitted Tenant's Work shall involve the removal of any fixtures, equipment or other property in the Demised Premises which are not Tenant's property, unless Landlord's prior written consent is first obtained and unless such fixtures, equipment or other property shall be promptly replaced at Tenant's expense and free of superior title, liens and claims, with fixtures. equipment or other property (as the case may be) of like utility and at least equal value (which replaced fixture, equipment or other property shall thereupon

become the property of Landlord), unless Landlord shall otherwise expressly consent in writing.

(H) Tenant, at its expense, and with diligence and dispatch, shall procure the cancellation or discharge of all notices of violation arising from or otherwise connected with Permitted Tenant's Work which shall be issued by any public authority having or asserting jurisdiction.

(I) Notwithstanding anything to the contrary contained hereinabove, Landlord reserves the right to deny Tenant and/or its contractor access to the Demised Premises and/or to request Tenant to withdraw therefrom and cease all work being performed by it or on its behalf by any person, firm or corporation other than Landlord, if Landlord shall, in its sole judgment, to be exercised in good faith, determine that the commencement and/or the continuance of the Permitted Tenant's Work shall interfere with, delay, hamper or prevent Landlord from proceeding with the completion of any work then being or to be performed by Landlord in the Building or the Demised Premises at the earliest possible date.

21

(J) Tenant agrees that should Tenant enter upon the Demised Premises for the purposes of performing any Permitted Tenant's Work, the labor employed by Tenant or anyone performing such work for or on behalf of Tenant shall always be harmonious and compatible with the labor employed by Landlord or any contractors or subcontractors of Landlord. Should a labor dispute arise or result, Landlord may require Tenant to withdraw from such premises until the completion of the work then being or to be performed by Landlord in the Building or Demised Premises.

14.03 All alterations, decorations, installations, additions or improvements upon the Demised Premises made by any party shall at the expiration of the term hereof become the property of the Landlord and be surrendered with the Demised Premises as part thereof at the end of the term. Tenant's special chandeliers, business and trade fixtures, machinery and equipment, whether or not attached to the Demised Premises, which are installed by or for the account of Tenant, and can be removed without permanent structural damage to the Demised Premises or the Building, and all furniture, furnishings and other articles of movable personal property shall be and shall remain Tenant's property and may be removed by it prior to the expiration date of this Lease; provided, however, that if any of Tenant's property is removed, Tenant shall repair or pay the cost of repairing any damage to the Demised Premises resulting from such removal. Any equipment or other property for which Landlord shall have granted any allowance or credit to Tenant shall not be deemed to have been installed by or for the account of Tenant and shall not be considered Tenant's property.

14.04 At or before the Expiration Date, or the date of any earlier termination of this Lease, Tenant at its expense, shall remove from the Demised Premises all of Tenant's property (including telecommunications and data wiring) except such items thereof as Tenant shall have expressly agreed in writing with Landlord were to remain and to become the property of Landlord, and shall repair any damage to the Demised Premises or the Building resulting from such removal. Any other items of Tenant's property (except money, securities and other like valuables) which shall remain in the Demised Premises after the Expiration Date or after a period of fifteen (15) days following an earlier termination date, may, at the option of the Landlord, be deemed to have been abandoned, and in such case either may be retained by Landlord as its property or may be disposed of, without accountability, in such manner as Landlord may see fit at Tenant's expense. If carpet is glued down, then at Landlord's option Tenant may be required to remove the carpet at Tenant's expense.

14.05 In the event Tenant or Tenant's contractor shall enter upon the Demised Premises or any other part of the building, as may be above permitted by Landlord, Tenant agrees to indemnify and save Landlord free and harmless from and against any and all damages, claims, costs and expenses arising from or claimed to arise from any act, or for any other reason whatsoever arising out of said entry or such work.

ARTICLE 15 REPAIRS AND MAINTENANCE

15.01 Tenant shall take good care of the Demised Premises and the fixtures and appurtenances therein, and at its sole cost and expense shall make all repairs thereto, as and when needed to preserve them in good working order and condition. In addition, Tenant, at its expense, shall promptly make all repairs, ordinary or extraordinary, interior or exterior, structural or otherwise, in and about the Demised Premises and the Building as shall be required by reason of (i) the performance or existence of work by Tenant necessary to suit the Demised Premises to Tenant's initial occupancy or Permitted Tenant's Work (ii) the installation, use or operation of Tenant's property in the Demised Premises, (iii) the moving of Tenant's property in or out of the Building, or (iv) the misuse or neglect of Tenant or any of its employees, agents or contractors. Tenant shall not be responsible, and Landlord shall be responsible, for any repairs to the Demised Premises as are required by reason of Landlord's neglect or other fault in the manner of performing any work included in the Work Letter or Permitted Tenant's Work which may be undertaken by Landlord for Tenant's account or as are otherwise required by reason of neglect or other fault of Landlord or its employees, agents or contractors.

15.02 Landlord shall keep and maintain the Building and its fixtures, appurtenances, systems and facilities serving the Demised Premises, in good working order, condition and repair and shall make all structural repairs, interior and exterior, except as indicated in Section 15.01 as and when needed in the Building, except for those repairs for which Tenant is responsible pursuant to any other provisions of this Lease, and subject to all other provisions of this Lease, including but not limited to the provisions of Article 21.

15.03 Except as expressly otherwise provided in this Lease, Landlord shall have no liability to Tenant by reason of any inconvenience, annoyance, interruption or injury to business arising from Landlord or any tenant making any repairs or changes or performing maintenance services, whether or not Landlord is required or permitted by this Lease or by law to make such repairs or

22

changes or to perform such services in or to any portion of the Building or the Demised Premises, or in or to the fixtures, equipment or appurtenances of the Building or the Demised Premises, provided that Landlord shall be reasonably diligent with respect thereto and shall perform such work, except in case of emergency, at times reasonably convenient to Tenant and otherwise in such manner and to the extent practical as will not unreasonably interfere with Tenant's use and occupancy of the Demised Premises.

ARTICLE 16 ELECTRICAL ENERGY

16.01 Tenant shall pay to Landlord such amounts and at such times as is set forth in Exhibit "E" attached hereto and made a part hereof.

16.02 Landlord shall in no way be liable for any failure of or defect in the character or supply of electrical energy supplied to the Demised Premises, except arising from Landlord's failure to pay the electric company for same or Landlord's gross negligence or wilful misconduct.

16.03 Tenant shall at its expense, install all lamps (including, but not limited to, fluorescent), starters and ballasts used in the Demised Premises.

ARTICLE 17 HEAT, VENTILATION AND AIR CONDITIONING

17.01 Landlord shall maintain and operate the heating, ventilation and air conditioning systems in the Building and shall furnish heat, ventilation and air conditioning in the Demised Premises through such systems, during the hours from

8:00 A.M. to 6:00 P.M. on business days (which term is used herein to mean all days except Saturdays, Sundays and the days observed by the Federal or the Connecticut government as legal holidays) ("Business Hours"). If Tenant shall require ventilating and air conditioning service or heating service at any other time (hereinafter called "after hours"), Landlord shall furnish after hours ventilating and air conditioning service or heating service upon reasonable advance notice from Tenant, and Tenant shall pay Landlord's then established charges therefor on Landlord's demand (which charges are currently \$33.00 per hour, subject to future change). Such charges shall not exceed 121% of Landlord's actual cost of labor, utilities and equipment depreciation used in providing such after hours air conditioning or heating service. If any of the other tenants of the Building shall request and receive after hours heating or air conditioning service pursuant to Landlord's obligation to provide the same to them, at the same time as Tenant, only that equitably pro-rated portion of such labor and utilities costs as shall be incurred for such common service shall be charged to Tenant.

17.02 Landlord will not be responsible for the failure of the air conditioning system to meet industry-standard performance specifications if such failure results from the occupancy of the Demised Premises with more than one person for each 150 square feet in any room or area within the Demised Premises, or if the Tenant installs and operates machines and appliances, the installed electrical load of which when combined with the load of all lighting fixtures exceeds seven watts per square foot of floor area in any one room or other area. If due to use of the Demised Premises in a manner exceeding the aforementioned occupancy and electrical load criteria, or due to rearrangement of partitioning after the initial preparation of the Demised Premises, interference with normal operation of the air conditioning in the Demised Premises results, necessitating changes in the air conditioning system servicing the Demised Premises, such changes shall be made by Landlord upon written notice to Tenant at Tenant's sole cost and expense. Tenant agrees to lower and close window coverings when necessary because of the sun's position whenever the air conditioning system is in operation, and Tenant agrees at all times to cooperate fully with Landlord and to abide by all the regulations and requirements which Landlord may prescribe for the proper functioning and protection of the air conditioning system. Landlord, throughout the term of this Lease, shall have free and unrestricted access to any and all air conditioning facilities in the Demised Premises.

17.03 In the event that Tenant requires supplemental heating, ventilation and air conditioning in to service the Demised Premises (a "Supplemental HVAC System"), the provisions of Article 14 shall be applicable and, without limiting the generality of Article 14: (a) the type of Supplemental HVAC System, its location and the method of installation shall be subject to Landlord's reasonable approval; (b) all costs of installation and removal of the Supplemental HVAC System at the end of the term shall be payable by Tenant; (c) Tenant shall at all times after the installation of the Supplemental HVAC System maintain a service contract for same with a contractor reasonably acceptable to Landlord; and (d) electrical consumption by the Supplemental HVAC System will be measured by Tenant's Check Meter (as defined in Exhibit E).

23

ARTICLE 18 LANDLORD'S OTHER SERVICES

18.01 Landlord, at its expense, shall provide public elevator service, by elevators serving the floors on which the Demised Premises are situated as specified and shown in the Building plans and specifications, during Business Hours, and shall have at least one passenger elevator subject to call at all other times. The elevators, or any of them, may be operated by automatic control and/or by manual control, as Landlord shall determine at any time or from time to time. Landlord shall not be obligated to furnish an operator for any automatic elevator and shall have no liability to Tenant for discontinuing the service of any operator theretofore furnished. If Tenant shall require Saturday or after hours service of elevators or of the loading area in the Building under such circumstances as in Landlord's reasonable judgment, will require service or attention by Landlord's personnel, Tenant shall pay Landlord, on demand, a reasonable charge attributable to such service or attention.

18.02 Landlord, at its expense, shall cause the Demised Premises, including the exterior and the interior of the windows thereof (subject to Tenant maintaining unrestricted access to such windows), to be cleaned in accordance with the

standards set forth in the attached Exhibit "B" (the "Cleaning Schedule"). Tenant shall pay to Landlord on demand the costs incurred by Landlord for (a) cleaning work in the Demised Premises or the Building required because of (i) misuse or neglect on the part of the Tenant or its employees or visitors, (ii) use of portions of the Demised Premises for preparation, serving, or consumption of food or beverages, reproducing operations, private lavatories or toilets or other special purposes requiring greater or more difficult cleaning work than office areas, (iii) unusual quantity of interior glass surfaces, (iv) non-standard building materials or finishes installed by Tenant or at its request, (v) increases in frequency or scope in any of the items set forth in Exhibit "B" as shall have been requested by Tenant, and (b) removal from the Demised Premises and the Building of (i) so much of any refuse and rubbish of Tenant as shall exceed that normally accumulated daily in the routine of ordinary business office occupancy, and (ii) all of the refuse and rubbish of Tenant's machines and the refuse and rubbish of any other eating facilities requiring special handling (known as "wet garbage"). Landlord and its cleaning contractor and their employees shall have after hours access to the Demised Premises and the use of Tenant's light, power and water in the Demised Premises as may be reasonably required for the purpose of cleaning the Demised Premises.

18.03 Landlord, at its expense, shall furnish adequate hot and cold water for drinking, lavatory, toilet, and ordinary cleaning purposes to the plumbing fixtures of central facilities of the Building serving the Demised Premises.

18.04 Landlord shall keep and maintain the public areas and the public facilities of the Building clean and in good order and the sidewalks adjoining the Building shall be kept in good repair and free of accumulation of snow and ice or unlawful obstructions.

18.05 Landlord, subject to its prior written approval, which shall not be unreasonably withheld or delayed, at its expense, and on Tenant's request, shall maintain listings on the Building directory of the names of Tenant, its organizational divisions and any other person or business entities lawfully occupying the Demised Premises or any part thereof, and the names of any of their officers and employees, provided that the names so listed shall not take up more than Tenant's Proportionate Share of the space on the Building directory, the size of which shall be determined by Landlord. The listing of any name other than that of Tenant on the Building directory or on any of the doors of the Demised Premises shall not be deemed to vest in the person or entity so listed any right or interest in this Lease or in the Demised Premises or to constitute the consent of Landlord required under Article 10, or a waiver thereof. In addition, Landlord will include the name of each of the two (2) entities comprising Tenant in the elevators in the same manner as other tenants of the Building. Notwithstanding anything to the contrary herein, initial listings on the Building directory and in the elevators shall be at Landlord's expense, and any subsequent changes and/or additions shall be at Tenant's expense. Tenant shall also be entitled to install, at Tenant's sole cost and expense and in compliance with all applicable laws and/or requirements of public authorities, signage on the front entrance to the Demised Premises and in the elevator lobby on the 22nd floor of the Building, subject to Landlord's prior written approval, which shall not be unreasonably withheld provided such signage is comparable to that found elsewhere in the Complex or in other first-class office buildings in the Central Business District of Stamford, Connecticut.

18.06 Landlord reserves the right, without any liability to Tenant, except as otherwise expressly provided in this Lease, and without being in breach of any covenant of this Lease, to stop, interrupt or suspend service of any of the heating, ventilating, air conditioning, electric, sanitary, elevator or other Building systems serving the Demised Premises, or the rendition of any of the other services required of Landlord under this Lease, whenever and for so long as may be necessary, by reason of accidents, emergencies, strikes or the making of repairs or changes which Landlord is required by law to make or in good faith deems advisable, or by reason of difficulty in securing proper supplies of fuel, steam, water, electricity, labor or supplies, or by reason of any other cause beyond Landlord's reasonable control, including Governmental restrictions on the use of materials or the use of any of the Building systems. In each instance Landlord shall exercise reasonable diligence to eliminate the cause of stoppage and to effect restoration of service and

shall give Tenant reasonable notice, when practicable, of the commencement and anticipated duration of such stoppage, and if any work is required to be performed in or about the Demised Premises for such purpose, the provisions of Section 15.03 shall apply. Tenant shall not be entitled to any diminution or abatement of rent or other compensation nor shall this Lease or any of the obligations of Tenant be affected or reduced by reason of the interruption, stoppage or suspense of any of the Building systems or services arising out of the causes set forth in this Section.

18.07 Throughout the Term of this Lease, Landlord will provide security in the Complex in the same manner as it does on the date hereof.

ARTICLE 19 ACCESS, CHANGES IN BUILDING FACILITIES, NAME

19.01 All walls, windows and doors bounding the Demised Premises (including exterior Building walls, corridor walls and doors and any corridor entrance), except the inside surfaces thereof, any terraces or roofs adjacent to the Demised Premises, and any space in or adjacent to the Demised Premises used for shafts, stacks, pipes, conduits, fan rooms, ducts, electric or other utilities, sinks or other Building facilities, and the use thereof, as well as access thereto through the Demised Premises for the purposes of operation, maintenance, decoration and repair are reserved to Landlord.

19.02 Tenant shall permit Landlord to install, use and maintain pipes, ducts and conduits within or through the Demised Premises, or through the walls, columns and ceilings therein, provided that the installation work is performed at such times and by such methods as will not unreasonably interfere with Tenant's use and occupancy of the Demised Premises, or damage the appearance thereof, reduce the floor area thereof by more than two percent (2%) (without an appropriate adjustment in rent) or materially affect Tenant's layout. Where access doors are required for mechanical fixtures in or adjacent to the Demised Premises, Landlord shall furnish and install such access doors and confine their location, wherever practical to closets, coat rooms, toilet rooms, corridors and kitchen or pantry rooms. Landlord and Tenant shall cooperate with each other in the location of Landlord's and Tenant's facilities requiring such access doors.

19.03 Landlord or Landlord's agents or employees shall have the right upon request made on reasonable advance notice to Tenant, or to an authorized employee of Tenant at the Demised Premises, to enter and/or pass through the Demised Premises or any part thereof, at reasonable times during reasonable hours, (i) to examine the Demised Premises or to show them to the fee owners, lessors of superior leases, holders of mortgages, insurance carriers, or prospective purchasers, mortgagees or lessees of the land or the Building, and (ii) for the purpose of making such repairs or changes or doing such repainting in or to the Demised Premises or in or to the Building or its facilities as may be provided for by this Lease or as Landlord may deem necessary or as Landlord may be required to make by law or in order to repair and maintain the Building or its fixtures or facilities. Landlord shall be allowed to take all materials into and store such materials upon the Demised Premises which may be required for such repairs, changes, repainting or maintenance. Landlord's rights under this Section shall be exercised in such a manner as will not unreasonably interfere with Tenant's use and occupancy of the Demised Premises. Landlord, its agents or employees, shall also have the right to enter on and/or pass through the Demised Premises, or any part thereof without notice at such times as such entry shall be required by circumstances of emergency affecting the Demised Premises or the Building.

19.04 During the period of 12 months prior to the Expiration Date Landlord may exhibit the Demised Premises to prospective tenants upon the same notice and subject to the same conditions as are provided in Section 19.03. If, during the last month of the term hereof, Tenant shall have removed all of Tenant's property therefrom, Landlord may, upon at least 48 hours notice to Tenant, enter and alter, renovate and redecorate the Demised Premises without incurring any liability to Tenant therefor.

19.05 Landlord reserves the right, at any time after completion of the Building, without incurring any liability to Tenant therefor, to make such changes in or to the Building and the fixtures and equipment thereof, as well as in or to the street entrances, halls, passages, elevators, and stairways thereof, as it may deem necessary or desirable; provided that there be no unreasonable lengthy interference with the use of the Demised Premises or in the services furnished to the Demised Premises, and no reduction in the area of the Demised Premises in

excess of two percent without an appropriate adjustment in rent.

19.06 The Landlord reserves the right to select a name for the Building and to make such change or changes of name as it may deem appropriate during Tenant's occupancy, and Tenant agrees not to refer to the Building by any other name than (i) the name selected by Landlord, or (ii) the postal address approved by the U.S. Post Office.

25

19.07 Landlord may limit and restrict, as provided in the Rules and Regulations attached hereto, the means of access to the Demised Premises outside of Business Hours, so long as Tenant's employees and authorized agents have reasonable access to all parts of the Demised Premises. Tenant and its agents, employees and visitors shall be entitled to access from the Demised Premises to, and the right to use, the toilets, lavatories and powder rooms only on the floor (or floors) on which the Demised Premises are located.

ARTICLE 20 SHORING, NOTICE OF ACCIDENTS, ETC.

20.01 If an excavation or other substructure work shall be undertaken or authorized upon land adjacent to the Building or in the vaults beneath the Building or in subsurface space adjacent to the said vaults, Tenant, without liability on the part of the Landlord therefor, shall afford to the person causing or authorized to cause such excavation or other substructure work license to enter upon the Demised Premises for the purpose of doing such work as such person shall deem necessary to protect or preserve any of the walls or structures of the Building or surrounding lands from injury or damage and to support the same by proper foundations, pinning and/or underpinning, and, except in case of emergency, if so requested by Tenant such entry shall be accomplished in the presence of a representative of Tenant, who shall be designated by Tenant promptly upon Landlord's request. The said license to enter shall be afforded by Tenant without any claim for damages or indemnity against the Landlord and Tenant shall not be entitled to any diminution or abatement of rent on account hereof.

20.02 Tenant shall give notice to Landlord promptly after Tenant learns of (i) any accident in or about the Demised Premises or the Building, (ii) any fire in the Demised Premises, (iii) all damages to or defects in the Demised Premises, including the fixtures, equipment and appurtenances thereof, for the repair of which Landlord might be responsible or which constitutes Landlord's property, and (iv) all damage to or defects in any parts of the Building's sanitary, electrical, heating, ventilating, air conditioning, elevator and other systems located in or passing through the Demised Premises.

ARTICLE 21 NON-LIABILITY AND INDEMNIFICATION

21.01 Neither Landlord nor any agent or employee of Landlord shall be liable to Tenant, its employees, agents, contractors and licensees, and Tenant shall hold Landlord harmless for any injury or damage to Tenant or to any other person or for any damage to, or loss (by theft or otherwise) of, any property of Tenant and/or of any other person, irrespective of the cause of such injury, damage or loss, unless (with respect to personal injury only) such injury was caused by or due to the negligence of Landlord, its agents or employees; it being understood that no property, other than such as might normally be brought upon or kept in the Demised Premises as incident to the reasonable use of the Demised Premises for the purposes herein permitted, will be brought upon or be kept in the Demised Premises. Landlord shall not be liable in any event for loss of, or damage to, any property entrusted to any of Landlord's employees or agents by Tenant without Landlord's specific written consent.

21.02 Tenant shall defend, indemnify and save harmless Landlord and its agents and employees against and from all liabilities, obligations, damages, penalties, claims, costs, charges and expenses, including reasonable architects' and attorneys' fees, which may be imposed upon or incurred by or asserted against Landlord and/or its agents by reason of any of the following occurring during the term of this Lease, or during any period of time prior to the Rent Commencement Date that Tenant may have been given access to or possession of all or any part of the Demised Premises pursuant to Section 3.02:

(A) any work or thing done in or about the Demised Premises or any part

thereof by or at the instance of Tenant, its agents, contractors, subcontractors, servants, employees, licensees or invitees;

(B) any negligence or otherwise wrongful act or omission on the part of Tenant or any of its agents, contractors, subcontractors, servants, employees, subtenants, licensees or invitees;

(C) any accident, injury or damage to any person or property occurring in, on or about the Demised Premises or any part thereof; or

(D) any failure on the part of Tenant to perform or comply with any of the covenants, agreements, terms, provisions, conditions or limitations contained in this Lease on its part to be performed or complied with.

26

In case any action or proceeding is brought against Landlord by reason of any such claim, Tenant upon written notice from Landlord, shall at Tenant's expense resist or defend such action or proceeding by counsel approved by Landlord in writing, which approval Landlord shall not be unreasonably withheld.

21.03 Whenever either party shall be obligated under the terms of this Lease to indemnify the other party, the indemnifying party may select legal counsel (subject to the consent of the indemnified party, which consent shall not be unreasonably withheld) and shall keep the indemnified party fully apprised at all times of the status of such defense. Legal counsel of the insurer for either party is hereby deemed satisfactory to both parties.

21.04 Except as otherwise expressly provided herein, this Lease and the obligations of Tenant to pay rent hereunder and perform all of the other covenants, agreements, terms, provisions and conditions hereunder on the part of Tenant to be performed shall in no way be affected, impaired or excused because Landlord is unable to fulfill any of its obligations under this Lease or is unable to supply or is delayed in supplying any service, express or implied, to be supplied or is unable to make or is delayed in supplying any equipment or fixtures if Landlord is prevented or delayed from so doing by reason of any cause whatsoever beyond Landlord's reasonable control, including, but not limited to, Acts of God, strikes, labor troubles, governmental preemption in connection with a national emergency or by reason of any rule, order or regulation of any department or subdivision thereof of any governmental agency or by reason of the conditions of supply and demand which have been or are affected by war, hostilities or other similar emergency; provided that Landlord shall in each instance exercise reasonable diligence to effect performance when and as soon as possible.

21.05 Notwithstanding any contrary provisions of this Lease whatsoever, including, without limitation, those pertaining to use and Permitted Use, Tenant shall not use, or permit the use of the Demised Premises, the Building or the Complex so as to create or result in, directly or indirectly, (a) any sudden or gradual spill, leak, discharge, escape, seepage, infiltration, abandonment, dumping, disposal or storage of any hazardous or industrial waste, substance or contamination, effluent, sewage, pollution or other detrimental or deleterious material or substance (including without limitation asbestos), or the disposal, storage or abandonment on the Complex of any material, tank or container holding or contaminated by any of the foregoing residues thereof, or the installation of any material or product containing or composed of any of the foregoing, in, on, from under or above the Complex (the foregoing occurrences being hereinafter collectively called "Environmental Hazard"), or (b) any violation, or state of facts or condition which would result in a violation, of any Federal, State or local statute, law, code, rule, regulation or order applicable to any Environmental Hazard (the foregoing being hereinafter collectively call "Legal Violation"). In the event of the violation of the foregoing by Tenant, in addition to all other rights and remedies of Landlord under this Lease, regardless of when the existence of the Environmental Hazard or Legal Violation is determined, and whether during the Term or after the Expiration Date, (I) Tenant shall, immediately upon notice from Landlord, at Tenant's sole cost and expense, at Landlord's option, either (x) take all action necessary to test, identify and monitor the Environmental Hazard and to remove the Environmental Hazard from the Complex and dispose of the same and restore the Complex to the condition existing prior to such removal, and/or to remedy any Legal Violation, all in accordance with applicable Federal, State and local statutes, laws, codes, rules, regulations or orders or (y) reimburse Landlord for all costs and expenses incurred by Landlord for engineering or environmental consultant or

laboratory services, in testing, investigating, identifying and monitoring the Environmental Hazard and in removing and disposing of the Environmental Hazard and in restoring the Complex, and/or in remedying any Legal Violation, and (ii) Tenant shall and hereby does defend with legal counsel acceptable to Landlord, indemnify and save harmless Landlord and Landlord's employees, agents, officers and directors ("Others in Interest") against and from all liabilities, obligations, damages, penalties, claims, costs, charges and expenses, including architects' and attorneys' fees and disbursements which may be imposed upon or incurred by or asserted against Landlord and Others in Interest, whether by any governmental authority, Tenant or other third party, by reason of any violation or alleged violation of any of the foregoing provisions of this Section.

21.06 Landlord shall defend, indemnify and save harmless Tenant and its agents and employees against and from all liabilities, obligations, damages, penalties, claims, costs, charges and expenses, including reasonable architects' and attorneys' fees, which may be imposed upon or incurred by or asserted against Tenant and/or its agents by reason of any of the following occurring during the term of this Lease, or during any period of time prior to the Rent Commencement Date that Tenant may have been given access to or possession of all or any part of the Demised Premises pursuant to Section 3.02:

(a) any work or thing done in or about the Demised Premises or any part thereof by or at the instance of Landlord, its agents, contractors, subcontractors, servants or employees;

(b) any negligence or otherwise wrongful act or omission on the part of Landlord or any of its agents, contractors, subcontractors, servants, or employees; or

27

(c) any failure on the part of Landlord to perform or comply with any of the covenants, agreements, terms, provisions, conditions or limitations contained in this Lease on its part to be performed or complied with.

Notwithstanding the foregoing, Landlord's indemnification obligations in this Section 21.06 shall be limited to the extent of the coverage provided under Landlord's insurance for such indemnity provided such indemnification obligation is a covered item under such insurance. In no event shall Landlord be liable to pay Tenant for any punitive or consequential damages the Tenant must pay, including, without limitation, damages for loss of business.

21.07 The agreements to indemnify, defend and hold harmless contained in this Article 21 are not intended to and shall not relieve any insurance carrier of its obligations under policies required to be carried by Landlord or Tenant, respectively, pursuant to this Lease to the extent that such policies cover the subject incident. The parties' respective obligations under this Article 21 shall survive the expiration or earlier termination of the term of this Lease.

28

ARTICLE 22 DESTRUCTION OR DAMAGE

22.01 If the Demised Premises and/or access thereto shall be partially or totally damaged or destroyed by fire or other casualty, then, Landlord shall, subject to its right under Section 22.03 hereof, repair such damage and restore and rebuild the Demised Premises and/or access thereto as nearly as may be reasonably practical to its condition and character immediately prior to such damage or destruction, with reasonable diligence after notice to it of the damage or destruction.

22.02 If the Demised Premises and/or access thereto shall be partially or totally damaged or destroyed by fire or other casualty not attributable to the fault, negligence or misuse of the Demised Premises by the Tenant, its agents or employees under the provisions of this Lease, the rents payable hereunder shall be abated to the extent that the Demised Premises shall be rendered untenantable from the date of such damage or destruction to the date the damage shall be substantially repaired, restored or rebuilt. Should Tenant reoccupy a portion of

the Demised Premises during the period that the repair, restoration, or rebuilding is in progress and prior to the date that the same are made completely tenantable, rents allocable to such portion shall be payable by Tenant from the date of such occupancy to the date the Demised Premises are made tenantable.

22.03 In case the Building shall be so damaged by such fire or other casualty that substantial renovation, reconstruction or demolition of the Building shall, in Landlord's sole opinion, be required (whether or not the Demised Premises shall have been damaged by such fire or other casualty), then Landlord may, at its option, terminate this Lease and the term and estate hereby granted, by notifying Tenant in writing of such termination, within 60 days after the date of such damage. If at any time prior to Landlord giving Tenant the aforesaid notice of termination or commencing the repair and restoration pursuant to Section 22.01, the holder of a superior mortgage or any person claiming under or through the holder of such superior mortgage takes possession of the Building through foreclosure or otherwise, such holder or person shall have a further period of 60 days from the date of so taking possession to terminate this Lease by appropriate written notice to Tenant. In the event that such a notice of termination shall be given pursuant to either of the two immediately preceding sentences, this Lease and the term and estate hereby granted shall expire as of the date of such termination with the same effect as if that were the date hereinbefore set for the expiration of the term of this Lease, and the fixed and additional rent due and to become due hereunder shall be apportioned as of such date if not earlier abated pursuant to Section 22.02. Nothing contained in this Section shall relieve Tenant from any liability to Landlord or to its insurers in connection with any damage to the Demised Premises or the Building by fire or other casualty if Tenant shall be legally liable in such respect.

22.04 No damages, compensation or claim shall be payable by Landlord for inconvenience, loss of business or annoyance arising from any repair or restoration of any portion of the Demised Premises or of the Building pursuant to this Article. Landlord shall use commercially reasonable efforts to effect such repair or restoration promptly and in such manner as not unreasonably to interfere with Tenant's use and occupancy.

22.05 Landlord will not carry insurance of any kind on Tenant's property, and, except as provided by law or its breach of any of its obligations hereunder, shall not be obligated to repair any damage thereto or replace the same.

22.06 The provisions of this Article shall be considered an express agreement governing any case of damage or destruction of the Demised Premises by fire or other casualty, and any law to the contrary, now or hereafter in force, shall have no application in such case.

22.07 Notwithstanding any of the foregoing provisions of this Article, if Landlord or the holder of any superior mortgage shall be unable to collect all of the insurance proceeds (including rent insurance proceeds) applicable to damage or destruction of the Demised Premises or the Building by fire or other cause, by reason of some action or inaction on the part of the Tenant or any of its employees, agents or contractors, then, without prejudice to any other remedies which may be available against Tenant, the abatement of Tenant's rents provided for in this Article shall not be effective to the extent of the uncollected insurance proceeds.

22.08 If the Demised Premises shall be totally damaged or rendered wholly untenantable or inaccessible by fire or other casualty, and Landlord has not exercised its option to terminate this lease (as set forth in Section 22.03 above), then Landlord shall, as soon as reasonably practicable following the occurrence of the subject fire or other casualty, deliver to Tenant written notice of Landlord's estimated time for restoration of the Premises. If the estimated date of completion of such restoration work is more than twelve (12) months following the date of occurrence of the subject fire or other casualty, then Tenant shall have the right to terminate this Lease by written notice delivered to Landlord within fifteen (15) days following receipt of such written notice by Landlord. In the case of proper exercise of any termination right afforded to Tenant under this Section 22.08, this Lease shall terminate as of the date of

delivery of such written notice by Tenant, and neither party shall have any further obligation or liability to the other hereunder (except for obligations or liabilities previously accrued which remain unsatisfied).

ARTICLE 23 EMINENT DOMAIN

23.01 In the event that the land, Building or any part thereof or the Demised Premises or any part thereof shall be taken in condemnation proceedings or by the exercise of any right of eminent domain or by agreement between the Landlord on the one hand and any governmental authority authorized to exercise such right on the other hand, Landlord shall be entitled to collect from any condemnor the entire award or awards that may be made in any such proceeding without deduction therefrom for any estate hereby vested in or owned by Tenant, to be paid out as in this Article provided. Tenant hereby expressly assigns to Landlord all of its right, title and interest in or to every such award and also agrees to execute any and all further documents that may be required in order to facilitate the collection thereof by Landlord.

23.02 At any time during the term of this Lease if title to the whole or substantially all of the land, Building and/or Demised Premises shall be taken in condemnation proceedings or by the exercise of any right of eminent domain or by agreement between the Landlord on the one hand and any governmental authority authorized to exercise such right on the other hand, this Lease shall terminate and expire on the date of such taking and the fixed rent and additional rent provided to be paid by Tenant shall be apportioned and paid to the date of such taking. For the purposes of this Article "substantially all of the land, Building and/or Demised Premises" shall be deemed to have been taken if the remaining portion of such land, Building or Demised Premises not so taken cannot reasonably or practicably be repaired or reconverted so as to permit the use thereof for substantially the same purposes for which such land, Building or Demised Premises were used immediately prior to such taking.

23.03 However, if substantially all of the land or Building is not so taken and if only a part of the entire Demised Premises shall be so taken, this Lease nevertheless shall continue in full force and effect, except that Tenant may elect to terminate this Lease if that portion of the Demised Premises then occupied by Tenant shall be reduced by more than 25%. Tenant shall give notice of such election to Landlord not later than thirty (30) days after (i) notice of such taking is given by Landlord to Tenant, or (ii) the date of such taking, whichever occurs first. Upon the giving of such notice by Tenant this Lease shall terminate on the date of service of Tenant's notice and the fixed rent and additional rent due and to become due, shall be prorated and adjusted as of the date of the taking. If Tenant fails to give such notice upon such partial taking, and this Lease continues in force as to any part of the Demised Premises not taken, the rents apportioned to the part taken shall be prorated and adjusted as of the date of taking and from such date the fixed rent and additional rent shall be reduced to the amount apportioned to the remainder of the Demised Premises.

23.04 In the event of any such taking of less than the whole of the Building which does not result in a termination of this Lease, or in the event of such a taking of all or any part of the Demised Premises which does not result in a termination of this Lease, Landlord, at its expense, shall proceed with reasonable diligence to repair, alter and restore the remaining parts of the Building and the Demised Premises to substantially the same condition as it was in immediately prior to such taking to the extent that the same may be feasible, so as to constitute a tenantable Building and Demised Premises, provided that Landlord's liability under this Section shall be limited to the proportionate amount received by Landlord as an award arising out of such taking.

ARTICLE 24 SURRENDER

24.01 On the last day of the term of this Lease, or upon any earlier termination of this Lease, or upon any reentry by Landlord upon the Demised Premises, Tenant shall quit and surrender the Demised Premises to Landlord broom clean, in good order, condition and repair except for ordinary wear and tear and damage by fire or other insured casualty; Tenant shall remove Tenant's property subject to the provisions of Article 14 hereof; and Tenant shall surrender to Landlord all keys to offices, lavatories and mail boxes and all Building identification and parking cards possessed by Tenant's employees.

24.02 In the event Tenant remains in possession of the Demised Premises after the termination of this Lease without the execution by Landlord and Tenant of a new lease, Tenant shall, at Landlord's option, be deemed to be occupying the Demised Premises as a tenant from month to month, at a monthly rental equal to one and one-half (1.5) times the fixed rent and additional rent payable during the last month of the term, subject to all of the other terms of this Lease insofar as the same are applicable to a month to month tenancy.

30

24.03 In the event Tenant remains in possession of the Demised Premises following the termination of this Lease and Landlord does not elect to treat Tenant as a month-to-month tenant as provided in the Section 24.02 above, Tenant hereby indemnifies and agrees to hold Landlord harmless from and against any loss, cost, liability, claim, damage, fine, penalty, and expense, including attorneys' fees and disbursements, resulting from such delay by Tenant in surrendering the Demised Premises upon the termination of this Lease as provided in this Article 24, including without limitation, any claims made by any succeeding tenant or prospective tenant based upon such delay. In the event Tenant remains in possession of the Demised Premises for any period of time following the Expiration Date, Tenant shall be in default, and in addition to the rent provided in Section 24.02 hereof, Landlord shall be entitled to all of its rights and remedies provided in this Lease.

ARTICLE 25 CONDITIONS OF LIMITATION

25.01 (A) This Lease and the term and estate hereby granted are subject, inter alia, to the limitation that whenever Tenant shall make an assignment for the benefit of creditors, or shall file a voluntary petition under any bankruptcy or insolvency law, or an involuntary petition alleging an act of bankruptcy or insolvency is filed against Tenant, or whenever a petition shall be filed by or against Tenant seeking any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any present or any future federal bankruptcy act or any other present or future applicable federal, state or other statute or law, or shall seek or consent to or acquiesce in the appointment of any trustee, receiver or liquidator of Tenant or of all or any substantial part of its properties, or whenever a permanent or temporary receiver of Tenant or of or for the property of Tenant shall be appointed, or if Tenant shall plead bankruptcy or insolvency as a defense in any action or proceeding, then, Landlord (a) at any time after receipt of notice of the occurrence of any such event, or (b) if such event occurs without the acquiescence of Tenant, at any time after the event continues for 60 days, may give Tenant a notice of intention to end the term of this Lease at the expiration of five (5) days from the service of such notice of intention, and upon the expiration of said five (5) day period this Lease and the term and estate hereby granted, whether or not the term shall theretofore have commenced, shall terminate with the same effect as if that day were the Expiration Date, but Tenant shall remain liable for damages as provided in Article 27.

(B) If this Lease is assigned to any person or entity pursuant to the provisions of the Bankruptcy Code, 11 U.S.C. Section 101 et seq. (the "Bankruptcy Code"), any and all consideration payable or otherwise to be delivered in connection with such assignment shall be paid or delivered to Landlord, shall be and remain the exclusive property of Landlord and shall not constitute property of Tenant or of the estate of Tenant within the meaning of the Bankruptcy Code. Any and all monies and other consideration constituting Landlord's property under the preceding sentence not paid or delivered to Landlord shall be held in trust for the benefit of Landlord and be promptly paid to or turned over to Landlord.

(C) If, pursuant to the provisions of the Bankruptcy Code, Tenant assumes this Lease and proposes to assign the same to any person or entity who shall have made a bona fide offer to accept an assignment of this Lease on terms acceptable to Tenant then notice of such proposed assignment, setting forth (i) the name and address of such person, (ii) all of the terms and conditions of such offer, and (iii) the adequate assurance to be provided Landlord to assure such person's future performance under this Lease, including, without limitation, the assurance referred to in Section 365(b)(3) of the Bankruptcy Code, shall be given to Landlord by Tenant no later than twenty (20) days after receipt by Tenant but in any event no later than ten (10) days prior to the date that approval to enter into such assignment and assumption, and Landlord shall thereupon have the prior right and option, subject to all legal restrictions under the Bankruptcy Code, and to be exercised by notice to Tenant given at any time prior to the effective date of such proposed assignment, to accept an assignment of this Lease upon the same terms and conditions and for the same consideration, if any, as the bona fide offer made by such person, less any brokerage commissions which may be payable out of the consideration to be paid by such person for the assignment of this Lease.

25.02 This Lease and the term and estate hereby granted are subject to the further limitation that:

(A) whenever Tenant shall default in the payment of any installment of fixed rent, or in the payment of any additional rent, on any day upon which the same shall be due and payable and such default shall continue for ten (10) days after the date on which the same was due and payable (five (5) days after written notice from Landlord that such payment is overdue with respect to the first two (2) late payments in any twelve (12) month period); or

(B) whenever Tenant shall do or permit anything to be done, whether by action or inaction, contrary to any of Tenant's obligations hereunder, and if such situation shall continue and shall not be remedied by Tenant within thirty (30) days after Landlord shall have given to the Tenant a notice specifying the same, or, in the case of a happening or default which cannot with due diligence

31

be cured within a period of thirty (30) days and the continuance of which for the period required for cure will not subject Landlord to the risk of criminal liability or termination of any superior lease or foreclosure of any superior mortgage, if Tenant shall not duly institute within such thirty (30) day period and promptly and diligently prosecute to completion all steps necessary to remedy the same; or

(C) whenever any event shall occur or any contingency shall arise whereby this Lease or any interest therein or the estate hereby granted or any portion thereof or the unexpired balance of the term hereof would by operation of law or otherwise, devolve upon or pass to any person, firm or corporation other than Tenant, except as expressly permitted in Article 10.

Then in any such event at any time thereafter, Landlord may give to Tenant a three (3) day notice ending the term of this Lease upon the expiration of such three (3) day period and thereupon this Lease and the term and estate hereby granted, whether or not the term has theretofore commenced, shall terminate with the same effect as if that day were the Expiration Date, but Tenant shall remain liable for damages as provided in Article 27.

25.03 Notwithstanding anything to the contrary set forth in this Lease, if:

(A) Tenant or its employees, agents, contractors or invitees (such employees, agents, contractors and invitees being hereinafter referred to as "Related Persons") shall violate the provisions of this Lease relating to parking; or

(B) Tenant or any of its Related Persons shall violate the provisions of this Lease dealing with excessive noise, loitering or other conduct that shall constitute a nuisance to the other tenants of the Building; or

(C) Tenant shall allow a population density in the Demised Premises of more than one (1) person per 150square feet;

and Landlord shall give Tenant notice of its breach of any of the foregoing provisions or any other provision of this Lease two (2) times during the Term of this Lease, then, notwithstanding that any such breach shall have been timely cured, any future breach of the subject provision shall be deemed to be a "Deliberate Default". Upon the occurrence of a Deliberate Default, Landlord,

without giving Tenant any notice of such default and without giving Tenant any opportunity to cure such default, may (in addition to resorting to the other remedies afforded to Landlord under this Lease or at law or equity) give to Tenant a notice of Landlord's intention to end the Term of this Lease at the expiration of three (3) days from the date of service of such notice of intention and, upon the expiration of said three (3) days, this Lease and the Term and estate hereby granted (whether or not the Term shall have theretofore commenced) shall terminate with the same effect as if that day were the Expiration Date, but Tenant shall remain liable for damages as provided in Article 27.

ARTICLE 26 RE-ENTRY BY LANDLORD - DEFAULT PROVISIONS

26.01 If this Lease shall terminate for any reason whatsoever, Landlord or Landlord's agents and employees may without further notice immediately or at any time thereafter enter upon and reenter the Demised Premises or any part thereof, and possess or repossess itself thereof either by summary dispossess proceedings, ejectment or by any suitable action or proceeding at law, or by agreement, or otherwise (excluding force) and may dispossess and remove Tenant and all other persons and property from the Demised Premises without being liable to indictment, prosecution or damage therefor, and may repossess the Demised Premises and the right to receive all rental income again as and of its first estate and interest therein. The words "enter" or "reenter", "possess or "repossess" as herein used, are not restricted to their technical legal meaning. In the event of any termination of this Lease, or of reentry by summary dispossess proceedings, ejectment or by any suitable action or proceeding at law, or by agreement, or by force or otherwise by reason of default hereunder on the part of Tenant, Tenant shall thereupon pay to Landlord the fixed rent and additional rent due up to the time of such termination of this Lease, or such recovery of possession of the Demised Premises by Landlord, as the case may be, and shall also pay to Landlord damages as provided in Article 27.

26.02 In the event of any breach or threatened breach by Tenant of any of the agreements, terms, covenants or conditions contained in this Lease, Landlord shall be entitled to enjoin such breach or threatened breach and shall have the right to invoke any right and remedy allowed at law or in equity or by statute or otherwise as though reentry, summary proceedings, and other remedies were not provided for in this Lease.

26.03 Each right and remedy of Landlord provided for in this Lease shall be cumulative and shall be in addition to every other right

32

or remedy provided for in this Lease or now or hereafter existing at law or in equity or by statute or otherwise, and the exercise or beginning of the exercise by Landlord of any one or more of the rights or remedies provided for in this Lease or now or hereafter existing at law or in equity or by statute or otherwise shall not preclude the simultaneous or later exercise by Landlord of any or all other rights or remedies provided for in this Lease or now or hereafter existing at law or in equity or by statute or otherwise.

26.04 If this Lease shall terminate under the provisions of Article 25, or if Landlord shall reenter the Demised Premises under the provisions of this Article, or in the event of the termination of this Lease or of reentry, by or under any summary dispossess or other proceeding or action or any provision of law by reason of default hereunder on the part of Tenant, Landlord shall be entitled to retain all monies, if any, paid by Tenant to Landlord, whether as advance rent, security or otherwise, but such monies shall be credited by Landlord against any fixed rent or additional rent due from Tenant at the time of such termination or reentry or, at Landlord's option, against any damages payable by Tenant under Article 27 or pursuant to law.

> ARTICLE 27 DAMAGES

27.01 If this Lease is terminated under the provisions of Article 25, or if Landlord shall reenter the Demised Premises under the provisions of Article 26 or in the event of the termination of this Lease, or of reentry by summary dispossess proceedings, ejectment or by any suitable action or proceeding at law, or by agreement, or otherwise, by reason of default hereunder on the part of Tenant, Tenant shall pay Landlord as damages, at the election of Landlord, either:

(A) on demand, a sum which at the time of such termination of this Lease or at the time of any such reentry by Landlord, as the case may be, represents the excess of (i) the aggregate of the fixed rent and the additional rent payable hereunder which would have been payable by Tenant (conclusively presuming the additional rent to be the same as was payable for the year immediately preceding such termination) for the period commencing with such earlier termination of this Lease or the date of such reentry, as the case may be, and ending with the Expiration Date, had this Lease not so terminated or had Landlord not so reentered the Demised Premises, over (ii) the aggregate rental value (calculated as of the date of such termination or reentry) of the Demised Premises for the same period; or

(B) sums equal to the fixed rent and the additional rent (as above presumed) payable hereunder which would have been payable by Tenant had this Lease not so terminated, or had Landlord not so reentered the Demised Premises. payable quarterly, in advance, but otherwise upon the terms therefor specified herein following such termination or such reentry and until the Expiration Date, provided, however, that if Landlord shall relet the Demised Premises or any portion or portions thereof during said period, Landlord shall credit Tenant with the net rents received by Landlord from such reletting, such net rents to be determined by first deducting from the gross rents as and when received by Landlord from such reletting the expenses incurred or paid by Landlord in terminating the Lease or in reentering the Demised Premises and in securing possession thereof, as well as the expenses of reletting, including altering and preparing the Demised Premises or any portion or portions thereof for new tenants, brokers' commissions, advertising expenses, and all other expenses properly chargeable against the Demised Premises and the rental therefrom; it being understood that any such reletting may be for a period shorter or longer than the remaining term of this Lease, but in no event shall Tenant be entitled to receive any excess of such net rents over the sums payable by Tenant to Landlord hereunder, nor shall Tenant be entitled in any suit for the collection of damages pursuant to this subsection to a credit in respect of any net rents from a reletting, except to the extent that such net rents are actually received by Landlord. If the Demised Premises or any part thereof should be relet in combination with other space, then proper apportionment shall be made of the rent received from such reletting and of the expenses of reletting, and Landlord shall have the right to grant reasonable rent concessions to attract one or more new tenants and to permit the term of any new lease covering part or all of the Demised Premises to be for a shorter or longer period than provided for herein.

If the Demised Premises or any part thereof be relet by Landlord for the unexpired portion of the term of this Lease, or any part thereof, before presentation of proof of such damages to any court, commission or tribunal, the amount of rent reserved upon such releting shall, prima facie, be the fair and reasonable rental value for the Demised Premises, or part thereof, so relet during the term of the reletting. In the event of the termination of this Lease following a default by Tenant, Landlord shall have a duty to employ commercially reasonable efforts to relet the Demised Premises.

27.02 In the event Landlord elects to collect damages from Tenant under Section 27.01(b) at any time subsequent to such election and upon ten days prior written notice to Tenant, Landlord may elect to collect a lump sum under Section 27.01(a), crediting Tenant with amounts theretofore received by Landlord as damages.

33

27.03 Suit or suits for the recovery of such damages, or any installments thereof, may be brought by Landlord from time to time at its election, and nothing contained herein shall be deemed to require Landlord to postpone suit until the date when the term of this Lease would have expired if it had not been so terminated under the provisions of Article 25, or under any provision of law,

or had Landlord not reentered the Demised Premises. Nothing herein contained shall be construed to limit or preclude recovery by Landlord against Tenant of any sums or damages to which, in addition to the damages particularly provided above, Landlord may lawfully be entitled by reason of any default hereunder or otherwise on the part of Tenant. Nothing herein contained shall be construed to limit or prejudice the right of the Landlord to provide for and obtain as liquidated damages by reason of the termination of this Lease or reentry on the Demised Premises for the default of Tenant under this Lease, an amount equal to the maximum allowed by any statute or rule of law in effect at the time, and governing the proceedings in which such damages are to be proved whether or not such amount be greater, equal to, or less than any of the sums referred to in Section 27.01.

27.04 The foregoing Sections of this Article shall apply even if the default by Tenant has occurred prior to the Rent Commencement Date and/or prior to Tenant taking possession of the Demised Premises. The parties acknowledge that this instrument is a lease and not a contract to make a lease.

ARTICLE 28 WAIVERS

28.01 Tenant, for itself, and on behalf of any and all persons claiming through or under Tenant, including creditors of all kinds, does hereby waive and surrender all right and privilege so far as is permitted by law, which they or any of them might have under or by reason of any present or future law, of the service of any notice of intention to reenter and also waives any and all right to redemption or reentry or repossession in case Tenant shall be dispossessed or ejected by process of law or in case of reentry or repossession by Landlord upon any expiration or termination of this Lease as herein provided.

28.02 Tenant waives Tenant's rights, if any, to designate the items against which any payments made by Tenant are to be credited, and Tenant agrees that Landlord may apply any payments made by Tenant to any items it sees fit, irrespective of and notwithstanding any designation or request by Tenant as to the items against which any such payments shall be credited.

28.03 Tenant waives Tenant's rights, if any, to assert a counterclaim in any summary proceeding brought by Landlord against Tenant, and Tenant agrees to assert any such claim against Landlord only by way of a separate action or proceeding.

28.04 TO THE EXTENT PERMITTED BY APPLICABLE LAW, LANDLORD AND TENANT HEREBY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY EITHER AGAINST THE OTHER ON ANY MATTER WHATSOEVER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS LEASE, THE RELATIONSHIP OF LANDLORD AND TENANT, OR TENANT'S USE OR OCCUPANCY OF THE DEMISED PREMISES, OR ANY EMERGENCY OR OTHER STATUTORY REMEDY WITH RESPECT THERETO.

ARTICLE 29 NO OTHER WAIVERS OR MODIFICATIONS

29.01 The failure of either party to insist in any one or more instances upon the strict performance of any one or more of the agreements, terms, covenants, conditions or obligations of this Lease, or to exercise any right, remedy or election herein contained, shall not be construed as a waiver or relinquishment for the future of the performance of such one or more obligations of this Lease or of the right to exercise such election, but the same shall continue and remain in full force and effect with respect to any subsequent breach, act or omission. The manner of enforcement or the failure of Landlord to enforce any of the Rules and Regulations set forth herein, or hereafter adopted against the Tenant and/or any other tenant in the Building shall not be deemed a waiver of any such Rules and Regulations. No executory agreement hereafter made between Landlord and Tenant shall be effective to change, modify, waive, release, discharge, terminate or effect an abandonment of this Lease, in whole or part, unless such executory agreement is in writing, refers expressly to this Lease and is signed by the party against whom enforcement of the change, modification, waiver, release, discharge or termination or effectuation of the abandonment is sought.

29.02 The following specific provisions of this Section shall not be deemed to limit the generality of the foregoing provisions of this Article:

(A) No agreement to accept a surrender of all or any part of the Demised Premises shall be valid unless in writing and signed by Landlord. The delivery of keys to an employee of Landlord or of its agent shall not operate as a termination of this Lease or a surrender of the Demised Premises. If Tenant shall at any time request Landlord to sublet the Demised Premises for Tenant's account, Landlord or its agent is authorized to receive said keys for such purposes without releasing Tenant from any of its obligations under this Lease, and Tenant hereby releases Landlord of any liability for loss or damage to any of Tenant's property in connection with such subletting.

(B) The receipt or acceptance by Landlord of rents with knowledge of breach by Tenant of any term, agreement, covenant, condition or obligation of this Lease shall not be deemed a waiver of such breach.

(C) No payment by Tenant or receipt by Landlord of a lesser amount than the correct fixed rent or additional rent due hereunder shall be deemed to be other than a payment on account, nor shall any endorsement or statement on any check or any letter accompanying any check or payment be deemed to effect or evidence an accord and satisfaction, and Landlord may accept such check on payment without prejudice to Landlord's right to recover the balance or pursue any other remedy in this Lease or at law provided. Payment by Tenant for any amount of fixed rent or additional rent shall not be deemed a waiver of any claims Tenant may have against Landlord.

(D) If, in connection with obtaining, continuing or renewing financing, for which the Building, land or the leasehold or any interest therein represents collateral in whole or in part, a banking, insurance or other lender shall request reasonable modifications of this Lease as a condition of such financing, Tenant will not unreasonably withhold, delay or defer its consent thereto, provided that such modifications do not increase the obligations of Tenant hereunder or adversely affect to a material degree the Tenant's leasehold interest hereby created.

ARTICLE 30 SUMS DUE LANDLORD

30.01 If Tenant shall default in the performance of any covenant, term, provision or condition herein contained, Landlord, without thereby waiving such default, may perform the same for the account and at the expense of Tenant without notice in a case of emergency and in any other case if such default continues after the expiration of the applicable notice and cure period provided herein for the cure thereof (if any). Bills for any reasonable and necessary expense incurred by Landlord in connection with any such performance by Landlord for the account of Tenant, and reasonable and necessary bills for all costs, expenses and disbursements, including (without being limited to) reasonable counsel fees, incurred in collecting or endeavoring to collect the fixed rent or additional rent or other charge or any part thereof or enforcing or endeavoring to enforce any rights against Tenant under or in connection with this Lease, or pursuant to law, including (without being limited to) any such cost, expense and disbursement involved in instituting and prosecuting summary proceedings, as well as bills for any property, material, labor or services provided, furnished or rendered, or caused to be provided, furnished or rendered, by Landlord to Tenant including (without being limited to) electric lamps and other equipment, construction work done for the account of Tenant, water, ice, drinking water, drinking cups, towel and other services, as well as for any charges for any additional elevator, heating, air conditioning or cleaning services and any charges for other services incurred by Tenant under this Lease, may be sent by Landlord to Tenant monthly, or immediately, at Landlord's option, and shall be due and payable by Tenant in accordance with the terms of said bills and if not paid when due, the amounts thereof shall immediately become due and payable as additional rent under the Lease together with interest thereon at the rate of 12% per annum from the date said bills should have been paid in accordance with their terms. In addition, Tenant shall reimburse Landlord for any and all reasonable attorney fees incurred by Landlord in connection with the preparation, review, negotiation and/or consummation of any amendment, modification, agreement or other understanding made at the request of, or as an accommodation to, Tenant with respect to this Lease.

> ARTICLE 31 CONSENTS - BROKER

31.01 Any provision of this Lease which requires Landlord not to unreasonably withhold or delay its consent shall never be the basis for an award of damages or give rise to a right of setoff to Tenant, but shall only be the basis for a declaratory judgment or specific injunction with respect to the matter in question.

31.02 (a) Tenant represents and warrants that the sole broker with whom it has dealt with in this transaction is Newmark & Company Real Estate, Inc. (the "Broker"), and that no other broker interested Tenant in the Demised Premises. Landlord shall be

35

responsible for the payment of any real estate commission to Broker but Tenant shall hold Landlord harmless from the claim of any other real estate broker or salesman claiming to have interested or have been responsible for Tenant's execution of this Lease. With respect to the foregoing, in the event Tenant elects to use any broker (the "New Broker") other than or together with the Broker in connection with the extension of this Lease (whether by way of a renewal option or a separate extension agreement), Landlord shall be responsible for the commission (if any) due and owing to the Broker in connection with such extension and Tenant shall be responsible for (and indemnify Landlord against) the commission claimed by the New Broker and any liabilities and expenses, including reasonable attorney fees, incurred by Landlord with respect thereto.

(b) Landlord represents and warrants that the sole broker with whom it has dealt with in this transaction is the Broker. Landlord shall be responsible for the payment of any real estate commission to Broker and shall hold Tenant harmless from the claim of any other real estate broker or salesman claiming to have dealt with Landlord in connection with this Lease.

ARTICLE 32 NOTICES

32.01 Any notice, statement, demand, request or other communication required or permitted to be given, rendered or made by either party to the other, pursuant to this Lease or pursuant to any applicable law or requirement of public authority, shall be in writing (whether or not so stated elsewhere in this Lease) and shall be deemed to have been properly given, rendered or made, if sent by registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

(a) if to Tenant, at the address hereinabove set forth (except that after the Rent Commencement Date, Tenant's address, unless Tenant shall give written notice to the contrary, shall be the Building), with a courtesy copy to Kane Kessler P.C., 1350 Avenue of the Americas, New York, New York 10019, Attn: Robert L. Lawrence, Esq., and

(b) if to Landlord, at 360 Hamilton Avenue, White Plains, New York 10601, Attn: Managing Director, with a copy to 225 Broadhollow Road, Melville, New York 11747, Attn: Vice President and Legal Counsel - Real Estate,

and shall be deemed to have been given, rendered or made on the date received or rejected by the addressee. Either party may, by notice as aforesaid, designate a different address or addresses for notices, statements, demands or other communications intended for it.

32.02 However, notices requesting after hours service pursuant to Sections 17.01 and 18.01 may be delivered, provided they are in writing, to the Building Superintendent or any other person in the Building designated by Landlord to receive such notices, and notice of fire, accident or other emergency shall be given by facsimile or by personal delivery of written notice to that address designated for this purpose from time to time by the respective parties hereto.

32.03 Whenever either party shall consist of more than one person or entity, any notice, statement, demand, or other communication required or permitted to be given, rendered or made to or by, and any payment to be made to such party, shall be deemed duly given, rendered, made or paid if addressed to or by (or in the case of payment by check, to the order of) any one of such persons or entities who shall be designated from time to time as the authorized

representative of such party. Such party shall promptly notify the other of the identity of such person or entity who is so to act on behalf of all persons and entities then comprising such party and of all changes in such identity.

ARTICLE 33 ESTOPPEL CERTIFICATE

33.01 Tenant agrees, at any time, and from time to time, within ten (10) days after Landlord's request therefor, to execute and deliver, without cost or expense to the Landlord, a statement in writing (i) certifying that this Lease is unmodified and in full force and effect (or if there have been modifications, that the same is in full force and effect as modified and stating the modifications), (ii) certifying the dates to which the fixed rent and additional rent have been paid, (iii) stating whether, to the best knowledge of the signer of the certificate, the Landlord is in default in performance of any of its obligations under this Lease, and, if so, specifying each such default of which the status of this Lease as Landlord shall reasonably request. Tenant hereby acknowledges that the statement delivered pursuant hereto may be relied upon by any other person with whom the Landlord may be dealing.

36

33.02 Landlord agrees, at any time, and from time to time, within ten (10) days after Tenant's request therefor, to execute and deliver, without cost or expense to the Tenant, a statement in writing (i) certifying that this Lease is unmodified and in full force and effect (or if there have been modifications, that the same is in full force and effect as modified and stating the modifications), (ii) certifying the dates to which the fixed rent and additional rent have been paid, (iii) stating whether, to the best knowledge of the signer of the certificate, the Tenant is in default in performance of any of its obligations under this Lease, and, if so, specifying each such default of which the Landlord may have knowledge, and (iv) containing such other information as to the status of this Lease as Tenant shall reasonably request. Landlord hereby acknowledges that the statement delivered pursuant hereto may be relied upon by any other person with whom the Tenant may be dealing.

33.03 Should the Tenant fail to execute and deliver to the Landlord the certificate and statement set forth in Section 33.01, above, then the Landlord may execute the statement as attorney-in-fact for the Tenant specifying to the best of Landlord's knowledge, the items called for in said Section 33.01.

37

ARTICLE 34 NO OTHER REPRESENTATIONS, CONSTRUCTION, GOVERNING LAW

34.01 Tenant expressly acknowledges and agrees that Landlord has not made and is not making, and Tenant in executing and delivering this Lease is not relying upon, any warranties, representations, promises or statements, except to the extent that the same are expressly set forth in the Lease, or in any other written agreement which may be made and executed between the parties concurrently with the execution and delivery of this Lease, which agreement shall expressly refer to this Lease.

34.02 If any of the provisions of this Lease, or the application thereof to any person or circumstances, shall, to any extent, be invalid or unenforceable, the remainder of this Lease, or the application of such provision or provisions to persons or circumstances other than those as to whom or which it is held invalid or unenforceable, shall not be affected thereby, and every provision of this Lease shall be valid and enforceable to the fullest extent permitted by law.

34.03 This Lease shall be governed in all respects by the laws of the State of Connecticut.

34.04 Landlord and Tenant understand, agree and acknowledge that this Lease has been freely negotiated by both parties and that, in the event of any controversy, dispute, or contest over the meaning, interpretation, validity, or enforceablity of this lease or any of its terms and conditions, there shall be no inference, presumption or conclusion drawn whatsoever against either party by virtue of that party having drafted this Lease or any portion hereof.

ARTICLE 35 PARTIES BOUND

35.01 The obligations of this Lease shall bind and benefit the successors and assigns of the parties with the same effect as if mentioned in each instance where a party is named or referred to, except that no violation of the provisions of Article 10 shall operate to vest any rights in any successor or assignee of Tenant and that the provisions of this Article shall not be construed as modifying the conditions of limitation contained in Article 25. However, the obligations of Landlord under this Lease shall not be binding upon Landlord herein named with respect to any period subsequent to the transfer of its interest in the Building as owner or lessee thereof and in the event of such transfer the obligations thereafter shall be binding upon each transferee of the interest of Landlord herein named as such owner or lessee of the Building, but only with respect to the period ending with a subsequent transfer within the meaning of this Article, and such transferee, by accepting such interest, shall be deemed to have assumed such obligations except only as may be expressly otherwise provided in this Lease. A lease of Landlord's entire interest in the Building as owner or lessee thereof shall be deemed a transfer within the meaning of this Article. For any transfer, Landlord shall require the transferee to assume and accept the obligations relating to the Security Deposit hereunder.

35.02 Tenant shall look solely to the estate and interest of Landlord, its successors and assigns, in the land and Building (or the proceeds thereof) for the collection of a judgment (or other judicial process) requiring the payment of money by Landlord in the event of any default by Landlord hereunder, and no other property or assets of Landlord shall be subject to levy, execution or other enforcement procedure for the satisfaction of Tenant's remedies under or with respect to either this Lease, the relationship of Landlord and Tenant hereunder or Tenant's use and occupancy of the Demised Premises.

ARTICLE 36 CERTAIN DEFINITIONS AND CONSTRUCTIONS

36.01 For the purposes of this Lease and all agreements supplemental to this Lease, unless the context otherwise requires:

(A) The term "Mortgage" shall include any indenture of mortgage and deed of trust to a trustee to secure an issue of bonds and debentures, and the term "Mortgagee" shall include such a trustee.

(B) The terms "include", "including" and "such as" shall each be construed as if followed by the phrase "without being limited to."

(C) The term "obligations of this Lease" and words of like import, shall mean covenants to pay rent and additional rent under this Lease and all of the other covenants and conditions contained in this Lease. Any provision in this Lease that one party or the other or both shall do or not do, or shall cause or permit or not cause or permit a particular act, condition or circumstance shall be

38

deemed to mean that such party so covenants or both parties so covenant, as the case may be.

(D) The term "Tenant's obligations hereunder" and words of like import, and the term "Landlord's obligations hereunder" and words of like import, shall mean the obligations to this Lease which are to be performed or observed by Tenant, or by Landlord, as the case may be. Reference to "performance" of either party's obligations under this Lease shall be construed as "performance and observance." Tenant's obligations hereunder shall be construed in every instance as conditions as well as covenants.

(E) Reference to Tenant being or not being "in default hereunder" or words of like import, shall mean that Tenant is in default in the performance of one or more of Tenant's obligations hereunder, or that Tenant is not in default in the performance of any of Tenant's obligations hereunder, or that a condition of the character described in Section 25.01 has occurred and continues or has not occurred or does not continue, as the case may be. (F) References to Landlord as having "no liability to Tenant" or being "without liability to Tenant" shall mean that Tenant is not entitled to terminate this Lease, or to claim actual or constructive eviction, partial or total, or to receive any abatement or diminution of rent, or to be relieved in any manner of any of its other obligations hereunder, or to be compensated for loss or injury suffered or to enforce any other kind of liability whatsoever against the Landlord under or with respect to this Lease or with respect to Tenant's use or occupancy of the Demised Premises.

(G) The term "laws and/or requirements of public authorities" and words of like import shall mean laws and ordinances of any or all of the federal, state, city and county governments and rules, regulations, orders and or directives of any or all departments, subdivisions, boards, agencies or offices thereof, or any other governmental, public or quasi-public authorities, having jurisdiction in the premises, and/or the direction of any public officer pursuant to law.

(H) The term "requirements of insurance bodies" and words of like import shall mean rules, regulations, orders and other requirements of the Board of Fire Underwriters and/or the Fire Insurance Rating Organization in Connecticut and/or any other similar body performing the same or similar functions and having jurisdiction or cognizance of the Building and/or the Demised Premises.

(I) The term "repair" shall be deemed to include restoration and replacement as may be necessary to achieve and/or maintain good working order and condition, and the term "untenantable" shall be deemed to include being inaccessible.

(J) Reference to "termination of this Lease" includes expiration or earlier termination of the term of this Lease or cancellation of this Lease pursuant to any of the provisions of this Lease or pursuant to law. Upon a termination of this Lease, the term and estate granted by this Lease shall end at noon on the date of termination as if such date were the date of expiration of the term of this Lease and neither party shall have any further obligation or liability to the other after such termination (i) except as shall be expressly provided for in this Lease, or (ii) except for such obligations as by their nature or under the circumstances can only be, or by the provisions of this Lease, may be, performed after such termination, and, in any event, unless expressly otherwise provided in this Lease, any liability for a payment which shall have accrued to or with respect to any period ending at the time of termination shall survive the termination of this Lease.

(K) The term "in full force and effect" when herein used in reference to this Lease as a condition to the existence or exercise of a right on the part of Tenant shall be construed in each instance as including the further condition that at the time in question no default on the part of Tenant exists, and no event has occurred which has continued to exist for such period of time (after the notice, if any, required by this Lease), as would entitle Landlord in either such instance to terminate this Lease or to dispossess Tenant.

(L) The term "Landlord" as used in this Lease means only the owner, or other mortgagee in possession, for the time being of the land and Building (or the owner of a lease of the Building or of the land and Building) of which the Demised Premises form a part, so that in the event of any sale or sales of said land and Building or of said Lease, or in the event of a lease of said Building, or of the land and Building, the said Landlord shall be and hereby is entirely freed and relieved of all covenants and obligations of Landlord hereunder, and it shall be deemed and construed without further agreement between the parties or their successors in interest, or between the parties and the purchaser, at any such sale, or the said lessee of the Building, or of the land and Building, that the purchaser or the lessee of the Building has assumed and agreed to carry out any and all covenants and obligations of Landlord, hereunder.

(M) The term "Tenant" shall mean Tenant herein named or any assignee or other successor in interest (immediate or remote) of Tenant herein named, when Tenant herein named or such assignee or other successor in interest, as the case may be, is in possession of the Demised Premises as owner of the Tenant's estate and interest granted by this Lease, and also if Tenant is not an individual or corporation, all of the individuals, firms and/or corporations and other entities comprising Tenant.

(N) Words and phrases used in the singular shall be deemed to include the plural and vice versa, and nouns and pronouns used in any particular gender shall be deemed to include any other gender.

(O) The rule of "ejusdem generis" shall not be applicable to limit a general statement following or referable to an enumeration of specific matter or matters similar to the matters specifically mentioned.

(P) All references in this Lease to numbered Articles, numbered Sections and Subsections and lettered Exhibits are references to Articles and Sections and Subsections of this Lease, and Exhibits annexed to (and thereby made a part of) this Lease, as the case may be, unless expressly otherwise designated in the context.

(Q) The term "rent" or "rents" shall, except where the context expressly implies to the contrary, be deemed to mean fixed rent and additional rent as such terms are defined in Section 1.04.

36.02 The various terms which are defined in other Articles of this Lease or are defined in exhibits annexed hereto, shall have the meaning specified in such other Articles and such exhibits for all purposes of this Lease and all agreements supplemental thereto, unless the context shall otherwise require.

36.03 The Article headings in this Lease and Index prefixed to this Lease are inserted only as a matter of convenience or reference, and are not to be given any effect whatsoever in construing this Lease.

ARTICLE 37 SUBORDINATION AND MISCELLANEOUS

37.01 (A) This Lease, and all rights of Tenant hereunder, are and shall be subject and subordinate in all respects to all covenants, agreements and restrictions of record as of the date hereof, with respect to the land and/or the Building, including but not limited to those restrictions and conditions contained in the deed from the City of Stamford, Connecticut to Landlord or Landlord's predecessor in title. In said deed, Landlord agrees in part, that in the sale, lease or occupancy of the property, it would not effect or execute any agreement, lease, conveyance or other instrument whereby the property or any part thereof is restricted upon the basis of race, sex, religion, color or national origin, and it would comply with all state and local laws in effect from time to time prohibiting discrimination or segregation by reason of race, sex, religion, color or national origin.

(B) Tenant shall not take or permit any action with respect to the Demised Premises which would violate any covenants, agreements and restrictions of record with respect to the land and/or the Building.

37.02 (A) Promptly after the execution of this Lease, Landlord shall have the right to submit a copy of the Lease to Landlord's mortgagee for approval. If the Lease has not been approved by the mortgagee within thirty (30) days after its execution, then Landlord shall have the right to cancel this Lease by written notice to Tenant within five (5) days after the expiration of such thirty (30) day period. Upon such cancellation, neither party shall have any further liability to the other by reason of this Lease.

(B) If Landlord fails to cure any default on its part under this Lease, the holder of any superior mortgage shall have the right, at its option, to cure such default in order to prevent termination of this Lease by Tenant.

ARTICLE 38 PARTNERSHIP TENANT

38.01 If Tenant is a partnership (or is comprised of two (2) or more persons or entities individually or as co-partners of a partnership) or if Tenant's

interest in this lease shall be assigned to a partnership (or to two (2) or more persons or entities individually or as co-partners of a partnership) pursuant to Article 10 (any such partnership and such persons or entities are referred to in this Article as "Partnership Tenant"), the following provisions of this Article shall apply to such Partnership Tenant: (a) the liability of each of the parties comprising Partnership Tenant shall be joint and several, and (b) each of the parties comprising Partnership Tenant hereby consents in advance to, and agrees to be bound by, any modifications of this lease which may hereafter be made, and by any

40

notices, demands, requests or other communications which may hereafter be given, by Partnership Tenant or by any of the parties comprising Partnership Tenant, and (c) any bills, statements, notices, demands, requests and other communications given or rendered to Partnership Tenant or to any of the parties comprising Partnership Tenant shall be deemed given or rendered to Partnership Tenant and to all such parties and shall be binding upon Partnership Tenant and all such parties, and (d) if Partnership Tenant shall admit new partners, all of such new partners shall, by their admission to Partnership Tenant, be deemed to have assumed performance of all of the terms, covenants and conditions of this lease on Tenant's part to be observed and performed, and (e) Partnership Tenant shall give prompt notice to Landlord of the admission of any such new partners, and upon demand of Landlord, shall cause each such new partner to execute and deliver to Landlord an agreement in form satisfactory to Landlord, wherein each such new partner shall assume performance of all of the terms, covenants and conditions of this lease on Tenant's part to be observed and performed (but neither Landlord's failure to request any such agreement nor the failure of any such new partner to execute or deliver any such agreement to Landlord shall vitiate the provisions of subdivision (d) of this Section).

38.02 Provided Tenant is not then in default under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), and further provided that Clarus Corporation is still one of the entities comprising the Tenant hereunder, Landlord will, at Tenant's request, release Kanders & Company, Inc. from future liability under this Lease. This Section 38.02 shall be deemed deleted from this Lease upon the occurrence of the release of Clarus Corporation pursuant to Section 38.03 hereof.

38.03 At any time after the expiration of the eighteenth (18th) month following the Rent Commencement Date, provided Tenant is not then in default under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), and further provided that Kanders & Company, Inc. has not previously been released pursuant to Section 38.02, Landlord will, at Tenant's request, release Clarus Corporation from future liability under this Lease provided Kanders & Company, Inc. provides proof acceptable to Landlord in its reasonable discretion that Kanders & Company, Inc. has a net worth of at least \$10,000,000.00 at that time. Landlord agrees to accept as such proof bank references (including approximate current account balances) together with a letter from a certified public accountant stating the estimated net worth of Kanders & Company, Inc.

In the event that such release is requested after the expiration of the seventh (7th) Lease Year, then, in lieu of the net worth test set forth above, Kanders & Company, Inc. may provide proof acceptable to Landlord in its reasonable discretion that Kanders & Company, Inc. has free and clear cash on hand equal to at least three (3) times Tenant's cash requirements for the trailing twelve (12) month period, in which case Clarus Corporation may be released from future liability under this Lease.

ARTICLE 39 TENANT'S AUTHORITY TO ENTER LEASE

39.01 In the event that the Tenant hereunder is a corporation, Tenant represents that the officer or officers executing this lease have the requisite authority to do so.

ARTICLE 40 PARKING

40.01 (A) Provided Tenant is not in default under any material terms, conditions or covenants of this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), Tenant is entitled to twenty-four (24) parking cards for employees use for parking in undesignated spaces in the Landmark Square parking garage, twelve (12) of which shall be self-park cards and twelve (12) of which shall be valet parking cards, at the following rates:

<table></table>	
<caption></caption>	
Lease Year	Parking Rate
<s></s>	<c></c>
1-3	\$25.00 per card per month
4-10	\$40.00 per card per month
11-15	The then prevailing rate per card per month

 |Tenant must provide name and registration of vehicle of each person in its employment who will have the right to use said parking

41

cards and all such persons shall be subject to all rules and regulations as may be prescribed by Landlord from time to time and to any modifications and/or additions thereto regarding the use of said parking cards in the parking garage.

(B) All parking by Tenant's employees shall be on a first come, first serve basis.

(C) All parking garage spaces, ramps and driveways, walkways, lobbies and elevators used by Tenant, its employees and patrons will be specifically and exclusively at their own risk, and Landlord shall not be liable for any damage to any vehicle or its contents, resulting from theft, collision, vandalism or any other cause whatsoever or for harm or injury to any person from any cause whatsoever, the failure of any garage attendant or other personnel or device to patrol, monitor, guard or service such parking garage, and Landlord shall in no way be liable for any acts or omissions of such personnel, or device in failing to prevent any such theft, vandalism or loss or damage by other cause. Tenant's' indemnity in Article 21 hereof shall include the parking garage and all related parts thereof and thereto as though specifically set forth therein.

(D) There shall not be any overnight parking in the garage. If any automobile owned by Tenant or by its personnel or visitors remains in the parking garage overnight and the same interferes with the cleaning or maintenance of said area (including, without limitation, snow removal), any costs or liabilities incurred by Landlord in removing said automobile to effectuate cleaning or maintenance, or any damages resulting to said automobile or to Landlord's equipment or equipment owned by others by reason of the presence or removal of said automobile shall be paid by Tenant to Landlord, as additional rent.

(E) In the event that Tenant validates parking for any person parking in the garage, Tenant shall pay Landlord's then prevailing parking rate for each such validation within ten (10) days after being billed therefor.

(F) Tenant acknowledges that parking fees including, without limitation, fees for parking cards and validations, may be billed on Landlord's behalf by the garage operator or other agent designated by Landlord and that such fees shall be deemed additional rent under this Lease.

ARTICLE 41 SATELLITE DISH

41.01 Upon request by Tenant in accordance herewith during the term of this Lease, Tenant may (subject to the terms hereof), at its own cost and expense, install, operate, and maintain a satellite dish (hereinafter the "Satellite

Dish") on the roof of the Building in a location to be designated by Landlord. Such installation of the Satellite Dish shall be deemed "Tenant's Work" under Article 14 and shall be performed in strict accordance therewith.

41.02 Tenant's right to install and operate the Satellite Dish as set forth above shall be subject to the further provisions of this Article 41. Tenant shall submit to Landlord, along with its request to install the Satellite Dish, (i) plans and specifications for the installation of the Satellite Dish, (ii) copies of all necessary governmental and quasi-governmental permits, licenses and authorizations for the installation of such Satellite Dish, which the Tenant shall obtain at its own expense; and (iii) a certificate of insurance evidencing insurance coverage as required by this Lease and any other insurance reasonably required by the Landlord for the installation and operation of the Satellite Dish. Landlord may withhold its approval to the installation and operation of the Satellite Dish if such installation and/or operation of the Satellite Dish may damage the structural integrity of the Building, interfere with any service provided by the Landlord or any tenant, interfere with any then existing tenant's business or reduce the amount of rentable space in the Building, or if such installation, and/or operation shall be in violation of any of the terms hereof

41.03 The installation and size of such Satellite Dish shall meet the specifications of Landlord and the Federal Communications Commission and shall comply with all other governmental requirements (local, state and federal) and all reasonable requirements of Landlord. In no event shall the Satellite Dish be more than eighteen (18") inches in diameter. Tenant shall, at its own cost and expense, maintain and repair the Satellite Dish and keep same in good condition for as long as same is installed and remains.

41.04 Tenant covenants and agrees that neither the Tenant nor its agents will cause any damage to the roof during the installation and operation of the Satellite Dish.

41.05 If Landlord's insurance premium or Real Estate Taxes (or any other taxes) are increased as a result of the Satellite Dish or if

any other tax or charge is imposed on Landlord in connection therewith, Tenant shall pay such increase or such other tax or charge, as additional rent, upon receipt of a bill from the Landlord therefor.

41.06 Tenant shall have no right to an abatement or reduction in the amount of fixed rent or additional rent set forth in this Lease if for any reason the Tenant is unable to use the Satellite Dish.

41.07 Tenant covenants and agrees that the installation, operation and removal of the Satellite Dish and all related cabling and wiring will be at its sole risk and expense. Tenant agrees to indemnify, defend and hold harmless Landlord against all claims, actions, damages, liability and expenses (including reasonable attorneys' fees and disbursements) in connection with the loss of life, personal injury, damage to property or business or any other loss or injury arising out of the installation, operation or removal of the Satellite Dish, unless caused by Landlord and not covered by Tenant's insurance. Tenant agrees to indemnify Landlord for all costs and expenses (including reasonable attorneys' fees and disbursements) incurred as a result of any litigation concerning the Satellite Dish.

41.08 Landlord, at its sole option, may require the Tenant, at any time prior to the expiration of this Lease, to terminate the operation of the Satellite Dish and to remove the Satellite Dish at Tenant's sole expense if it is causing physical damage to the structural integrity of the Building, interfering with any other service provided by the Building, or interfering with the business of any other tenant that was operating in the Building prior to installation of Tenant's Satellite Dish.

41.09 At the expiration or sooner termination of this Lease or upon termination of the operation of the Satellite Dish by Landlord, Tenant or otherwise, Tenant shall be required to remove the Satellite Dish and all related cabling and

wiring from the Building, at its sole cost and expense, on or prior to such date. Tenant shall repair all damage attributable to the installation or removal of such Satellite Dish and all related cabling and wiring and shall leave the portion of the Building where the Satellite Dish was located in good order and repair, reasonable wear and tear excepted. If Tenant does not promptly remove the Satellite Dish and all related cabling and wiring when so required, Tenant hereby authorizes Landlord to remove and dispose of the Satellite Dish and all related cabling and wiring and expenses incurred thereby. Tenant agrees that Landlord shall not be liable for any property (including the Satellite Dish and all related cabling and wiring) disposed of or removed by Landlord in accordance herewith.

41.10 Access to the roof by Tenant's contractor to install the Satellite Dish shall be coordinated with Landlord's property manager. Tenant shall be permitted access to the roof for the purpose of maintenance, repair or replacement of the Satellite Dish during the hours of 9:00 a.m. to 5 p.m. Monday through Friday (except holidays) upon reasonable prior written notice to Landlord.

ARTICLE 42 RENEWAL OPTION

42.01 Tenant shall have the right, to be exercised as hereinafter provided, to extend the term of this Lease with respect to the entire Demised Premises for one (1) period of five (5) years (the "Renewal Term") upon the following terms and conditions:

(a) That at the time of the exercise of such right and at the commencement of the Renewal Term, Tenant shall not be in default in the performance of any of the terms, covenants or conditions which Tenant is required to perform under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any).

(b) That Tenant shall notify Landlord in writing that Tenant intends to exercise this option at least twelve (12) months prior to the Expiration Date.

(c) That the Renewal Term shall be upon the same terms, covenants and conditions as in this Lease provided, except that (i) there shall be no further option to extend this Lease beyond the Renewal Term referred to above; (ii) the Demised Premises shall be delivered in its then "as is" condition; and (iii) the fixed rent to be paid by Tenant during the Renewal Term shall be ninety-five (95%) percent of the fair market annual minimum rent (as hereinafter defined) during the first year of the Renewal Term, but in no event less than \$45.00 per rentable square foot per year, subject to escalations during each of the second through fifth years of the Renewal Term of three (3%) percent per annum over the fixed rent payable for the prior year.

"Fair market annual minimum rent" as such term is used in this Article 42 shall mean the rate then being received by landlords when entering into new leases for comparable space for a comparable term in Class A buildings in the Central Business

43

District of Stamford, Connecticut. "Fair market annual minimum rent" shall not mean "net effective rent to Landlord". In determining fair market annual minimum rent, no adjustment shall be made in consideration of and Tenant shall not be entitled to a credit for Tenant improvements, brokerage commissions, rent concessions and other concessions which Landlord may typically offer to other tenants.

(d) This Renewal Option is offered exclusively to Clarus Corp. and Kanders & Company, Inc. and shall not be transferable by operation of law or otherwise except to a permitted assignee of this Lease.

(e) Time shall be of the essence with respect to all of Tenant's obligations under this Article 42.

ARTICLE 43 CANCELLATION OPTION

43.01 (a) Provided Tenant has complied with all of the terms, covenants and conditions of this Lease and is not then in default of its obligations hereunder

beyond the expiration of the applicable notice and cure period provided herein for the cure thereof, if any, Tenant shall have the right to cancel this Lease as of the last day of the eighth (8th) or tenth (10th) Lease Year (in either case, the "Cancellation Date") by notifying Landlord in writing (the "Cancellation Notice"), at least twelve (12) months prior to the desired Cancellation Date of Tenant's intent to exercise this Cancellation Option and by delivering to Landlord, together with the Cancellation Notice, a check (subject to collection) in the amount of the applicable Cancellation Fee (as hereinafter defined). The "Cancellation Fee" shall mean the sum of: (i) (A) if the Cancellation Date is at the end of eighth (8th) Lease Year, an amount equal to the fixed rent that would have been payable hereunder during the twelve (12) calendar months following the Cancellation Date, or (B) if the Cancellation Date is at the end of the tenth (10th) Lease Year, an amount equal to the fixed rent that would have been payable hereunder during the nine (9) calendar months following the Cancellation Date; and (ii) the Unamortized Costs (as defined below). The term "Unamortized Costs", as used herein, shall be deemed to mean those portions of all costs incurred by Landlord in connection with this Lease that have not been amortized as of the Cancellation Date, which costs shall include, without limitation, each of the following: (y) construction, architectural, engineering and other costs and fees associated with the performance of any and all construction performed by Landlord or Landlord's contractors on behalf of Tenant (including the Landlord's Initial Work), and (z) \$191,820.00, representing the Broker's commission. For the purpose of calculating the Unamortized Costs, the total costs described above shall be amortized over the entire Term of this Lease as if same were a ten (10) year, self-amortizing loan at an annual interest rate of eight (8%) percent, payable in equal monthly installments of principal and interest combined. The amount of the Unamortized Costs shall be equal to the amount that would be outstanding under such loan as of the Cancellation Date.

(b) Upon satisfaction by Tenant of each of the above conditions, and upon the Demised Premises having been surrendered to Landlord and vacated by Tenant (in accordance with the provisions of Article 24 hereof) on or before the Cancellation Date as if that date were the Expiration Date under this Lease, this Lease shall be deemed canceled and terminated as of the Cancellation Date. Time is of the essence with respect to all time periods referenced in this Article 43. In the event that Tenant shall fail to fully and timely comply with each of the conditions herein contained, Tenant will be deemed to have waived all of its rights contained in this Article 43. Tenant acknowledges that, following exercise of this Cancellation Option by Tenant, Tenant may not revoke such cancellation without the prior written consent of Landlord (which may be granted or withheld in Landlord's sole discretion).

(c) This Cancellation Option is offered exclusively to Clarus Corp. and Kanders & Company, Inc. and shall not be transferable by operation of law or otherwise except to a permitted assignee of this Lease.

44

ARTICLE 44 RIGHT OF FIRST OFFER

44.01 (a) In the event that Landlord, at its option, obtains the legal right to lease the penthouse space or any portion thereof located directly above the Demised Premises consisting of approximately 1,500 square feet (the "Penthouse Space"), and desires to lease the Penthouse Space, then, provided Tenant is not then in default under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), Landlord shall notify Tenant of the availability of the Penthouse Space ("Landlord's Notice"), provided that Landlord shall not be liable to Tenant for any costs, expenses, damages or liabilities which are or may be incurred by Tenant by reason of Landlord's unintentional failure to so notify Tenant. Tenant shall have a period of five (5) days after receipt of Landlord's Notice in which to notify Landlord that Tenant will either lease the Penthouse Space ("Tenant's Notice"), time being of the essence with respect to all of Tenant's obligations hereunder.

(b) In the event that Tenant elects not to lease the Penthouse Space or fails to deliver Tenant's Notice in strict accordance with the terms hereof (in which case Tenant shall be deemed to have elected not to lease the Penthouse Space), Landlord shall thereafter not be permitted to lease the Penthouse Space

to any other party.

(c) Tenant's exercise of this Right of First Offer by the giving of Tenant's Notice to Landlord shall be self-operative and no additional document of confirmation of Tenant's exercise of this Right of First Offer shall be necessary. Notwithstanding the foregoing, at Landlord's option, Landlord and Tenant shall execute a lease modification agreement (the "Lease Modification Agreement") to confirm Tenant's exercise of this Right of First Offer. In the event that Tenant properly and timely exercises this Right of First Offer as provided above, Tenant shall lease the Penthouse Space from Landlord upon all of the same terms as this Lease, except: (i) that the fixed rental rate for the Penthouse Space shall be equal to the fair market annual minimum rent (as defined below), (ii) for other matters dependent upon the size of the Penthouse Space, such as Tenant's Proportionate Share, which shall be adjusted accordingly, (iii) that Tenant will accept the Penthouse Space in its "as is" condition and Landlord shall not be required to perform any work in or to the Penthouse Space or incur any expense in order to prepare such space for Tenant's occupancy, and (iv) for such other terms and conditions as may be mutually agreed to by Landlord and Tenant.

"Fair market annual minimum rent" as used in this Article 44 shall mean the rate then being received by landlords when entering into new leases for comparable size space for a comparable term in Class A buildings in the Central Business District of Stamford, Connecticut. "Fair market annual minimum rent" as used in this Article 44 shall not mean "net effective rent to Landlord". In determining fair market annual minimum rent, adjustment shall be made in consideration of factors then customarily taken into account in determining fair market annual minimum rent, including, without limitation, the improvements which Tenant agrees to make to the Penthouse Space at Tenant's expense.

(d) This Right of First Offer is offered exclusively to Clarus Corp. and Kanders & Company, Inc. and shall not be transferable by operation of law or otherwise except to a permitted assignee of this Lease.

ARTICLE 45 PENTHOUSE SPACE

45.01 (a) In the event that Tenant desires to lease the entire Penthouse Space (as such term is defined in Article 44 hereof) and Tenant desires to take all actions necessary to make the Penthouse Space legally rentable at Tenant's sole cost and expense (including, without limitation, all costs of permits, approvals, construction, legal expenses, etc.), then, provided Tenant is not then in default under this Lease beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), Tenant may notify Landlord that Tenant will take such actions and thereafter lease the Penthouse Space pursuant to the terms of this Article 45. Landlord will cooperate with Tenant at Tenant's expense (including Landlord's attorneys' fees and disbursements) in order to facilitate Tenant's efforts pursuant to this Article 45. All provisions of Article 14 of this Lease shall be applicable to any Alterations to be performed by Tenant with respect to the Penthouse Space.

(b) Upon exercise of its rights under Section 45.01, Landlord and Tenant shall execute a lease modification agreement to confirm that Tenant will lease the Penthouse Space from Landlord upon all of the same terms as this Lease, except: (i) that the fixed rental rate for the Penthouse Space shall be equal to the fair market annual minimum rent (as defined below), (ii) for other

45

matters dependent upon the size of the Penthouse Space, such as Tenant's Proportionate Share, which shall be adjusted accordingly, (iii) that Tenant will accept the Penthouse Space in its "as is" condition and Landlord shall not be required to perform any work in or to the Penthouse Space or incur any expense in order to prepare such space for Tenant's occupancy, and (iv) for such other terms and conditions as may be mutually agreed to by Landlord and Tenant.

"Fair market annual minimum rent" as used in this Article 45 shall mean the rate then being received by landlords when entering into new leases for comparable size space for a comparable term in Class A buildings in the Central Business District of Stamford, Connecticut. "Fair market annual minimum rent" as used in this Article 45 shall not mean "net effective rent to Landlord". In determining fair market annual minimum rent, adjustment shall be made in consideration of factors then customarily taken into account in determining fair market annual minimum rent, including, without limitation, the improvements which Tenant agrees to make to the Penthouse Space at Tenant's expense.

(c) This Article 45 is applicable only to Clarus Corp. and Kanders & Company, Inc. and shall not be transferable by operation of law or otherwise except to a permitted assignee of this Lease.

ARTICLE 46 BUILDING RENOVATIONS

46.01 Landlord will use commercially reasonable efforts to complete the following renovations at the Building within the time periods provided: (a) by December 31, 2004, paint the lobby, remove the original Building directory from the lobby, install a security card access system for the Building, install lighting in the elevator portion of the lobby and renovate the two (2) cubby spaces behind each set of elevator cabs; and (b) by June 30, 2004, repair the peeling paint on the northeast portion of the facade of the curtain wall of the Building.

46

IN WITNESS WHEREOF, the parties hereto have caused this instrument to be executed by their duly authorized officers hereunder affixed the date and year first above written.

<C>

<TABLE> <CAPTION> <S> SIGNED AND DELIVERED IN THE PRESENCE OF:

RECKSON OPERATING PARTNERSHIP, L.P. BY: RECKSON ASSOCIATES REALTY CORP., GENERAL PARTNER

BY: /s/ Marion Schiraldi Name: Marion Schiraldi

Its: Senior Vice President and Managing Director

CLARUS CORPORATION

BY: /s/ Nigel P. Ekern Name: Nigel P. Ekern Its: Chief Administrative Officer

KANDERS & COMPANY, INC.

BY: /s/ Warren B. Kanders Name: Warren B. Kanders Its:President

</TABLE>

47

EXHIBIT A RENTAL PLAN

EXHIBIT B CLEANING SCHEDULE

Subject to Section 18.02, Landlord covenants and agrees to provide the following basic cleaning services to the Demised Premises:

Daily (5 days per week Monday to Friday, legal holidays excepted):

- * Sweep hard surface floors with treated dust mop
- Empty and wipe all ash trays.
 Empty wests paper backets
- Empty waste paper baskets
 Low dusting
- Low dusting
- * Vacuum all carpeted areas

Weekly:

* Damp mop all hard surface floors.

Every six (6) months:

Clean windows inside and out

RECYCLING

Tenant is advised that the Building is subject to a mandatory recycling program for disposables and waste materials. Tenant shall abide by all rules and regulations as may be promulgated for compliance with the program.

49

EXHIBIT C RULES AND REGULATIONS

1. The sidewalks, entrances, passages, lobby, elevators, vestibules, stairways, corridors or halls outside the Demised Premises shall not be obstructed or encumbered by any tenant or used for any purpose other than ingress and egress to and from the Demised Premises and Tenant shall not permit any of its employees, agents or invitees to congregate in any said areas. No door mat of any kind whatsoever shall be placed or left in any public hall or outside and entry door of the Demised Premises.

2. No awnings or other projections shall be attached to the outside walls of the Building. No curtains, blinds, shades or screens shall be attached to or hung in, or used in connection with any window or door of the Demised Premises without the prior written consent of Landlord. Such curtains, blinds, shades or screens must be of a quality type, design and color, and attached in the manner, approved by Landlord.

3. No sign, insignia, advertisement, object, notice or other lettering shall be exhibited, inscribed, painted, or affixed by any tenant on any part of the outside of the Demised Premises or the Building without the prior written consent of Landlord. In the event of the violation of the foregoing by any tenant, Landlord may remove the same without any liability, and may charge expense incurred in such removal to the tenant or tenants violating this rule. Interior signs and lettering on doors and directory tablet outside of the Demised Premises shall, if and when approved by Landlord, be inscribed, painted or affixed for each tenant by Landlord at the expense of such tenant, and shall be of a size, color and style acceptable to Landlord.

4. The sashes, sash doors, skylights, windows and doors that reflect or admit light and air into the halls, passageways or other public places in the Building shall not be covered or obstructed by Tenant, nor shall any bottles, parcels, or other articles be placed on the window sills.

48

5. No showcases or other articles shall be put in front of or affixed to any part of the exterior of the Building, nor placed in the halls, corridors or vestibules outside the Demised Premises.

6. The water and wash closets and other plumbing fixtures shall not be used for any purpose other than those for which they were designed or constructed, and no sweepings, rubbish, rags, acids or other substances shall be thrown or deposited therein. All damages resulting from any misuse of the fixtures shall be borne by the tenant who, or whose servants, employees, agents, visitors or licenses shall have caused the same.

7. No boring, cutting or stringing of wires in violation of applicable laws, codes or regulations shall be permitted, except with the prior written consent of Landlord, and as Landlord may reasonably direct. No tenant shall lay linoleum, or other similar floor covering, so that the same shall come in direct contact with the floor of the Demised Premises.

8. No bicycles, vehicles, animals, fish or birds of any kind shall be brought into or kept in or about the premises.

9. No noise, including, but not limited to, music or the playing of musical instruments, recordings, radio, or television which, in the judgment of Landlord, might disturb other tenants in the Building, shall be made or permitted by any Tenant. Nothing shall be done or permitted in the Demised Premises by Tenant which would unreasonably impair or interfere with the use or enjoyment by any other tenant of any other space in the Building. No tenant shall throw anything out of the doors, windows or skylights or down the passageways.

10. Tenant, its servants, employees, agents, visitors or licensees, shall not at any time bring or keep upon the Demised Premises any explosive fluid, chemical or substance, nor any inflammable or combustible objects or materials, in violation of any applicable law, ordinance or governmental regulation.

11. Additional locks or bolts of any kind which shall not be operable by the Grand Master Key for the Building shall not be placed upon any of the doors or window by any tenant, nor shall any changes be made in locks or the mechanism thereof which shall make such locks inoperable by said Grand Master Key. Each tenant shall, upon the termination of its tenancy, turn over to the Landlord all keys of stores, offices and toilet rooms, either furnished to, or otherwise provided by, such tenant and in the event of the loss of any keys furnished by Landlord, such tenant shall pay to Landlord the cost thereof. Tenant shall have the right to install additional security systems for the Demised Premises, which systems will be coordinated with those operated by Landlord and its managing agent. Keys

50

or cards used in connection with such systems shall be furnished to Landlord or its managing agent.

12. All removals, or the carrying in or out of any safes, freight, furniture, packages, boxes, crates or any other object or matter of any description must take place during such hours and in such elevators as Landlord or its agent may reasonably determine from time to time. Landlord reserves the right to inspect all objects and matter to be brought into the Building and to exclude from the Building all objects and matter which violate any of these Rules and Regulations or the Lease of which these Rules and Regulations are a part. Landlord may require any person leaving the Building with any package or other object or matter to submit a pass, listing such package or object or matter, from the tenant from whose premises the package or object or matter is being removed, but the establishment and enforcement of such requirement shall not impose any responsibility on Landlord for the protection of any tenant against the removal of property from the premises of such tenant. Landlord shall, in no way, be liable to Tenant for damages or loss arising from the admission, exclusion or ejection of any person to or from the Demised Premises or the Building under the provisions of this Rule 12 or of Rule 16 hereof.
13. Tenant shall not occupy or permit any portion of the Demised Premises to be occupied as an office for a public stenographer or public typist, or for the warehousing, manufacture or sale to the general public of beer, wine, liquor, narcotics, or as a barber, beauty or manicure shop, or as an employment bureau. Tenant shall not engage or pay any employees on the Demised Premises, except those actually working for Tenant or its affiliates. Tenant shall not use the Demised Premises or any part thereof, or permit the Demised Premises or any part thereof to be used, for manufacturing, or sale at auction of merchandise, goods or property of any kind, except for promotional purposes.

14. Tenant shall not obtain, purchase or accept for use in the Demised Premises cleaning, floor polishing or other similar services from any persons not authorized by Landlord in writing to furnish such services, provided always that the charges for such services by persons authorized by Landlord are not excessive. Such services shall be furnished only at such hours, in such places, within the Demised Premises, and under such regulations as may be fixed by Landlord. Tenants shall not purchase or contract for waxing, rug shampooing, Venetian blind washing, furniture polishing, lamp servicing, cleaning of electric fixtures, removal of garbage or towel service in the Demised Premises except from companies or persons approved by the Landlord. Landlord will be reasonable in its decision.

15. Landlord shall have the right to prohibit any advertising, or identifying sign by any tenant which in Landlord's reasonable judgment tends to impair the reputation of the Building or its desirability as a building for offices, and upon written notice from Landlord, such tenant shall refrain from or discontinue such advertising or identifying sign.

16. Landlord reserves the right to exclude from the Building during hours other than Business Hours (as defined in the foregoing Lease) all persons who do not present a pass to the Building signed by Landlord. All persons entering and/or leaving the Building during hours other than Business Hours may be required to sign a register. Landlord will furnish passes to persons for whom any tenant requests same in writing. Tenant, provided Tenant has such pass, will be permitted access to the Building and the Demised Premises 24 hours a day, seven days a week.

17. All entrance doors in the Demised Premises shall be left locked by Tenant when the Demised Premises are not in use. Entrance doors shall not be left open at any time. Landlord or its agents or contractors will turn off lights upon completion of cleaning services.

18. Unless Landlord shall furnish electrical energy hereunder as a service included in the rent, Tenant shall, at Tenant's expense, provide artificial light and electrical energy for the employees of Landlord and/or Landlord s contractors while doing janitor service or other cleaning in the Demised Premises and while making repairs or alterations in the Demised Premises.

19. The Demised Premises shall not be used for lodging or for any illegal purpose.

20. The requirements of tenants will be attended to only upon application at the office of the Building. Employees of Landlord shall not perform any work or do anything outside of their regular duties, unless under special instructions from Landlord.

21. Canvassing, soliciting and peddling in the Building are prohibited and each tenant shall cooperate to prevent the same.

51

22. There shall not be used in any space or in the public halls of the Building, either by any tenant or by jobbers or any others, in the moving or delivery or receipt of safes, freight, furniture, packages, boxes, crates, paper, office material, or any other matter or thing, any hand trucks except those equipped with rubber tires, side guards and such other safeguards as Landlord shall reasonably require. No hand trucks shall be used in passenger elevators, and no such passenger elevators shall be used for the moving, delivery or receipt of the aforementioned articles.

23. Tenant shall not cause or permit any odors of cooking or other processes or any unusual or objectionable odors to emanate from the Demised Premises which would annoy other tenants or create a public or private nuisance. No cooking shall be done in the Demised Premised except as is expressly permitted in the foregoing Lease.

24. Tenant shall cooperate with Landlord in obtaining maximum effectiveness of the cooling system and if requested by Landlord shall lower and close blinds, drapes and curtains when the sun's rays fall directly on the windows of the Demised Premises.

25. Subject to the provisions of Section 13.01, Landlord in its reasonable judgment, reserves the right to rescind, alter or waive any rule or regulation at any time prescribed for the Building when, in its reasonable judgment, it deems it necessary or desirable for the reputation, safety, care or appearance of the Building, or the preservation of good order therein, or the operation or maintenance of the Building or the equipment thereof, or the comfort of tenants or others in the Building.

26. Cigarette, pipe and cigar smoking is allowed only in tenant's demised premises. Smoking is prohibited in all interior public areas including but not limited to: public corridors, elevators, elevator lobbies and toilet rooms.

27. Tenant shall provide and maintain, at its expense, the hand-held fire extinguishers that are required to be maintained in the Demised Premises by the governmental agency having jurisdiction over this matter.

52

EXHIBIT D WORK LETTER

Tenant has examined the Demised Premises and, except for Landlord's Initial Work, Tenant agrees to accept possession of the Demised Premises in its "as is" condition which shall exist on the date of this Lease, and further agrees that, except as otherwise expressly set forth herein, Landlord shall have no obligation to perform any work, supply any materials, incur any expenses or make any installations, in order to prepare the Demised Premises for Tenant's occupancy.

Tenant shall perform Tenant's Initial Work, at Tenant's cost and expense (subject to the Tenant Improvement Allowance described herein).

1. (A) Tenant shall submit final and complete dimensioned and detailed plans and drawings of partition layouts, including openings, ceiling and lighting layouts, finishes and colors, engineering drawings, and any and all other information as may be necessary to complete Tenant's Initial Construction, which plans shall be at Tenant's sole cost and expense (all herein called the "Plans"). The Plans shall be 1' 0" = 1/8" scale.

(B) All Plans shall be submitted to Landlord for review and Landlord shall approve the Plans, or approve same as noted, within fifteen (15) business days after submission thereof to Landlord.

(C) Any architect, engineer, designer or consultant acting for or on behalf of Tenant shall be deemed an agent of, and authorized to bind, Tenant in all respects. Tenant or Tenant's contractor shall submit all Plans which require any approval or permit, to the governmental authority having jurisdiction thereover and Tenant shall pay any and all costs for obtaining permits or filing Plans.

(D) All plans (including the Plans) with respect to the Demised Premises and required to be submitted by Tenant, shall comply and conform with the Building plans filed with the Building Department of the City of Stamford, and with all the rules, regulations and other requirements of any governmental department having jurisdiction over the construction of the Demised Premises. Any changes required by any governmental department affecting the construction of the Demised Premises, at the sole cost and expense of Tenant, shall be complied with by Tenant in completing the Demised Premises.

2. (A) Intentionally Omitted.

(B) The term "tenant's delay" shall have the meaning ascribed thereto in Section 4.02 of this Lease.

(C) Tenant does herewith agree that in the event substantial completion of Tenant's Initial Construction in the Demised Premises pursuant to this Exhibit and/or any further agreements pertaining thereto, entered into between Landlord and Tenant, is delayed by any tenant's delay, Tenant shall, in addition to the provisions of Section 4.02 of this Lease, pay the reasonable costs and damages Landlord may sustain by reason of such tenant's delay.

3. Intentionally Omitted.

4. (A) Tenant shall be responsible, at Tenant's sole cost and expense, for the installation and removal of all computer and telephone equipment and related wiring, as well as interior decorative treatments ("Tenant's Installations"). All Tenant Installations shall be in conformity with the regulations promulgated by the State of Connecticut and City of Stamford Fire Marshall's Offices.

(B) Without limiting the generality of any other provisions of this Lease, any entry by Tenant in or on the Demised Premises shall be at Tenant's sole risk and, upon request of Landlord, Tenant shall pay for and deliver to Landlord, policies and certificates of insurance in amounts and with such companies as shall be reasonably satisfactory to Landlord, such as, but not limited to General Comprehensive Commercial Liability, Property Damage and Workers' Compensation to protect Landlord and Tenant during the period of making such Tenant's Installations. Landlord shall be named as an additional insured in such policies and the same shall be continued in effect, by Tenant at its cost and expense, during the period of the performance of

53

Tenant's Installations.

(C) All Tenant's Installations shall be in accordance with the rules and regulations of any governmental department or bureau having jurisdiction thereover, and shall not conflict with or be in violation of, or cause any violation of Landlord's basic building plans and/or the construction of the Building, and all Tenant's Installations shall be completed free of all liens and encumbrances. All permits which may be required by Tenant for Tenant's Installations shall be procured and paid for by Tenant only after having obtained Landlord's written approval of such work, or, if Landlord shall deem the same advisable, Landlord may procure such permit and Tenant shall pay for the same. No plans and/or specifications required to be filed by Tenant pursuant to any work contemplated to be performed by it within the Demised Premises shall be filed or submitted to any governmental authority having jurisdiction thereover without first having obtained Landlord's approval of the same.

(D) Intentionally Omitted.

(E) Tenant agrees that should Tenant enter upon the Demised Premises for the purposes of performing any work, the labor employed by Tenant or anyone performing such work for or on behalf of Tenant shall always be harmonious and compatible with the labor employed by Landlord or any contractors or subcontractors of Landlord. Should a labor dispute or disruption occur, Landlord may require Tenant to withdraw from such premises until the completion of any work being performed by Landlord in the Demised Premises.

(F) In the event Tenant or Tenant's contractor shall enter upon the Demised Premises or any other part of the Building, as may be above permitted by Landlord, Tenant agrees to indemnify and save Landlord free and harmless from and against any and all damages, claims costs and expenses, arising from or claimed to arise from any act, or for any other reason whatsoever arising out of said entry or such work.

(G) Any costs for services incurred or repairs required due to Tenant's contractors or Tenant's Installations shall be paid for by Tenant.

5. Intentionally Omitted.

6. Landlord shall contribute up to a maximum of \$460,000.00, plus additional sums equal to the actual out-of-pocket cost of the installation of a

building standard sprinkler system in the Demised Premises and the actual, out-of-pocket cost of the installation of sixty (60) tons of base building HVAC to service the Demised Premises (specifically limited to providing (i) the HVAC units and the main trunk line into the Demised Premises with respect to the HVAC units to be installed within the core of the Demised Premises, and (ii) the HVAC units only (i.e. without the distribution system therefor) with respect to the HVAC units to be installed above the drop ceiling), towards the cost of Tenant's Initial Work (the "Tenant Improvement Allowance"). Tenant shall, as Tenant's Initial Work progresses, submit statements to Landlord from time to time, but not more often than once per month, setting forth the cost of those aspects of the Tenant's Initial Work which have been completed through the date of such statement (such statement to be prepared in accordance with standard progress payment application forms issued by the American Institute of Architects). Each such statement shall be accompanied by a certificate from an authorized officer of Tenant requesting reimbursement for sums paid or to be paid (such requested amount, the "Requested Allowance Payment") and certifying that: (i) the Requested Allowance Payment is then due and payable by Tenant or has theretofore been paid by Tenant: and (ii) the intended use thereof is for satisfaction of amounts paid or amounts due and owing Tenant's contractor in connection with Tenant's Initial Work. Following receipt of any such statement and accompanying certificate, Landlord shall pay to Tenant (or to Tenant's contractor, if so directed by Tenant) a portion of the Tenant Improvement Allowance in an amount equal to ninety (90%) percent of the subject Requested Allowance Payment (but in no event shall the aggregate amount of such payments by Landlord exceed the Tenant Improvement Allowance). The ten (10%) percent of each Requested Allowance Payment not so paid to Tenant shall be retained by Landlord (subject to the following provisions of this Section), and the total amount so retained is hereinafter collectively referred to as the "Retained Allowance". Provided that no event of default shall have occurred and be continuing

54

hereunder beyond the expiration of the applicable notice and cure period provided herein for the cure thereof (if any), Landlord shall pay the entire Retained Allowance to Tenant, following Tenant's written request therefor and Landlord's receipt of the following materials: (A) a certification from Tenant's architect or engineer that Tenant's Initial Work has been substantially completed in strict accordance with the final construction documents approved by Landlord and that Tenant's Initial Work has been performed and completed in strict accordance with the requirements and provisions of this Lease and all governmental permits, licenses and authorizations required for Tenant's Initial Work; (B) a certificate from an authorized officer of Tenant certifying: (i) the total cost of Tenant's Initial Work (which certification shall substantiate such total cost to the reasonable satisfaction of Landlord); and (ii) that Tenant's Initial Work has been fully paid for or will be fully paid for upon receipt of the Retained Allowance; (C) lien waivers from all contractors and subcontractors which performed all or any portion of Tenant's Initial Work, stating that they have been fully paid for the work completed by such contractors or subcontractors; and (D) a certificate of occupancy for the Demised Premises. Upon delivery of such certificate, evidence and other documentation, Tenant shall have the reasonable right to control the release and payment of the Retained Allowance to its contractors, subcontractors and materialmen. Prior to paying the Retained Allowance, Landlord shall have the right, provided Landlord exercises such right within ten (10) days after Tenant's request for the Retained Allowance, to (a) inspect all of Tenant's Initial Work to determine if it has been completed substantially in accordance with the final construction documents, and (b) review all invoices and evidence of payment thereof for Tenant's Initial Work, to the extent such payments have been made. To the extent such payments have not been made, Landlord shall have the right (but not the obligation) to make such payments directly to such contractors, subcontractors or materialmen. The right to receive reimbursement for the cost of Tenant's Initial Work as set forth herein shall be the exclusive benefit of Tenant, it being the express intent of the parties hereto that in no event shall such right be conferred upon or for the benefit of any third party, including without limitation, any contractor, subcontractor, materialman, laborer, architect, engineer, attorney or any other person, firm or entity. If the total charges for Tenant's Initial Work are less than the maximum amount of the Tenant Improvement Allowance set forth above, Tenant shall not be entitled to receive the remainder

of such Tenant Improvement Allowance in any form. Tenant hereby acknowledges that, except as hereinafter provided, in no event shall any portion of the Tenant Improvement Allowance be paid or applied against any "soft costs". The term "soft costs", as used herein, shall generally include, without limitation, the fees and charges of any architects, engineers and other consultants engaged by Tenant in connection with Tenant's Initial Work; the fees and charges incurred in connection with obtaining governmental and quasi-governmental permits, authorizations and approvals; the costs and charges incurred in connection with the installation of Tenant's data and telecommunication wiring and cabling in and about the Demised Premises (or any portion thereof); and the costs and expenses incurred by Tenant in connection with the acquisition, transport and/or installation of Tenant's personnel, furniture, fixtures, equipment or decorations in the Demised Premises (or any portion thereof). Notwithstanding the foregoing, Tenant shall be permitted to apply up to \$46,000.00 of the Tenant Improvement Allowance in the aggregate towards architectural and engineering costs.

7. Telephone and computer installation and all costs therefor shall be the sole responsibility of Tenant, and Tenant shall make all necessary arrangements. Such telephone installations are subject to approval by Landlord and must be performed in strict compliance with all government codes, including a low voltage permit from the City of Stamford. At the expiration or earlier termination of the Lease, Tenant shall be responsible for the removal of all telecommunications and data wiring and any affiliated equipment.

8. Tenant shall, within thirty (30) days after the Rent Commencement Date, install vertical blinds as approved by Landlord, at Tenant's sole cost and expense on all exterior windows; such blinds shall be G71 vertical blinds in fabric kinetics, #0112, alabaster, and no other type of window covering will be used.

INSTRUCTIONS TO MOVING CONTRACTORS

1. The directions of the Landlord and/or his managing agent will be followed at all times.

2. No furniture and/or building materials will be moved in or out of the building from 7:00 a.m. to 6:00 p.m., Monday through Friday, unless approved by the Landlord and/or his managing agent.

3. The moving contractor must submit, not later than two weeks prior to the move, a written schedule which indicates the date and time the move will commence and also the same for the completion of the move.

55

4. All routes over finished floors will be protected with masonite, plywood or similar material runway, which is to be picked up at the close of work each day.

5. Appropriate warning signs are to be posted in all public corridors and lobbies used.

6. Temporary staging of furniture and equipment in public areas is not permitted.

7. All areas traveled are to be broom cleaned at the close of each day. Elevators are to be swept and debris carried from the car, NOT swept across the door opening.

8. Workmen should use the toilet facilities provided by the Landlord or Managing Agent.

9. The load limit of 3,000 pounds in the passenger elevator is NOT to be exceeded.

10. Generally only two (2) trailers will be allowed at the Loading Dock. Arrangements to accommodate more than two (2) must be made in advance with the Property Management Office.

11. Only rubber wheeled dollies and carts, in good operating condition, may be used. Excess oil and grease must be removed from wheels to prevent

staining flooring.

12. Reasonable care must be taken at all times to avoid any personal injury or property damage.

13. All packing and crating materials must be removed at the end of each day, and NOT be left to accumulate over night (fire hazard).

14. The moving contractor must utilize labor that will work in harmony with other labor in the building. In addition, Landlord's office should receive not later than two weeks prior to move, insurance certificates evidencing the following minimum coverages:

- (i) Insurance carrier must have a Best's rating of A VIII or better.
- (ii) Certificates of insurance must be provided prior to the commencement of any work performed.
- (iii) Type of insurance:

<tab< th=""><th></th><th></th><th></th><th></th><th></th></tab<>					
<cap< td=""><td>ΓION></td><td></td><td></td><td></td><td></td></cap<>	ΓION>				
<s></s>	<c> <<</c>	C>	<c></c>		
	A. Commercial General Liab	ility:			
	General Aggregate	l Aggregate \$2,000,0			00
	Products/Completed Operations Aggregate			\$1,000,000	
	Personal & Advertising Inj	ury, pe	er occur	rence	\$1,000,000
	Each Occurrence \$1,000,00			00	
	Fire Damage Liability \$ 50,		50,00	00	
	Medical Expense		\$	5,000	
	Broad Form Property Dam	age Inclu		cluded	
	Explosion, Collapse & Une	dergrou	ınd Haz	ard	Included
	B. Automobile Liability:				
	Bodily Injury/Property Damage Liability				

Each Occurrence (Combined Single Limit) \$1,000,000

C. Excess or Umbrella Liability:

Each Occurrence \$5,000,000 (Coverage to be at least on a "follow form" basis)

</TABLE>

56

D. Worker's Compensation & Employer's Liability:

Per statutory requirements of the state in which work is to be provided or performed.

E. Connecticut State Short Term Disability:

Per statutory requirements of the state in which work is to be provided or performed.

All certificates are to stipulate that ten (10) days prior notice of cancellation will be given to the Tenant and to:

RECKSON ASSOCIATES REALTY CORP. 360 Hamilton Avenue White Plains, New York 10601

INSTRUCTIONS TO TENANT'S CONTRACTORS

1. It shall be Tenant's contractor's responsibility to schedule the performance of his work and notify the Landlord's Property Manager of his proposed schedule, so that the contractor's elevator usage for material deliveries and rubbish removal may be coordinated with the over-all project (reserved) hoisting usage.

Elevator use for construction and moves may only occur during the hours of 6:00 P.M. through 7:00 A.M. on business days and only with prior notice and approval of the Landlord. Elevator use on non-business days may occur at any time and again only with prior notice and approval of the Landlord. Tenant is responsible for all building standard charge normally associated with such use.

2. The Tenant's contractor shall notify Landlord's Property Manager at least four (4) weeks prior to his proposed starting date to perform Tenant's Work (two (2) weeks prior to the proposed starting date for Tenant's Initial Work) and at that time will discuss the arrangements and requirements of his schedule. The following items will be discussed to determine the scheduled reservation time:

- a. Material delivery, schedule dates times
- b. Number of vehicles
- c. Elevator Service and hoist reservation time
- d. Docking arrangements and reservation time
- e. Insurance requirements
- f. Names and telephone numbers of contact and coordinator
- g. General instructions rules and regulations

3. The Tenant's contractor shall confirm his schedule with the Landlord's Property Manager, not less than 48 hours in advance of his pre-scheduled material deliveries. It shall be Tenant's contractor's sole responsibility to confirm his reservation times, and in the event that the confirmation is not verified and re-executed it shall be deemed that his reservations are to be voided (cancelled) and allocated to others. He shall then be required to re-schedule both his deliveries and his reservations through the Landlord's Property Manager. In all fairness to the other Tenants going into the building, if Tenant's contractor fails to meet or confirm the date, his contractor will have to wait until there is free time in the material delivery and hoisting schedule before they will be allowed to perform Tenant's Work. Landlord's Property Manager will make every effort to accommodate the Tenant's contractor as early as possible, but it is very likely that to re-schedule would effect a serious time delay; you can clearly see, then, that it is extremely important to confirm the schedule not less than 48 hours in advance, and if possible, preferably three to five days ahead of time.

4. Tenant and its contractors shall remain responsible for the scheduling and transportation of material and equipment used in the performance of Tenant's Work and for the removal from the Building of waste and debris resulting from the performance of Tenant's Work, and Landlord shall not be responsible for coordination of the work of Tenant's contractors with the work of Landlord's contractors. However, Landlord and Tenant shall cooperate in their respective performances of Landlord's and Tenant's Work in order to enable the same to be properly coordinated. Tenant shall not be under any obligation to employ any of Landlord's contractors or to pay any charge to any of them by reason of Tenant's having other contractors or purchasing any materials or labor or employing any labor from other sources. Tenant and its contractors shall not be under any obligation to pay for water, electricity, heat, ventilation or cooling provided in the Premises during the performance of any of Tenant's Work during normal working hours of the Building construction project.

5. Temporary staging of materials and equipment in public areas is not permitted.

6. Should large equipment or materials need to be transported via dollies and/or carts, then the contractor transporting such equipment or

materials shall protect all routes over finished floors with a minimum of 3/8" plywood runway, which will be picked up at the close of work each day.

7. All areas traveled are to be broom cleaned at the close of each day. Elevators are to be swept and debris carried from the car, NOT swept across the door opening.

8. Workmen should use the toilet facilities provided by the general contractor.

58

9. The hoisting load limit of 3,000 pounds in passenger elevator is NOT to be exceeded.

10. All packing and crating materials must be removed at the end of each day, and not be left to accumulate over night (fire hazard).

11. The Tenant's contractor must utilize labor that will work in harmony with other labor in the Building. In addition, Landlord's Property Manager should receive not later than two weeks prior to contractor's performance of Tenant's Work, insurance certificates evidencing the following minimum coverages:

(i) Insurance carrier must have a Best's rating of A VIII or better.

(ii) Certificates of insurance must be provided prior to the commencement of any work performed.

(iii) Type of insurance:

<table> <caption></caption></table>					
<> <c> <c> <c> <c> <</c></c></c></c>					
A. Commercial General Liability:					
General Aggregate \$2,000,000 Products/Completed Operations Aggregate \$1,000,000					
Personal & Advertising Injury, per occurrence \$1,000,000					
Each Occurrence \$1,000,000					
Fire Damage Liability \$ 50,000					
Medical Expense \$ 5,000					
Broad Form Property Damage Included					
Explosion, Collapse & Underground Hazard Included					
B. Automobile Liability:					
Bodily Injury/Property Damage Liability					
Each Occurrence (Combined Single Limit) \$1,000,000					
C. Excess or Umbrella Liability:					
Each Occurrence \$5,000,000 (Coverage to be at					
least on a "follow form" basis)					

| D. Worker's Compensation & Employer's Liability: | | | | |
| Per statutory requirements of the state in which work is to be provided or performed. |
| E. Connecticut State Short Term Disability: | | | | |
| Per statutory requirements of the state in which work is to be provided or performed. | | | | |
All certificates are to stipulate that ten (10) days prior notice of cancellation will be given to the Tenant and to:

RECKSON ASSOCIATES REALTY CORP. 360 Hamilton Avenue White Plains, New York 10601

EXHIBIT E ELECTRICITY SCHEDULE

1. Landlord and Tenant have agreed that Landlord will furnish electrical energy (the "Landlord's Standard Electrical Service") to Tenant for use in the Demised Premises.

2. It is agreed that the Tenant will pay for the installation of a demand watt-hour check meter(s) and related wiring and equipment (collectively, the "Check Meter"), (the location(s) of which is to be approved by Landlord), to monitor Tenant's consumption of electricity in the Demised Premises. Maintenance (including zeroing in of the meter in conjunction with the Landlord), repair and recalibration of the Check Meter shall be Tenant's responsibility at Tenant's sole cost and expense. Tenant shall pay to Landlord on a monthly basis, as additional rent, the cost of such electrical consumption at the average rate Landlord is paying the utility therefor (including without limitation sales taxes and surcharges thereon, plus \$25.00 per month as compensation for Landlord's administrative expenses.)

3. Landlord shall furnish to the Demised Premises Landlord's Standard Electrical Service through the transmission through Landlord's facilities initially installed by Landlord in the Building, in the form of alternating electrical energy, to be used by Tenant, for which Tenant shall pay Landlord as provided in Paragraph 2 above.

4. It is specifically understood that:

(a) All installations of electrical fixtures, appliances and non-standard office equipment within the Demised Premises shall be subject to Landlord's prior written approval which will not be unreasonably withheld.

(b) In the event that Tenant shall require additional electrical energy for use in the Demised Premises and if Landlord's facilities are inadequate for such additional requirements, and if electrical energy for such additional requirements is available to Landlord, Landlord, upon written request from Tenant and at the sole cost and expense of Tenant, will furnish and install such additional wires, risers, conduits, feeders, panels and switch boards as reasonably may be required to supply such additional requirements of the Tenant provided that: (1) same shall be permitted by applicable laws and not in violation of any insurance regulations or recommendations, (2) in Landlord's reasonable judgment, the same are necessary and will not cause permanent damage or injury to the Building or the Demised Premises or cause or create a dangerous or hazardous condition or entail excessive or unreasonable alterations or repairs or interfere with or disturb other tenants or occupants of the Building, and (3) such installation does not preclude expansion of electrical service for other tenants. Tenant covenants and agrees that at all times its use of electrical current shall never exceed the capacity of the feeders of the Building or the risers or wiring installation. It is further covenanted and agreed by Tenant that all of the aforesaid costs and expenses are chargeable and collectable as additional rent and shall be paid by Tenant to Landlord within ten (10) days after rendition of any bill or statement to Tenant therefor.

60

EXHIBIT F COMMON AREAS

59

61

EXHIBIT G LETTER OF CREDIT

[LETTERHEAD OF ISSUER OF LETTER OF CREDIT]

_____, 2003

Reckson Operating Partnership, L.P. c/o Reckson Associates Realty Corp. 225 Broadhollow Road Melville, New York 11747 Attention: Tom Carey, Corporate Controller

REF: IRREVOCABLE LETTER OF CREDIT NO.

GENTLEMEN:

WE HEREBY OPEN OUR UNCONDITIONAL IRREVOCABLE CLEAN LETTER OF CREDIT NO. IN YOUR FAVOR AVAILABLE BY YOUR DRAFT(S) AT SIGHT FOR AN AMOUNT NOT TO EXCEED IN THE AGGREGATE \$______ EFFECTIVE IMMEDIATELY.

ALL DRAFTS SO DRAWN MUST BE MARKED "DRAWN UNDER IRREVOCABLE LETTER OF CREDIT OR [ISSUING BANK], NO. _____, DATED _____, 200_."

THIS LETTER OF CREDIT IS ISSUED, PRESENTABLE AND PAYABLE AT OUR OFFICE AT , NEW YORK [MUST BE IN NEW YORK CITY OR STAMFORD, CONNECTICUT] OR ___, NEW YORK [NEW YORK CITY OR STAMFORD, SUCH OTHER OFFICE IN CONNECTICUT] AS WE MAY DESIGNATE BY WRITTEN NOTICE TO YOU, AND EXPIRES WITH OUR CLOSE OF BUSINESS ON . IT IS A CONDITION OF THIS LETTER OF CREDIT THAT IT SHALL BE AUTOMATICALLY EXTENDED FOR ADDITIONAL TWELVE (12) MONTH PERIODS THROUGH [SIXTY (60) DAYS AFTER LEASE EXPIRATION DATE], UNLESS WE INFORM YOU IN WRITING BY CERTIFIED OR REGISTERED MAIL DISPATCHED BY US AT LEAST SIXTY (60) DAYS PRIOR TO THE THEN EXPIRATION DATE OF THIS LETTER OF CREDIT THAT THIS LETTER OF CREDIT SHALL NOT BE EXTENDED. IN THE EVENT THIS LETTER OF CREDIT IS NOT EXTENDED FOR AN ADDITIONAL PERIOD AS PROVIDED ABOVE, YOU MAY DRAW HEREUNDER. SUCH DRAWING IS TO BE MADE BY MEANS OF A DRAFT ON US AT SIGHT WHICH MUST BE PRESENTED TO US BEFORE THE THEN EXPIRATION DATE OF THIS LETTER OF CREDIT. THIS LETTER OF CREDIT CANNOT BE MODIFIED OR REVOKED WITHOUT YOUR CONSENT. THIS LETTER OF CREDIT IS PAYABLE IN MULTIPLE DRAFTS AND SHALL BE TRANSFERABLE BY YOU WITHOUT ADDITIONAL CHARGE.

WE HEREBY DO UNDERTAKE TO PROMPTLY HONOR YOUR SIGHT DRAFT OR DRAFTS DRAWN ON US, INDICATING OUR LETTER OF CREDIT NO. ______, FOR THE AMOUNT AVAILABLE TO BE DRAWN ON THIS LETTER OF CREDIT UPON PRESENTATION OF YOUR SIGHT DRAFT IN THE FORM OF SCHEDULE A ATTACHED HERETO DRAWN ON US AT OUR OFFICES SPECIFIED ABOVE DURING OUR USUAL BUSINESS HOURS ON OR BEFORE THE EXPIRATION DATE HEREOF.

EXCEPT AS EXPRESSLY STATED HEREIN, THIS UNDERTAKING IS NOT SUBJECT TO ANY AGREEMENTS, REQUIREMENTS OR QUALIFICATION, OUR OBLIGATION UNDER THIS LETTER OF CREDIT IS OUR INDIVIDUAL OBLIGATION AND IS IN NO WAY CONTINGENT UPON REIMBURSEMENT WITH RESPECT THERETO OR UPON OUR ABILITY TO PERFECT ANY LIEN, SECURITY INTEREST OR ANY OTHER REIMBURSEMENT.

IN THE EVENT THE APPLICANT BECOMES A DEBTOR IN A CASE UNDER TITLE 11 OF THE UNITED STATES CODE (THE "BANKRUPTCY CODE"), OR IN ANY OTHER INSOLVENCY OR SIMILAR PROCEEDING, OUR OBLIGATIONS TO THE BENEFICIARY HEREUNDER SHALL NOT BE REDUCED,

LIMITED, IMPAIRED, DISCHARGED, DEFERRED, SUSPENDED, STAYED, TERMINATED OR OTHERWISE AFFECTED BY REASON THEREOF OR BY REASON OF ANY PROVISIONS OF THE BANKRUPTCY CODE (INCLUDING BUT NOT LIMITED TO, SECTIONS 362 AND 502(B) OF THE BANKRUPTCY CODE), OR THE PROVISIONS OF ANY OTHER INSOLVENCY OR SIMILAR LAW.

THIS LETTER OF CREDIT IS SUBJECT TO THE INTERNATIONAL STANDBY PRACTICES 1998, INTERNATIONAL CHAMBER OF COMMERCE PUBLICATION NO. 590, AND SHALL BE DEEMED TO BE A CONTRACT MADE THEREUNDER, AND AS TO MATTERS NOT GOVERNED BY THE INTERNATIONAL STANDBY PRACTICES SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF _____ AND APPLICABLE U.S. LAW.

ALL FEES AND EXPENSES ARE FOR THE ACCOUNT OF THE APPLICANT.

[ISSUER OF LETTER OF CREDIT]

- -----

63

SCHEDULE A TO LETTER OF CREDIT

FOR VALUE RECEIVED

TO: [ISSUER OF LETTER OF CREDIT]

, NEW YORK [LOCATION OF ISSUER IN NEW YORK CITY OR STAMFORD, CONNECTICUT]

EXHIBIT 31.1

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER

I, Nigel P. Ekern, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Clarus Corporation;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2003

/s/ Nigel P. Ekern

EXHIBIT 31.2

CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER

I, Susan Luckfield certify that:

1. I have reviewed this quarterly report on Form 10-Q of Clarus Corporation;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2003

/s/ Susan Luckfield

EXHIBIT 32.1

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Nigel P. Ekern, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of Clarus Corporation on Form 10-Q for the quarter ended September 30, 2003, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of Clarus Corporation.

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Date: November 12, 2003

By: /s/ Nigel P. Ekern

Name: Nigel P. Ekern Title: Chief Administrative Officer (Principal Executive Officer)

EXHIBIT 32.2

CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Susan Luckfield, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of Clarus Corporation on Form 10-Q for the quarter ended September 30, 2003, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of Clarus Corporation.

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Date: November 12, 2003

By: /s/ Susan Luckfield

Name: Susan Luckfield Title: Controller (Principal Financial Officer)